Naval Research Laboratory

Washington, DC 20375-5320



AD-A270 661

NRL/MR/5521--93-7390

Control of Integrated Voice/Data Multi-Hop Radio Networks Via Reduced-Load Approximations

EVAGGELOS GERANIOTIS

Locus, Inc., Alexandria, VA University of Maryland, College Park, MD

IE-HONG LIN

University of Maryland College Park, MD

Prepared for

Communication Systems Branch Information Technology Division

September 30, 1993



Approved for public release; distribution unlimited.



REPORT	Form Approved OMB No. 0704-0188		
Public reporting burden for this collection of infor gethering and maintaining the data needed, and o collection of information, including suggestions fr David Hildhams, Saite 1204, Advance VA 2320	mation is estimated to average 1 hour per re- completing and reviewing the collection of info or reducing this burden, to Washington Head	sponse, including the time for reviewing instru- tomation. Send comments regarding this burd- pusters Services, Directorate for Information	ctions, searching existing data sources. In estimate or any clier sepect of this Operations and Reports, 1215 Jefferson 4 (1988) Michingers, DC 2004
1. AGENCY USE ONLY (Leave Blank)	2. REPORT DATE	3. REPORT TYPE AND DATES COVE	RED
	September 30, 1993	Interim Report 12/91-12/92	
4. TITLE AND SUBTITLE		۸ــــــــــــــــــــــــــــــــــــ	5. FUNDING NUMBERS
Control of Integrated Voice/Dat	PE -61153N PR -RR015-09-41		
6. AUTHOR(S)	WU -DN159-036		
Evaggelos Geraniotis* and Ie-H	ong Lin**		
7. PERFORMING ORGANIZATION NAME	8. PERFORMING ORGANIZATION		
Naval Research Laboratory			REPORT NUMBER
Washington, DC 20375-5320			NRL/MR/5520-93-7390
9. SPONSORING/MONITORING AGENCY NAME(S) AND ADDRESS(ES)			10. SPONSORING/MONITORING
Office of Naval Research			AGENCY REPURT NUMBER
800 North Quincy Street			
	· · · · · · · · · · · · · · · · · · ·		
TI. SUPPLEMENTARY NUTES			
*Locus, Inc. and the University **University of Maryland	of Maryland		
128. DISTRIBUTION/AVAILABILITY STAT	TEMENT		12b. DISTRIBUTION CODE
Approved for public release; dis	stribution unlimited.		
13. ABSTRACT (Meximum 200 words)		,,,,,,, _	
In this report we develop reduce	ed-load approximation techniques	based on the stochastic knapsacl	and the Pascal distribution,
which enable the computationally ef	fficient and accurate evaluation of av in integrated voice/data multi-	f the probability of voice blocking	g, the probability of data summation techniques are also
used to verify the accuracy of the a	pproximations. The reduced-loa	d approximations exhibit excellen	t to very satisfactory accuracy
for the entire range of network and	traffic parameters of interest, wh	hile the computational effort nece	ssary for their evaluation is
They are also used successfully to a	exact expressions (which is profil approximate the derivatives (sens	itivities) of the above performanc	e measures with respect to
network and traffic parameters. Th	nese approximations are applicabl	e to single-rate and multi-rate voi	ice models, as well as to models
with voice activity and silence period	ods. Besides the accurate and tim	ne-efficient performance evaluation of these approximations enables the	on of integrated voice/data multi-
optimization purposes.	nai enterency and the accuracy o	a unese approximations enables un	
14. SUBJECT TERMS			15. NUMBER OF PAGES
Communications network	Radio network		171
Voice/Data Integration Admission Control	Approximation		16. PRICE CODE
17. SECURITY CLASSIFICATION OF REPORT	18. SECURITY CLASSIFICATION OF THIS PAGE	19. SECURITY CLASSIFICATION OF ABSTRACT	20. LIMITATION OF ABSTRACT
UNCLASSIFIED	UNCLASSIFIED	UNCLASSIFIED	UL .

NSN 7540-01-280-5500

.

Ψ

CONTENTS

1. INTRODUCTION
1.1. Outline of the Report
2. SYSTEM MODEL FOR THE JOINT VOICE ADMISSION CON- TROL AND DATA ROUTING PROBLEM
2.1. Network Model
2.1.1. General Multi-Hop Multi-Rate Network Model
2.1.2. Multi-Hop Radio Network Model
2.2. Source Models
2.2.1. Multi-Hop Multi-Rate Network
2.2.2. Single-Rate Multi-Hop Radio Network
2.3. Cost Function / Performance Measures
2.4. Basic Notation
2.4.1. General Multi-Rate Multi-Hop Network
2.4.2. Single-Rate Multi-Hop Radio Network
3. STEADY-STATE DISTRIBUTION FOR THE SYSTEM STATE
4. REDUCED LOAD APPROXIMATIONS FOR THE VOICE BLOCK- ING PROBABILITY
4.1. Kelly's Approximation
4.2. Knapsack Approximation
4.3. Pascal Approximation
4.4. Mitra's Approximation
4.5. Knapsack and Pascal Approximations for the Radio Network Model of Section 2.2
5. KNAPSACK APPROXIMATION TO THE PROBABILITY OF DATA QUEUEING
5.1. General Multi-Hop Multi-Rate Networks
5.2. Radio Network Model of Sections 2.1.2, 2.2.2, and 2.4.2
6. PASCAL APPROXIMATION TO THE PROBABILITY OF DATA QUEUEING
6.1. General Multi-Rate Networks
6.2. Radio Network Model of Sections 2.1.2, 2.2.2, and 2.4.2
7. EXTENSION OF APPROXIMATIONS TO M/D/c DATA MODELS AND THE AVERAGE QUEUEING DELAY AS THE PERFOR- MANCE MEASURE
7.1. Approximations to the Probability of Data Queueing for M/D/c

Data Models
7.2. Approximations for the Average Queueing Delay of Data
7.2.1. Knapsack Approximation
7.2.2. Pascal Approximation
7.3. Network Model Modification for Guaranteeing Finite Data Delays
8. EVALUATION OF PERFORMANCE MEASURES VIA THE MONTE-CARLO SUMMATION METHOD
8.1. The Monte-Carlo Summation Method
8.2. Monte-Carlo Summation for the Probability of Voice Blocking
8.3. Monte-Carlo Summation for the Probability of Data Queueing8.4. Monte-Carlo Summation for the Average Queueing Data Delay
9. ADMISSION CONTROL VIA REDUCED-LOAD APPROXIMA- TIONS
9.1. Models for Admission Control
9.2. The Knapsack Approximation Method via Conditioning on Addi- tional Constraints
9.2.1. Probability of Voice Blocking
9.2.2. Probability of Data Queueing
9.2.3. Extension to Multiple Control Constraints
9.3. The Knapsack Approximation via the Introduction of Fictitious Links
10. REVENUE SENSITIVITY VIA REDUCED-LOAD APPROXIMA- TIONS
10.1. Revenue Sensitivities for General Multi-Rate Networks
10.1.1. Sensitivity of Voice Revenue
10.1.2. Sensitivity of Data Revenue
10.2. Revenue Sensitivities for the Multi-Hop Radio Network of Sec- tions 2.1.2 and 2.2.2
11. NUMERICAL RESULTS AND PERFORMANCE COMPARISONS
11.1. Network and Traffic Models for the Paradigms Used
11.2. Comparisons of Approximations for the Radio Network of Fig- ure 2 without Admission Control
11.2.1. Results for a Radio Network with Eight Transceivers per Node
11.2.2. Results for a Network with Fewer Transceivers per Node

11.2.3. Results for a System with Transceivers Reserved for Data
11.3. Comparisons of Approximations for the Multi-Rate Network of Figure 3 without Admission Control
11.4. Threshold Admission Control for the Radio Network of Figure 2
11.5. Voice and Data Revenue Sensitivities for Networks of Figures 2 and 3
11.6. Required Computational Effort for Various Approximations12. CONCLUSIONS
12.1. Accuracy and Computational Advantage of Approximations
12.2. Application of New Approximation Techniques to a Broad Range of Network Architectures and Traffic Types
12.3. Application to Control and Optimization Problems
12.4. Application to High-Speed Networks
ACKNOWLEDGEMENT
REFERENCES
APPENDIX A: Proof of Part (a) of Proposition 3.1 (System Steady-State Distribution)
APPENDIX B: M/M/c and M/D/c Data Models
APPENDIX C: Proof of Proposition 5.1 (Knapsack Approximation)
APPENDIX D: Four-Dimensional Knapsack Approximation (Proofs of Propositions 5.2 and 9.1)
APPENDIX E: Proof of Proposition 6.1 (Pascal Approximation)
APPENDIX F: Proof of Proposition 6.2 (Pascal Approximation)
APPENDIX G: The Knapsack and Pascal Approximations to the Probabil- ities of Blocking Voice and Queueing Data Are Identical for Net-
works with Single-Rate Traffic
APPENDIX H: Proof of Proposition 10.1

DTIC QUALLY LINE LINE &

v

Accesio	n For			
NTIS DTIC Unanno Justific	CRA&I TAB ounced ation			
By Dist. ibution /				
Availability Codes				
Dist	Avail and for Special			
A-1				

CONTROL OF INTEGRATED VOICE/DATA MULTI-HOP RADIO NETWORKS VIA REDUCED-LOAD APPROXIMATIONS

1. INTRODUCTION

In this report, we address one of the major issues in multi-media networks: the combined (joint) admission control of voice calls and routing of data packets through the network. These two problems have been traditionally dealt with separately [1]-[3]. However, with the advent of ISDN and multi-media services [3], it has become necessary to design routing and call set up or scheduling schemes jointly, so that the network resources (bandwidth of links, processors at nodes) are used efficiently and all user requirements about the offered quality of services (QOS) of the different traffic types (e.g., data, voice, and video) are met.

One of the major obstacles in the joint optimization of voice admission control (or scheduling) and data routing schemes and, in general, of schemes which control admission, access, and routing for multi-media traffic, is the difficulty in obtaining (or the complete lack of) closed-form expressions for the performance measures characterizing QOS of the different traffic types.

In many practical situations where such optimizations are required, simplistic approximations of low accuracy are used to evaluate the performance measures of interest. These approximations neglect the interaction and interdependencies caused by the multihop network operation and the sharing of the network resources by the different traffic types. Consequently, the control schemes derived from the optimization of these coarsely approximated performance measures are sub-optimal and the network resources may not be utilized efficiently.

By contrast, the emphasis of this report lies in (i) the identification of appropriate methods for approximating accurately the performance measures involved in the problems of admission control of voice calls and of routing of data packets in integrated networks and (ii) the derivation of optimal schemes for admission control and routing on the basis of these approximations. In this context, some existing results on reduced-load approximations for voice traffic [4]-[10] are used, their applicability is extended from wired networks to wireless networks, and their accuracy is validated for a broader range of network and traffic parameters; in addition, new results for data traffic are derived for the first time. All approximations used are compared to each other and to confidence intervals of the

Manuscript approved June 23, 1993.

actual performance measures derived via the Monte-Carlo summation method for a broad range of parameters of the traffic types.

Although the approximations and performance measures used in this report pertain to voice and data traffic and are used for the optimization of voice admission control and data routing schemes, they are in principle applicable to other problems (e.g., scheduling or set up of calls and data routing) and different traffic types (e.g., data, voice, and video). Actually, regarding the latter issue, our results cover multi-rate scenarios, according to which different traffic types may have different bandwidth requirements; thus, if voice but not video traffic is involved, voice traffic of different bit rates (and thus quality) can be accommodated. When video traffic is also present, our approach requires substantial modification to accommodate the variable rate traffic of video sources; however, it can be definitely extended to this case and thus it finds application to true multi-media scenarios.

Moreover, in our model of voice sources, we accommodate both periods of activity and silence. If the silent periods can be sensed by the network nodes, then the data users can, at least in theory, take advantage of this and utilize the released bandwidth, thus increasing the efficiency of the protocols. In practice, this monitoring of silence periods and talkspurts can be readily implemented only in certain situations and architectures and at the expense of channel bandwidth and additional complication in the network protocols; this issue is discussed further in Section 2.2.1.

Finally as elaborated in Section 12.4 the approximations of this report are also applicable (after suitable modification) to interesting problems of high-speed networks such as (i) call set-up and admission control in Asynchronous Transfer Mode (ATM) and (ii) multicasting of hierarchically encoded data.

The cost function that can be employed in the optimization of the voice admission control and data routing schemes consists of the weighted sum of

(i) the average blocking probabilities along the paths of voice calls

and

(ii1) the average probabilities of queuing of data along the links of the network or

(ii2) the average data delays along the links of the network

Closed-form expressions for the above quantities are either not available [as is the case for (ii1) and (ii2)], or, even when they are available [as is the case for the product form of (i)], they are very difficult to compute for moderate to large size networks. This difficulty

is actually amplified by the fact that in optimization problems such as joint voice and data routing or voice admission control and data routing, the performance measures above may have to be evaluated repeatedly for several differents paths or links. It is exactly this difficulty that we attempt to circumvent in our work.

The approach followed in this report can then be summarized as follows. We use existing approximations or develop new ones for the average probability of voice blocking, the average probability of data queueing, and the average data-packet queueing data delay, and employ those instead of the exact expressions to derive near-optimal admission control schemes based on thresholds for the voice traffic (see Table 20). We do not derive optimal routing schemes for the data traffic in this report but since we approximate accurately the data revenue (and voice revenue) sensitivities with respect to the link capacities, the offered voice loads, and offered data loads, we can obtain near-optimal data routing schemes by using standard routing algorithms [2] based on these derivatives (sensitivities).

For voice blocking probabilities, we consider approximation methods suggested by Kelly [4]-[5] and the knapsack, Pascal, and Monte-Carlo summation approximation methods employed by Ross [6]-[9]. Mitra's approximation method [10] is also critically considered. All these approximations are known to be **asymptotically correct** (accurate) under a limiting regime, according to which both the capacity of the links of the radio network and the average input voice traffic (loads) increase to large values, while their ratio remains constant.

These approximation techniques are based on several distinct concepts and have varying degrees of accuracy and convergence range. Kelly's approximation [4]-[5] is based on an inter-link independence assumption for voice traffic. The knapsack approximation (used by Ross in [6] and [8]) is based on a stochastic knapsack concept. The Pascal approximation (used by Ross in [6] and [9]) is based on a birth-death process for modeling voice traffic that follows the Pascal distribution. The Monte-Carlo summation method employs acceptance/rejection methods used primarily in simulation techniques in order to evaluate multi-dimensional integrals under constraints [7]. This method generates confidence intervals for estimators of the actual performance measures of interest, and thus can be used to provide reliable means of comparison among all other approximations even when the exact expressions are not available or require prohibitive computational effort. Mitra's approximation [10] uses a Taylor series expansion on the normalization constant of the product form of the voice steady state probability distribution.

As part of our effort, the accuracy and convergence range of the knapsack and Pascal approximations were verified by comparing the approximate results with those obtained using the exact expressions (where feasible) or confidence intervals generated via the Monte-Carlo summation method. The agreement was found to be very satisfactory, not only for the average performance measures, i.e., the probabilities of voice blocking and data queueing (or the data queueing delay) when averaged over the traffic of all circuits or links, respectively, but also for similar performance measures corresponding to the individual circuits or links of the network.

Consequently, we use the knapsack approximation for the optimization of the thresholds for the admission control of voice calls. Besides the wired multi-rate loss networks of [6]-[10], we also considered radio networks modeled as in [11]-[12] where the transceivers at the nodes (rather than the link capacities) are the network resources. The application of these approximations to radio networks is new, and different technical problems than those of the multi-rate loss networks had to be addressed. However, the results based on Mitra's approximation were rather disappointing and will not be used in our optimization, despite the fact that we had extended the approach to general network topologies and multi-rate networks. A brief discussion of Mitra's approach and our assessment of its applicability to the problems considered in this report is presented in Section 4.4.

For the probabilities of queueing and the queueing delays of data, we use Kleinrock's independence assumption for data traffic; according to this assumption, the distribution of interarrival data packet times to the various internal nodes of the network remains exponential, even after the data packets have been serviced at intermediate links. Voice always maintains priority over data in our models. Due to the much longer average duration of typical voice calls compared to that of data packets (whose arrival process is characterized by a Poisson distribution), it is reasonable to assume that the voice state of the network changes much slower than that of the data state.

We can thus evaluate the probability of queueing of data (and the queueing delay) conditioned on the voice state (i.e., the number of active and inactive voice calls in the network); then we average with respect to the steady-state probability distribution of the voice state. Both M/M/c and M/D/c models for data queueing at the links have been

considered. We have performed the aforementioned averaging of the M/M/c or M/D/c expressions (conditioned on the voice state) with respect to the voice state, according to the knapsack and Pascal approximation methods. This is a novel approach first appearing in this report. We limited attention to these data models because of constraints in the preparation time and the length of this report; the applicability of our approach is not restricted to these models, it can be applied (with proper modification) to any other data models as long as the assumption holds that changes in the state of the network voice traffic are much slower than those in the data traffic.

Finally, the sensitivities of suitably defined voice and data revenue measures with respect to link capacities (or number of node transceivers), voice loads, and data loads were evaluated via the knapsack approximation and shown to be very close to the actual values (refer to Tables 21 and 22). Again, these approximate sensitivities are much more computationally efficient than the cumbersome (and usually prohibitive) exact expressions. Actually, as our results establish, there will be almost negligible loss in revenue when voice-control schemes use these approximate sensitivities in place of the exact ones.

With the help of the aforementioned revenue sensitivities, important practical problems of allocating additional network resources in response to increasing voice and/or data network traffic demand can be easily handled with our approach, as well as problems of data routing in which the derivatives of the data delay (or the probability of queueing) are used by standard optimal routing algorithms. In this context near-optimal schemes for the joint voice admission control and data routing can be derived for both singlerate and multi-rate networks. This can also be accomplished for near-optimal schemes for joint call set-up and data routing. The range of applications of this methodology actually includes single-rate and multi-rate networks, wired or wireless (radio) networks, as well as high-speed networks.

It should be noted that, as summarized in Table 23, the computational effort required for the knapsack and Pascal approximations and the Monte-Carlo summation method compares very favorably with that necessary for the evaluation of the exact expressions. The reduced complexity permits the use of the approximations of this report for on-line optimization purposes.

1.1. Outline of the Report

The report is organized as follows. In Section 2, the network models, the source model for voice and data traffic, and the cost function together with the individual performance measures for voice and data of interest are described in detail. In Section 3, the steady state probability distribution of the voice state over the entire network is derived and the evaluation of the probability of voice-call blocking and of data-packet queueing is discussed. In Section 4, four approximation methods for evaluating the probability of voice blocking are reviewed, namely, Kelly's, knapsack, Pascal, and Mitra's. In Section 5, the knapsack approximation is extended and applied to the probability of data queueing. Subsequently, in Section 6 the Pascal approximation is extended and applied to the probability of data queueing. Following is Section 7 with the application of the knapsack and Pascal approximation techniques to the average data packet queueing delay. In Section 8, the Monte-Carlo Summation method is described in detail for the evaluation of the probabilities of voice blocking and data queueing. In Section 9, the knapsack approximation method is applied to voice admission control problems. In Section 10, the sensitivities of suitably defined voice and data revenue functions with respect to the link capacities, voice loads, and data loads are evaluated via the knapsack approximation. In Section 11, the various approximations are compared to each other and to confidence intervals generated via the Monte-Carlo summation method; the use of the approximations in obtaining near optimal thresholds for admission control is also described. Finally, in Section 12 several conclusions are drawn from this study.

2. SYSTEM MODEL FOR THE JOINT VOICE ADMISSION CONTROL AND DATA ROUTING PROBLEM

In this section we present the network and traffic models of interest in this study. A general purpose multi-hop multi-rate-voice/data network model and a voice/data multi-hop radio model are described first, followed by detailed models for the voice and data traffic, and by the definition of suitable performance measures.

2.1 Network Model

In both network models FDMA (frequency division multiple-access) is the multiplexing technique used; thus frequency channels (rather than time slots) are used to carry the packetized traffic (voice or data). Circuit-switching is the primary mode of communication for the network (for voice traffic), whereas packet-switching is used for data traffic.

2.1.1 General Multi-Hop Multi-Rate Network Model

The network we consider can be defined by a triplet $(\mathcal{N}, \mathcal{L}, \underline{c})$ where \mathcal{N} is a set of nodes, \mathcal{L} is the set of all possible directed links (each directed link *l* connects two nodes in \mathcal{N}) and $\underline{c} = [c_l, l \in \mathcal{L}]_{1 \times |\mathcal{L}|}$ is the (row) vector containing the capacity (number of channels) c_l for each link $l \in \mathcal{L}$, where $|\mathcal{L}|$ denotes the number of elements in set \mathcal{L} . The set of consecutive links directed from source node *n* to destination node *m* constitutes the path *p*; we denote by \mathcal{P} the set of all such paths; similarly \mathcal{P}_l denotes the set of all paths that use link *l*; i.e.,

$$\mathcal{P}_l = \{ p \in \mathcal{P} \mid l \in p \}, \ l \in \mathcal{L}$$

and is used frequently in our analysis. The routes followed by the network traffic are characterized by the $|\mathcal{L}| \times |\mathcal{P}|$ routing matrix **A** whose elements $A_{lp} = 1$, if the *p*-th path $(p \in \mathcal{P})$ uses link $l \ (l \in \mathcal{L})$, and $A_{lp} = 0$ otherwise.

2.1.2 Multi-Hop Radio Network Model

The notation is basically the same as in the previous section. However, in this model (motivated by the work of [11]), the number of transceivers at each node is the important resource instead of the capacity of the links of the previous section. Let T_n denote the number of transceivers at node n ($n \in \mathcal{N}$). The vector of node transceivers $\underline{T} = [T_n, n \in \mathcal{N}]_{1 \times |\mathcal{N}|}$ replaces \underline{c} defined above. Moreover, instead of \mathcal{P}_l defined above, the set \mathcal{P}_n defined as

$$\mathcal{P}_n = \{ p \in \mathcal{P} \mid n \in p \}, \ n \in \mathcal{N}$$

and denoting the set of all paths p intersecting at node n appears frequently in our analysis.

2.2 Source Models

2.2.1 Multi-Hop Multi-Rate Network

We assume that the data packet and voice call arrival processes from outside the network with originating node n and destination node m are Poisson distributed with rates F_{nm}^d and F_{nm}^v , respectively. Moreover, $F_l^d(F_l^v)$ denotes the data (voice) flow in link $l, F_p^d(F_p^v)$ denotes the data (voice) flow input from outside the network to path p, μ_l^d is the data service rate on the *l*-th link $(l \in \mathcal{L})$, and μ_p^v is the voice service rate on the *p*-th path $(p \in \mathcal{P})$. The units of all these quantities (arrival and service rates) are packets per sec.

Since the network we are considering is of middle-size or larger and the data traffic loads are moderate to heavy, we may assume that Kleinrock's independence assumption holds for data. According to this approximation, which has been verified through simulations for networks with data-only traffic, the data-packet serial arrival process at each link, which includes both arrivals from outside the network, as well as packets forwarded by upstream nodes, can be accurately approximated with a Poisson process independent of the interaction taking place inside the network among the various nodes (queueing and servicing). For a multi-media network this assumption/approximation for the data traffic has not been verified but we expect it is valid for moderate to large size networks. In particular, we expect it to be valid for scenarios characterized by voice traffic that changes much slower than the data traffic; because as elaborated in Section 2.3 below, in these cases we can condition on the state of the voice traffic and work with conditional performance measures for the data traffic.

Two models are considered for data traffic. In the first, the packet length is exponentially distributed, resulting in an M/M/c queueing model, where c is the number of servers (channels) available for data. * The mean packet length is denoted as $1/\mu_{nm}^d$ (in secs). In the second, the packet length is a deterministic constant $(1/\mu_{nm}^d)$, resulting in an M/D/c model.

The length of voice calls is exponentially distributed with parameter μ_{nm}^{v} . Actually, we assume that every call is composed of active and inactive (silent) periods which are exponentially distributed with parameters α and β (in sec⁻¹), respectively. The mean

^{*} The number of channels available for data depends on the number of voice calls in progress.

duration of active periods is $1/\alpha$ (secs) and the mean duration of inactive periods is $1/\beta$. The speaker "activity fraction" is defined as $\frac{1/\alpha}{1/\alpha+1/\beta} = \frac{\beta}{\alpha+\beta}$ and typically takes the value 0.4 for normal conversational speech (in half-duplex mode). In full-duplex mode, the value 40% is valid for the activity of each of the source and destination nodes, in the sense that 40% of the time each node talks and another 40% of the time it listens to the other side talking. Therefore, the assigned circuit for the voice call remains occupied for 80% of the time; an intermediate node will be in transmit mode 80% of the time, i.e., 40% of the time transmitting in the "downstream" direction and 40% of the time transmitting in the "upstream" direction. The quantity $\beta/(\alpha + \beta)$, and its complement with respect to 1, i.e., the speaker silence fraction $\alpha/(\alpha + \beta)$, enter in the key expressions for the probability distribution of the state of voice calls in the network (see Section 3).

Under the model of the previous paragraph, the rate of active arriving calls is $F_{nm}^{\nu}\beta/(\alpha+\beta)$ and the rate of inactive arriving calls is $F_{nm}^{\nu}\alpha/(\alpha+\beta)^*$; an active call turns inactive with rate α and a silent call becomes active with rate β .

We assume that the portion of the channel capacity (the time-varying number of channels c in the M/M/c and M/D/c queueing models) left unused by voice calls is used by data users. This represents a very desirable situation with the most efficient use of channel resources but not necessarily an easy one to achieve. It is required that the status of all voice conversations using a link is monitored and that this information is fed to the data users that have access to that link so that they can use the channel; they must however stop using the channel once the next talkspurt begins. If the silent period is less than a packet length and thus the data message must be interrupted by the resumption of the voice call, we may assume that the data message completes its packet transmission before the voice call resumes; since the packet length is so much smaller than the typical length of the voice conversation (or even the length of a talkspurt), the effect on the resumed voice conversation is anticipated to be negligible.

In systems that can not take advantage of voice call silent periods, all calls are assumed to be active throughout their duration. Under this model it is commonly accepted that the probability of blocking remains relatively insensitive to the specific form for the probability

^{*} Although it would be more reasonable to assume that all calls are active when they arrive, the assumption that they may be in either the active or the silent mode simplifies the mathematical model, while resulting in little impact on numerical results.

distribution of the duration of voice calls [17] and it basically depends only on the mean of call length $1/\mu_{nm}^{v}$. With this simpler model the entire analysis is also simplified and can be obtained as a special case of the analysis for the two-state (active, inactive) voice call model presented in the following sections. This translation requires that we set $\alpha = 0$, $\beta = 1$, that the number of inactive calls denoted by n_{p}^{s} is set to 0 in all equations of the subsequent sections which involve it, and that we use the conventions 0! = 1 and $0^{0} = 1$.

It is assumed that the bandwidth of each of the channels comprising the links of the network is equal to the data rate (in bits per sec) of the data traffic and is obtained from the data packet arrival rate (in packets per sec). Thus the capacity c_l of any link l ($l \in \mathcal{L}$) represents the number of channels and is an integer. For voice traffic we consider single-rate and multiple-rate scenarios. In the single-rate case all voice calls have identical data rate (denoted by r) and require the same bandwidth for transmission. In the multi-rate case we assume that all voice sources using path p have data rate r_p as in [6]. The values of r and r_p used in the following sections are normalized with respect to the bandwidth of a single channel. These normalized r and r_p are not necessarily integers.

At this point let us clarify that by using the multi-rate traffic model of the previous paragraph we can model multi-class voice traffic. Thus, voice traffic of several different quality specifications (such as fully compressed, partially compressed, or uncompressed voice, secure voice etc.) and bandwidth requirements can be modeled. Indeed, all we need to do is to characterize the paths $p \in \mathcal{P}$ not only by the collection of consecutive links included in them, but, also by the (possibly) different data rates r_p of voice traffic that flow through them. In this way more than one element of \mathcal{P} may follow the same physical path (route) inside the network but carry different amounts of information (have different data rates). Finally, the multi-class traffic of the above model need not be limited to voice only; traffic types requiring higher bandwidth than voice, such as video of different bandwidth requirements and rate variabilities [such as video telephony (teleconferencing) with medium high variable rate or full-motion video (television) with high variable rate] can also be dealt with in the same manner.

Since voice calls can be blocked but can not tolerate delay whereas data can be delayed, we assume that voice has priority over data. Data arrivals are routed in a packet-switching manner so that the input data flow in each node is separated into several subflows and each subflow takes a different path to its destination. Each voice call, after being admitted, is allocated a channel along a fixed multihop path to its destination in a virtual circuit manner, that is, once the path is chosen, the call uses only this path to transmit until the call is finished.

On the other hand, data messages (following a $M/D/c'_l$ traffic model) can use only the residual capacity c'_l , that is the portion of the link capacity c_l that remains unused after the active voice calls on paths employing link *l* have occupied the necessary number of c' annels. The appropriate expression for the data link capacity c'_l is provided in Section 2.4.1 below.

An alternative model that always guarantees that a portion of the capacity of each link is allocated to data communications is also considered in this report [refer to equation (7.17) of Section 7.3].

2.2.2 Single-Rate Multi-Hop Radio Network

The model for voice calls for the radio model of Section 2.1.2 is similar to the one described above, except that we address only the case in which a single-rate model for all calls is adopted, that is,

$$r_p = r = 1$$
 for all $p \in \mathcal{P}$.

Also data messages use the same data rate as voice calls.

For the data traffic we use an $M/D/c'_i$ model where the number of data messages that can be transmitted simultaneously over link l = (n, m) (connecting nodes n and m) is limited by the residual capacity c'_i of the link; that is, the remaining number of channels once the active voice calls have occupied the necessary transceivers at nodes n and m. The appropriate expression for the residual capacity is given in Section 2.4.2. Data packets are queued at buffers available at the nodes. The above M/D/c model assumption can not be fully justified here as it was done for the general wired network of Section 2.2.1., because of the need to coordinate the transceivers at the two nodes of each link. A radio network will most probably use some more complicated access protocol for the data traffic, resulting in arrivals to intermediate data links (or nodes) which are not Poisson. Therefore Kleinrock's independence assumption is less likely to be valid in a radio environment. However, the simple M/D/c model enables us to evaluate the accuracy of the approximations described in this report without having to evaluate complicated protocols for the data traffic. For different protocols than the M/D/c, certain conditional probabilities pertaining to data behavior [which are conditioned on the number of ongoing (active) voice calls] will have to be used instead of the M/D/c formulas; except for this change the basic steps of our approach are applicable to these cases as well. A more detailed discussion of this issue is provided at the end of Section 2.4.2.

An alternative model that always guarantees that some transceivers (and thus some link capacity) are dedicated to data traffic is also considered in Section 7.3. [refer to equations (7.18)-(7.19)].

2.3 Cost Function / Performance Measures

The performance measures of interest to our study are:

- (i) the probability of blocking of calls B_p along each path $p \ (p \in \mathcal{P})$, as well as its average over the voice arrival process at all paths, \overline{B} ,
- (ii) the probability of queuing data Q_l at each link $l \ (l \in \mathcal{L})$, as well as its average over the data arrival process at all links, \bar{Q} ,
- (iii) and the average queueing delay of data W_l at each link $l \ (l \in \mathcal{L})$, as well as its average over the data arrival process at all links, \overline{W} .

The first of these quantities B_p is defined as the probability that an arriving call of class p (i.e., destined to follow voice path p) finds all available channels busy at one or more links on path p, and is therefore blocked. A precise mathematical definition of B_p is provided by equation (3.12b) of Section 3 as the quotient of two normalization constants. The quantity Q_l is the probability that a data packet arriving at link l finds all channels of that link busy and is queued, and W_l is the average delay at the queue of link l experienced by a typical data packet. Since W_l provides only the average value of the delay and the distribution $\neg f$ the delay is very difficult to obtain, we also provide the probability of queueing Q_l (which is easier to obtain) in order to supplement the information given by W_l about queueing at link l. Links are considered in isolation here because of Kleinrock's independence assumption, which we assume is valid for the data traffic; this is discussed in more detail below.

The aforementioned average quantities \bar{B}, \bar{Q} , and \bar{W} are defined by

$$\bar{B} = \frac{\sum_{p \in \mathcal{P}} \rho_p^v B_p}{\sum_{p \in \mathcal{P}} \rho_p^v},\tag{2.1a}$$

$$\bar{Q} = \frac{\sum_{l \in \mathcal{L}} \rho_l^d Q_l}{\sum_{l \in \mathcal{L}} \rho_l^d},$$
(2.1b)

and

$$\bar{W} = \frac{\sum_{l \in \mathcal{L}} \rho_l^d W_l}{\sum_{l \in \mathcal{L}} \rho_l^d},$$
(2.1c)

respectively, where the utilization factors for voice and data traffic $(\rho_p^v \text{ and } \rho_l^d)$, are defined in equations (2.7) and (2.8) later in this section.

The cost function C of interest to problems of joint admission control and data routing (or joint voice and data routing) consists of weighted sums of \overline{B} and \overline{Q} or of \overline{B} and \overline{W} , that is,

$$C = K_1 \ddot{B} + K_2 \bar{Q} \tag{2.2a}$$

or

$$C' = K_1 \bar{B} + K_2 \bar{W} \tag{2.2b}$$

where $K_1 + K_2 = 1$ and $K_1, K_2 \ge 0$. Therefore, in order to perform any optimization involving the above cost functions we need to have expressions (closed-form or accurate approximations) available for the probability of voice blocking (\bar{B}) , the probability of data queueing (\bar{Q}) , and the average queueing delay of data (\bar{W}) . Since such computations are typically invoked many times during optimization the evaluation of the exact expressions (or the approximations) should be computationally efficient, otherwise optimization is not computationally feasible.

Let us revisit now B_p , the performance measure for the voice calls. Since voice calls have preemptive priority over data, the performance of voice calls is not affected by the data; consequently, the well-known product-form solution of the probability of blocking ([3], [4]) is valid. The difficulty in evaluating B_p of course lies in the computational complexity of the expression for the probability of blocking. Our approach here is to consider approximations to the probability of blocking that are asymptotically accurate under specific limiting regimes. Although B_p is of greatest interest in most applications, two additional voice blocking probabilities are also used: B_{lp} (approximations to it are denoted by L_{lp} in Section 4), the probability that the capacity of link l along the path of a voice call of class p ($p \in \mathcal{P}$) is not available [whose precise mathematical definition is given in (3.12a) as the quotient of two normalization constants]; and B_l , the probability of blocking of voice calls due to unavailability of the capacity of link l ($l \in \mathcal{L}$). Both B_p and B_l are approximated with the help of B_{lp} ; details are provided in Section 4. Next consider Q_l and W_l , the performance measures for data. Since the states of data at different links are dependent—for example, for all links on the path of a voice call the data states depend on the state of calls of the path and as such are mutually dependent—it is not possible to consider Q_l for link l without the coupling with the other links. To obtain Q_l $(l \in \mathcal{L})$ we have to consider the state of all links at a particular time, which is very difficult when the output data process of each link is not Poisson.

To facilitate the analysis of data we need an additional assumption (beyond Kleinrock's independence assumption). Since the state of voice calls changes much slower than the data states, we assume that data can reach steady-state during the sojourn time of calls within a particular state. The validity of this assumption has not been verified via simulation; in the future, we plan to check the accuracy of this approximation via simulation, at least for small-size networks with voice and data traffic. However, a similar assumption has been made in [14] for the analysis of Voice/Data (VD) Interleaved-Frame Fixed-Length (IFFL) protocols in the context of movable-boundary channel-access schemes for integrated voice/data networks. It was shown that this assumption is pessimistic at moderate to high throughput levels. VD-IFFL protocols work with time-slotted networks, and use reservations for the voice traffic and IFFL (which combines reservation with contention) for the data traffic. Under the above assumption and the aforementioned Kleinrock's assumption (discussed in detail in Section 2.2.1) we can examine each link in isolation and consider M/M/c' or M/D/c' models for the data conditioned on the state of the voice taking on a particular value.

2.4 Basic Notation

The notation introduced here is used in all subsequent sections. However, additional notation is introduced in each section as necessary. In this section attaching an "a" or a "b" to the number of an equation respectively signifies the single-rate or the multi-rate general (not necessarily radio) network scenarios; attaching a "c" signifies a single-rate radio network with transceivers at the nodes.

2.4.1 General Multi-Rate Multi-Hop Network

The system state is described by a triplet that contains global (networkwide) information in the voice state, along with local information on the data state at one particular link.

Denote the state as $(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d})$, where

 $\underline{N}^{v} = [n_{p}^{v} : n_{p}^{v} = \text{number of calls taking path } p \text{ in the system, } p \in \mathcal{P}]_{1 \times |\mathcal{P}|}$ $\underline{N}^{s} = [n_{p}^{s} : n_{p}^{s} = \text{number of calls in silent phase out of } n_{p}^{v} \text{ calls, } p \in \mathcal{P}]_{1 \times |\mathcal{P}|}$ $n_{l}^{d} \triangleq \text{number of data messages at link } l$

Thus the number of ongoing voice calls along the paths (circuits) and the number of data messages along the links are the network states of interest to this report. System performance must be evaluated for each of the \mathcal{L} links in the network. Links can be considered in isolation because of Kleinrock's interlink independence assumption (discussed at the beginning of Section 2.2.1) and the additional assumption about the state of voice varying much slower than that of data (cited at the end of Section 2.3) which decouples the variations of data traffic from voice traffic and gives meaning to Kleinrock's assumption for the conditional data traffic (conditioned on the voice state). We reiterate here that it is not necessary for the validity of our approach to consider the data links in isolation; we do it because it simplifies the computations (at least with respect to the data part) and allows us to test our approximations for performance measures (for data traffic) which are available in closed form (when conditioned on the voice state). Our approach can be extended to apply to performance measures (for data) which reflect the interaction between several data links; however, this requires additional computational complexity. This issue is discussed further in Section 2.4.2.

Note that $|\mathcal{P}|$ denotes the number of the elements in set \mathcal{P} , i.e., it is the number of call types and thus the number of paths, \underline{N}^{v} and \underline{N}^{s} are $|\mathcal{P}|$ -dimensional vectors, and n_{l}^{d} is a scalar. Denote the set of all possible $(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d})$ for the single-rate case as Ω , where

$$\Omega \triangleq \left\{ \left(\underline{N^{v}}, \underline{N^{s}}, n_{l}^{d}\right) \middle| 0 \le n_{p}^{s} \le n_{p}^{v}, 0 \le n_{p}^{v} \le c_{p}/r, p \in \mathcal{P}; \sum_{p \in \mathcal{P}_{l}} n_{p}^{v} \le c_{l}/r, l \in \mathcal{L} \right\}$$
(2.3*a*)

where $c_p = \min \{c_l, l \in p\}$ denotes the capacity of path p, and \mathcal{P}_l was defined in Section 2.1.1. The terms c_p/r and c_l/r in (2.3a) represent the actual numbers of voice channels in c_p and c_l ; recall that the link capacities c_l are normalized to the data traffic rate and thus represent the maximum number of data channels, and division by r (the voice data rate) is required in order to provide the corresponding number of voice channels.

Actually, the intuitive constraint $0 \le n_p^v \le c_p/r$ for $p \in \mathcal{P}$, which guarantees that the number of calls on path p does not exceed the (voice) capacity of path p defined as $c_p =$

min $\{c_l, l \in p\}$, is satisfied whenever the more powerful constraint $\sum_{p \in \mathcal{P}_l} n_p^v \leq c_l/r$, $l \in \mathcal{L}$ is met. Therefore, the necessary (with the minimum number of constraints) form of Ω is

$$\Omega \stackrel{\Delta}{=} \left\{ \left(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d} \right) \middle| 0 \leq n_{p}^{s} \leq n_{p}^{v}, p \in \mathcal{P}; \sum_{p \in \mathcal{P}_{l}} n_{p}^{v} \leq c_{l}/r, \ l \in \mathcal{L} \right\}.$$
(2.3*a*')

Note that in the multi-rate case Ω becomes

$$\Omega \stackrel{\Delta}{=} \left\{ \left(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d} \right) \middle| 0 \leq n_{p}^{s} \leq n_{p}^{v}, 0 \leq r_{p} n_{p}^{v} \leq c_{p}, p \in \mathcal{P}; \sum_{p \in \mathcal{P}_{l}} r_{p} n_{p}^{v} \leq c_{l}, \quad l \in \mathcal{L} \right\}.$$
(2.3b)

The following additional notation is used in subsequent sections. In the single-rate case

$$n_l^v \stackrel{\Delta}{=} \sum_{p \in \mathcal{P}_l} n_p^v \tag{2.4a}$$

$$n_i^s \stackrel{\Delta}{=} \sum_{p \in \mathcal{P}_i} n_p^s \tag{2.5a}$$

denote the total number of calls and "silent" calls, respectively, that use link l; both n_l^v and n_l^a are clearly integers and satisfy the inequalities

$$0\leq n_l^{\bullet}\leq n_l^{v}\leq c_l.$$

The corresponding definitions in the multi-rate case are

$$k_l^{\upsilon} \stackrel{\Delta}{=} \sum_{p \in \mathcal{P}_l} r_p n_p^{\upsilon} \tag{2.4b}$$

$$k_l^s \stackrel{\Delta}{=} \sum_{p \in \mathcal{P}_l} r_p n_p^s \tag{2.5b}$$

where k_l^v and k_l^s now denote the total number of channels occupied by all calls and by the inactive (silent) calls, respectively, on link *l*; clearly these must satisfy

$$0\leq k_l^s\leq k_l^v\leq c_l.$$

Since, the voice data rates r_p may not be integers, k_l^v and k_l^s may assume non-integer values as well.

Moreover, n_l^d (the number of data messages at link l) follows an $M/D/c'_l$ model with residual data capacity

$$c'_{l} = c_{l} - (n^{v}_{l} - n^{s}_{l})$$
(2.6a)

for the model of (2.3a) or

$$c'_{l} = c_{l} - (k^{v}_{l} - k^{s}_{l})$$
(2.6b)

for the model of (2.3b).

Finally, since F and μ appear in the form $\rho = F/\mu$ in all the formulas derived hereafter, we can simply use the variable

$$\rho_p^v = F_p^v / \mu_p^v \tag{2.7}$$

$$\rho_l^d = F_l^d / \mu_l^d \tag{2.8}$$

$$\rho_l^v = \sum_{p \in \mathcal{P}_l} \rho_p^v \tag{2.9}$$

$$\rho_{nm}^v = F_{nm}^d / \mu_{nm}^v \tag{2.10}$$

$$\rho_{nm}^d = F_{nm}^d / \mu_{nm}^d \tag{2.11}$$

and

$$\rho_{nm}^{v} = \sum_{p \in \mathcal{P}_{nm}} \rho_{p}^{v}. \qquad (2.12)$$

2.4.2 Single-Rate Multi-Hop Radio Network

In this model, the link capacities c_l $(l \in \mathcal{L})$ are replaced by the number of transceivers at node n, T_n $(n \in \mathcal{N})$. The key sets of paths \mathcal{P}_l $(l \in \mathcal{L})$ are replaced by the sets \mathcal{P}_n , $n \in \mathcal{N}$. Now the number of calls n_p^v of class p (along path $p, p \in \mathcal{P}_n$) satisfies the constraints

$$\sum_{p \in \mathcal{P}_n} n_p^v \le T_n, \quad n \in \mathcal{N}$$
(2.3c)

instead of (2.3a) or (2.3b), and the residual data capacity of link l = (n, m) connecting nodes n and m is

$$c'_{l} = \min\left\{T_{n} - \sum_{p \in \mathcal{P}_{n}} (n_{p}^{v} - n_{p}^{s}), T_{m} - \sum_{p \in \mathcal{P}_{m}} (n_{p}^{v} - n_{p}^{s})\right\}, \ l \in \mathcal{L}, \ n, m \in \mathcal{N}$$
(2.6c)

This formulation implicitly assumes that all transceivers at nodes n and m that are not currently supporting active voice calls are available to support data traffic between nodes n and m. Thus it neglects the possibility of data traffic between either of these nodes and their other neighbors. Therefore, Eq. (2.6c) should be accepted with reservation and only as representative of the simplest data model that we can handle with our approach.

In a more realistic scenario the performance measure for the data operating under a particular protocol over link l = (n, m) (connecting nodes n and m) depends not only the T_n and T_m of these nodes and the total number of active voice calls $\sum_{p \in \mathcal{P}_n} (n_p^v - n_p^s)$ and $\sum_{p \in \mathcal{P}_m} (n_p^v - n_p^s)$ but also on the corresponding values of these quantities for all the neighbors of the nodes n and m. Thus the residual capacity l = (n, m) depends on the way in which the unused transceivers at nodes n and m are allocated to support data links with their other neighbors as well.

To determine this residual capacity one must very carefully enumerate all possible pairs of transceivers that are occupied by voice traffic for each voice state, and then use a (not yet defined) protocol to determine how the unused transceivers are to be allocated to support data traffic between the pairs of users. Typically, a transceiver may be paired with several of its neighbors (one at a time) to form a link-activation schedule, in which case the data-traffic queueing model will have to be revised to reflect the fact that the server is not always available (as in a queueing system with vacations). As long as this allocation depends only on the voice state (i.e., not on the data queue sizes at each node), our formulation that addresses the data state at each link in isolation remains valid.

Once this allocation is made, the performance measure of interest can be thought as a functional which depends on several terms of the form $\sum_{p \in \mathcal{P}_n} (n_p^v - n_p^s)$ where \hat{n} is an immediate neighbor of n or m. If the number of neighbors of the nodes n and m is small then we can use the approach of Section 9.2 (based on additional conditioning) to average the performance measure for data with respect to the steady state distribution of the voice state vector. If the number of neighbors of n and m is larger, we may use the approach of Section 9.3 (based on fictitious links) to carry out this averaging.

In this report the M/D/c model of (2.6c) is sufficient for establishing the accuracy and computational efficiency of the knapsack and Pascal approximation techniques for a first application to integrated voice/data radio networks. In the future we will apply these techniques to more complicated data protocols.

3. STEADY-STATE DISTRIBUTION FOR THE SYSTEM STATE

In this section we provide the steady-state distribution of the state of the network, that is of $(\underline{N}^{v}, \underline{N}^{a}, n_{l}^{d})$, for the single-rate and multi-rate cases under the assumption that the the state of voice calls changes much slower than the state of data.

Proposition 3.1. When the state of voice calls changes much slower than the data-packet state, the steady state probability of $(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d})$ can be closely approximated by

$$P(\underline{N}^{v}, \underline{N}^{s}, n_{l}^{d}) = P(\underline{N}^{v}, \underline{N}^{s}) \cdot P(n_{l}^{d} | \underline{N}^{v}, \underline{N}^{s})$$
(3.1)

where

(a) the steady state probability of the voice is

$$P(\underline{N^{v}}, \underline{N^{s}}) = \frac{1}{G} \cdot \prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{s!}(n_{p}^{v} - n_{p}^{s})!} \frac{\underline{\alpha}^{n_{p}^{s}} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v}}}$$
(3.2)

and

$$G = \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega^{v}} \prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{s!}(n_{p}^{v} - n_{p}^{s})!} \frac{\alpha^{n_{p}^{s}} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v}}}$$
(3.3)

for

$$\Omega^{v} = \begin{cases} \left\{ (\underline{N}^{v}, \underline{N}^{s}) \middle| & 0 \le n_{p}^{s} \le n_{p}^{v}, p \in \mathcal{P}; 0 \le n_{l}^{v} \le c_{l}, \quad l \in \mathcal{L} \end{cases} & \text{single-rate case} \\ \left\{ (\underline{N}^{v}, \underline{N}^{s}) \middle| & 0 \le n_{p}^{s} \le n_{p}^{v}, p \in \mathcal{P}; 0 \le k_{l}^{v} \le c_{l}, \quad l \in \mathcal{L} \end{cases} & \text{multi-rate case} \end{cases}$$

$$(3.4)$$

(b) the conditional steady state probability of the data for an M/M/c model is

$$P(n_{l}^{d}|\underline{N}^{v},\underline{N}^{s}) = \begin{cases} P_{0}\frac{(\rho_{l}^{d})^{n_{l}^{d}}}{n_{l}^{d}!}, & n_{l}^{d} \leq c_{l}' \\ P_{0}\frac{(\rho_{l}^{d})^{n_{l}^{d}}c_{l}'c_{l}'}{c_{l}'!c_{l}'^{n_{l}^{d}}}, & n_{l}^{d} > c_{l}' \end{cases}$$
(3.5)

where

$$P_{0} = \left[\sum_{n=0}^{c_{l}^{\prime}-1} \frac{\left(\rho_{l}^{d}\right)^{n}}{n!} + \frac{\left(\rho_{l}^{d}\right)^{c_{l}^{\prime}}}{c_{l}^{\prime}!\left(1 - \frac{\rho_{l}^{d}}{c_{l}^{\prime}}\right)}\right]^{-1},$$
(3.6)

 $\rho_p^v = F_l^p / \mu_p^v, \ \rho_l^d = F_l^d / \mu_l^d, \ \text{and}$

$$c'_{l} = \begin{cases} c_{l} - n^{v}_{l} + n^{s}_{l} \ge 1 & \text{single} - \text{rate case} \\ c_{l} - k^{v}_{l} + k^{s}_{l} \ge 1 & \text{multi} - \text{rate case} \end{cases}$$
(3.7)

Note that

$$P(n_l^d = \infty | \underline{N}^v, \underline{N}^s) = 1 \text{ when } c_l' = 0, \text{ or } c_l' \ge 1 \text{ but } \rho_l^d / c_l' \ge 1$$
(3.8)

i.e., the data-packet queue becomes infinite when there is no residual capacity for data or when the offered data load of the link exceeds the residual link capacity.

(c) the conditional steady state probability of the data for an M/D/c model can not be obtained in closed form but can be approximated with arbitrary accuracy using Tijms' iterative algorithm (see [13]) on a M/D/c'_l system (recall $c'_l = c_l - k^v_l + k^s_l$); the procedure is sketched in Appendix B.

Proof: The proof of the main result (a) is provided in Appendix A.

Comments:

1. Note that the expression for $P(\underline{N}^{v}, \underline{N}^{s})$ in Eq. (3.2) is a modified version of the traditional product form for the steady state probability of the voice state. In the more frequently studied case, in which the occurrence of silent periods is not addressed, we have (see for example [4] or [6])

$$P(\underline{N}^{\nu}) = \frac{1}{G} \cdot \prod_{p \in \mathcal{P}} \frac{(\rho_p^{\nu})^{n_p^{\nu}}}{n_p^{\nu}!}$$
(3.9)

where

$$G = \sum_{\underline{N}^{v} \in \Omega^{v}} \prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{v!}}$$
(3.10)

is the normalization constant associated with the state space

$$\Omega^{v} = \left\{ \underline{N}^{v} \middle| \quad 0 \le n_{p}^{v} p \in \mathcal{P}; 0 \le \sum_{p \in \mathcal{P}_{l}} r_{p} n_{p}^{v} \le c_{l}, \quad l \in \mathcal{L} \right\}.$$
(3.11)

The reason that we need $P(\underline{N}^{v}, \underline{N}^{s})$ instead of the usual $P(\underline{N}^{v})$ is that we assume that silence periods of voice calls can be detected and the data traffic can use the released bandwidth, thus increasing the overall efficiency of the resource allocation.

2. Under the M/M/c data model and a single-rate network, $(n_l^d | \underline{N}^v, \underline{N}^s)$ corresponds to an $M/M/c_l - n_l^v + n_l^s$ system when $c_l' = c_l - n_l^v + n_l^s \ge 1$ and $\rho_l^d/c_l' < 1$, whereas $Pr(n_l^d = \infty | \underline{N}^v, \underline{N}^s) = 1$ when $c_l' = 0$ or $c_l' \ge 1$ but $\rho_l^d/c_l' \ge 1$. The same is valid for $(n_l^d | \underline{N}^v, \underline{N}^s)$ under an $M/D/c_l - n_l^v + n_l^s$ data model. For the multi-rate case we only need to replace n_l^v and n_l^s by k_l^v and k_l^s in the above comments.

3. The probability of voice blocking at link l of a call of class $p(B_{lp})$ can be expressed ([4]-[6]) as the ratio of two normalization constants (also termed partition functions) G defined in (3.10); specifically, the probability that such blocking does not occur is

$$1 - B_{lp} = \frac{G(c_1, c_2, \cdots, c_{l-1}, c_l - r_p, c_{l+1}, \cdots, c_{\mathcal{L}})}{G(c_1, c_2, \cdots, c_{l-1}, c_l, c_{l+1}, \cdots, c_{\mathcal{L}})}$$
(3.12a)

where the normalization constants G of (3.10) are denoted as functions of the $1 \times |\mathcal{L}|$ row vector of link capacities <u>c</u>. The right-hand side of (3.12a) represents the steadystate probability of the vector of the number of voice calls over all paths of the network, when a single (voice) channel is removed from link l (and thus its capacity decreases to $c_l - r_p$). This is indeed the probability of no blocking because a new arriving call (requiring bandwidth r_p) can be accommodated by the link capacity. The evaluation of B_{lp} in (3.12a) requires the computation of the normalization constant G of (3.10). This may be a very computationally demanding task even for moderate size networks, because of the large number of points in the state space over which the summation is performed. Therefore, accurate and computationally efficient approximations are necessary. Several such approximations have been developed under different conditions (limiting regimes) (see [4]-[10]). In Section 4 we review some of them and select to work with two: the knapsack and the Pascal approximations, which we extend and modify (in Sections 5 and 6) so that they apply to performance measures pertaining to the data such as the average probability of queueing and the queueing delay of data at the links of the network.

Finally, B_p , the blocking probability for a call of type $p \ (p \in \mathcal{P}, [4]-[6])$, can be evaluated from the expression

$$1 - B_p = \frac{G(\underline{c} - r_p \underline{A}_p^T)}{G(\underline{c})}$$
(3.12b)

where <u>c</u> was defined above and \underline{A}_p^T is the transpose of the p-th column (a $|\mathcal{L}| \times 1$ vector) routing matrix **A** defined at the beginning of Section 2. Equation (3.12b) is similar to that

in (3.12a) with one major difference; in (3.12a) only the capacity of link l has been reduced by r_p (corresponding to availability of channel capacity along link l for an additional voice call) in the argument of the normalization constant G in the numerator; in (3.12b) the capacities of all links l used by path p ($l \in p$) have been reduced by r_p (corresponding to availability of channel capacity along all links l along path p for an additional voice call). The expression in (3.12b) requires about the same level of computational complexity as that of (3.12a); this becomes prohibitive for even moderate size networks. However, the approximations obtained for B_{lp} in Section 4 can also be used (with proper modification) for B_p as well.

4. The probability of data queueing at link l is given by

$$Q_{l} = Pr(\text{data queued in link } l)$$

$$= Pr(c_{l} - n_{l}^{v} + n_{l}^{s} \le n_{l}^{d})$$

$$= 1 - Pr(c_{l} - n_{l}^{v} + n_{l}^{s} > n_{l}^{d} \text{ where } (n_{l}^{v}, n_{l}^{s}) \ne (c_{l}, 0))$$

$$= 1 - \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega^{v}} \left[\sum_{n_{l}^{d} = 0}^{c_{l}^{\prime} - 1} I[(n_{l}^{v}, n_{l}^{s}) \ne (c_{l}, 0)] P(n_{l}^{d} | \underline{N}^{v}, \underline{N}^{s}) \right] P(\underline{N}^{v}, \underline{N}^{s}) \quad (3.13)$$

for a single-rate system; for a multi-rate system we must only replace the n_i^v and n_i^s by k_i^v and k_i^s defined above, respectively, in the above equation; I denotes the indicator function taking value 1 when its argument is true and 0 if it is false. Consequently, Q_i can be evaluated directly from the conditional steady state probabilities of the data $P(n_i^d | \underline{N}^v, \underline{N}^s)$ and the steady state probability of the voice $P(\underline{N}^v, \underline{N}^s)$. However, the computational complexity of the summation involved in the definition of $P(\underline{N}^v, \underline{N}^s)$ [see (a) of Proposition 3.1] coupled with the additional summation necessary for obtaining Q_i becomes prohibitive for large $|\mathcal{P}|$. This computation is carried out through computationally efficient approximations in Sections 5 and 6.

5. The queueing delay of data at link l can also be evaluated (actually approximated) using the basic results of this section. The details are provided in Section 7.

6. The results of this section were derived for a general multi-rate multi-hop wired network; they are valid for the single-rate multi-hop radio network of Sections 2.1.2 and 2.2.2 after we replace c_l $(l \in \mathcal{L})$ with T_n $(n \in \mathcal{N})$, \mathcal{P}_l with \mathcal{P}_n , and set $r_p = 1$ for all $p \in \mathcal{P}$.

4. REDUCED-LOAD APPROXIMATIONS FOR THE VOICE BLOCKING PROBABILITY

As discussed in the comments following Proposition 3.1 in Section 3, the evaluation of closed form expressions for the probability of voice blocking at link l or along path p require substantial computer resources and time. In particular, when these expressions must be computed numerous times (i.e., for different control policies), as is the case in a variety of optimization problems including admission control and routing, it becomes imperative that computationally efficient approximations are developed that exhibit satisfactory accuracy and allow the speedy evaluation of these quantities.

In this section we describe a number of approximations that can be used to evaluate the probability of voice blocking with computational complexity considerably lower than that of the brute force approach. These are termed "reduced-load" approximations, and have been applied to multi-rate loss networks by a number of researchers [4]-[10] with very encouraging results. The term reduced-load pertains to a reduced- (or thinned-) load approximation of the traffic from all paths using a particular link, because of traffic blocked at other links, and is discussed in detail in Section 4.2. The approximations are known to be asymptotically correct in the limiting regime characterized by heavy offered traffic loads and large link capacities. These approximations had not been applied to data traffic analysis before our work in this report.

In the context of the advancement of this approximation theory and its applications to practical multi-media network problems, our main contribution elaborated upon in this report is fourfold. First, we establish that two of these approximation techniques (knapsack and Pascal) exhibit satisfactory accuracy even when applied to situations different from those of the limiting regime; we actually show that they maintain excellent accuracy over the entire range of useful traffic scenarios and architectures. Second, we show how to use these approximations to evaluate performance measures for data (such as the probability of queueing for data and the average queueing delay for data) in networks with multi-media traffic. Third, we extend the application of these approximations to radio network models that are distinctly different from the multi-media networks of [4]-[10], which use optical fiber or copper as the transmission medium. Fourth, we apply these techniques to systems with admission control schemes and to the subsequent optimization of the thresholds or other control parameters involved. The purpose of reviewing these approximation techniques here is twofold. First, as we show in Section 11 through comparisons with results based on the exact expressions and the Monte-Carlo summation method, these techniques have satisfactory accuracy and we can use them to approximate the average voice blocking probability of links or paths (routes) for radio networks of interest to our project. Second, we need to introduce, motivate, and describe the fundamental principles of these techniques before we extend and modify them in order to approximate the average probabilities of queueing and the average waiting delays of data at the links of the network. These extensions are described in detail in Sections 5, 6, and 7, and enable us to approximate accurately and with reasonable computational complexity important performance measures for the data in multi-media networks.

In this section we describe the principle of four of these approximation techniques: Kelly's, Knapsack, Pascal, and Mitra's approximations. We limited our consideration to these four for two reasons. First, these four are the ones that have been applied most successfully to a variety of networking problems with very satisfactory results. Second, we were able to extend and modify two of those (the knapsack and Pascal approximations) to accommodate performance measures for the data (such as average probability of queueing and average queueing delay). Kelly's approximation (4]-5) was included in this review because it is a useful starting point for introducing the notation and the principle of the reduced load approximations. Mitra's approximation was included because we considered it early during the course of this work, we extended its application to more general topologies than the tree network of the original paper [10], and we extended and modified it so that it applies also to the probability of queueing and the queueing delay of data. Unfortunately, as we found out through comparisons with the exact expressions and the Monte-Carlo summation method, this approximation is accurate (actually converges) only under rather restrictive assumptions about the loads of the links and thus it is not presented in greater detail in this work.

4.1 Kelly's Approximation

Consider a network supporting multi-rate traffic (say with data rate r_p for path $p \in \mathcal{P}$). Denote by L_l the approximate probability that "all circuits are busy on the link l", or equivalently the probability of blocking for link l ($l \in \mathcal{L}$). Under the assumption ([4]-[5]) that these events occur independently from link to link, class p connections

(voice calls using path p) arrive to link l according to a Poisson process with offered load $r_{pi} \prod_{\ell \in p, \ell \neq l} (1 - L_{\ell})^{r_p}$ and thus the total arriving traffic at link l (belonging to all classes $p \in \mathcal{P}$) is also Poisson with aggregate load

$$\sum_{\boldsymbol{p}\in\mathcal{P}_l} r_{\boldsymbol{p}} \rho_{\boldsymbol{p}}^{\boldsymbol{v}} \prod_{\boldsymbol{\ell}\in\boldsymbol{p}, \boldsymbol{\ell}\neq l} (1-L_{\boldsymbol{\ell}})^{r_{\boldsymbol{p}}}.$$
(4.1)

where the expression $\prod_{\ell \in p, \ell \neq l} (1 - L_{\ell})^{r_p}$ represents the "thinning" or "reduced load" effect associated with the blockage of calls of type p at the other links along the path. Hence, under the link independence assumption, we must have

$$L_{l} = E\left[c_{l}; \sum_{p \in \mathcal{P}_{l}} r_{p} \rho_{p}^{v} \prod_{\ell \in p, \ell \neq l} (1 - L_{\ell})^{r_{p}}\right], \quad l \in \mathcal{L}$$

$$(4.2)$$

where

$$E[c;\rho] = \frac{\rho^c/c!}{\sum_{n=0}^{c} \rho^n/n!}$$
(4.3)

is the Erlang loss formula (see [1] or [4]). In the above notation we used L_l instead of B_l to distinguish between the exact value of the voice blocking probability and the reduced-load approximation L_l . If $r_p = r$ for all $p \in \mathcal{P}$, then (4.2) becomes the standard reduced load approximation for single-rate loss networks.

Repeated substitutions are often used for finding a solution $(L_1, L_2, \dots, L_{|\mathcal{L}|})$ to the fixed-point equation (4.2). Although oscillation can occur in (4.2), repeated substitutions typically converge to a fixed point for networks of practical interest. One of the features of this approximation scheme is that (4.2) has a unique fixed-point solution. The proof of uniqueness relies on the monotonicity properties of the Erlang loss formula; unfortunately, these properties are not possessed by the knapsack and Pascal approximation schemes discussed below.

Once a fixed point is found, the probability B_p that a call is blocked along path p (or equivalently a class-p connection is blocked as we saw in the discussion of Section 2.2) can be approximated by

$$B_{p} = 1 - \prod_{l \in p} (1 - L_{l})^{r_{p}}, \qquad (4.4)$$

a formula that invokes again the link independence assumption.

The above approximation is asymptotically accurate (correct) under the following limiting regime: the ratio of offered load and of the number of channels in each link is held fixed while the individual values of these two quantities become asymptotically large.

4.2 Knapsack Approximation

For this approximation scheme as well as the one of the following section we work first with a single-link multi-rate system and then extend the result to multi-link networks.

This technique is termed the knapsack approximation because the single-link multirate system corresponds to a *stochastic knapsack* resembling the knapsack model in combinatorial optimization. The term stochastic knapsack is motivated by the fact that typically the system modeled resembles a knapsack to which items (states) are added or from which items are taken out according to a probability distribution. This approximation was successfully applied to circuit-switching problems by Ross [6]-[9].

Consider a single-link system with link capacity c_l , which supports classes $p \in \mathcal{P}_l$ (or equivalently several paths p use link l) with data rates r_p and offered loads ρ_p . The probability that a class-p connection is blocked (or a voice call along path p is blocked), when arriving to the stochastic knapsack, is given by

$$K_{lp}[c_l; \rho_q^v, q \in \mathcal{P}_l] = 1 - \frac{\sum_{n=0}^{c_l - r_p} w(n)}{\sum_{n=0}^{c_l} w(n)}, \quad p \in \mathcal{P}_l$$
(4.5)

where

$$w(n) = \frac{1}{n} \sum_{q \in \mathcal{P}_l} r_q \rho_q^v w(n - r_q), \quad n = 1, 2..., c_l$$
(4.6a)

with initial condition

$$w(0) = 1.$$
 (4.6b)

The intuitive explanation of (4.5) is that $1 - K_{lp}$ represents the probability of no blocking, i.e., the probability that up to $c_l - r_p$ $(l \in p)$ voice calls are in the system (inside the stochastic knapsack) so that an arriving call (requiring r_p channels) can be accommodated by the capacity of link *l* without being blocked. In (4.5) w(n) provides the probability that *n* voice calls are currently in the system, and the recursion (4.6a) for the update of w(n)is typical of stochastic knapsack system models.

For a multi-rate loss network with **multiple links**, we denote by L_{lp} the approximate probability that "less than r_p channels are available on link l" (probability of voice blocking

on link l along path p); L_{lp} is an approximation to the quantity B_{lp} defined by (3.12a) in Section 3. As in Kelly's approximation, we again assume here that these events occur independently from link to link. This approximation decouples the blocking phenomena on different links and enables the evaluation of an approximate expression for the amount of unblocked traffic traveling through the network and finally for the desired blocking probability. Under the interlink blocking independence assumption, class-q connections arrive to link l according to a Poisson process with offered load

$$\rho_q^v \prod_{\ell \in q, \ell \neq l} (1 - L_{\ell q}). \tag{4.7}$$

This load is termed thinned or reduced, because it is smaller than the corresponding load ρ_q^v (from circuit p) of the link l (for $l \in q$), when considered in isolation, in a manner that reflects the effect of blocking at the other links in the network through the approximate probabilities $L_{\ell q}$. This aspect of the approach gives the name reduced-load approximation to the knapsack approximation method (as well as the Pascal approximation method described in Section 4.3). Consequently, under the link independence assumption, we have

$$L_{lp} = K_{lp}[c_l; \rho_q^v \prod_{\ell \in q, \ell \neq l} (1 - L_{\ell q}), q \in \mathcal{P}_l], \quad p \in \mathcal{P}_l, \ l \in \mathcal{L}.$$

$$(4.8)$$

Equations (4.8) define a continuous mapping from the compact convex set $[0,1]^{|\mathcal{P}_1|\times|\mathcal{P}_2|\times\cdots\times|\mathcal{P}_{l_c_1}|}$ into itself; thus, by the Brouwer fixed-point theorem, there exists a solution $(L_{lp}, p \in \mathcal{P}_l, l = 1, 2, \cdots, |\mathcal{L}|)$ to (4.8). The method of successive approximations (i.e., repeated substitutions) can be employed to find such a solution. Once a solution to (4.8) has been obtained, the probability of blocking a class-*p* connection can be approximated by

$$B_p = 1 - \prod_{l \in p} (1 - L_{lp})$$
(4.9)

and the probability of blocking any voice call on link l can be approximated by

$$B_l = \sum_{p \in \mathcal{P}_l} L_{lp}.$$
 (4.10)

In contrast to the fixed-point equation (4.2) the solution to (4.8) of the knapsack approximation is not necessarily unique. This is because the knapsack equations (4.5)-(4.6)

do not have the nice monotonicity properties of the Erlang formula (4.3). Multiple solutions to the fixed-point equations of (4.8) can alert the designer of potential instabilities in the network. Ross [6] provides an example where the network alternates between long periods of carrying only narrowband connections and long periods of carrying only wideband connections. Kelly's approximation does not expose this instability, since it always gives rise to a unique fixed point. But the knapsack approximation gives one solution with almost 100% blocking of narrowband connections and another solution with almost 100% blocking of wideband calls, which reflects the instability that actually exists in the system.

The knapsack approximation is shown in [6] and [8] to be asymptotically correct under the same limiting regime as Kelly's approximation (see end of Section 4.1). However, comparison of the accuracies of the two approximations, the exact expressions, and the Monte-Carlo summation method indicate that the knapsack approximation maintains satisfactory accuracy even far outside the limiting regime.

4.3 Pascal Approximation

The Pascal approximation technique is based on the use of a birth-death process with Pascal distribution to model the voice state in the system. This was applied to circuit-switched problems by Ross [6]-[9].

We again address first the case of a single link. Consider a birth-death process on the state space $0, 1, \dots, c_l$, which, when in state *n*, has a death rate of *n* and a birth rate of $\epsilon^2/\sigma^2 + n(1 - \epsilon/\sigma^2)$, where ϵ and σ^2 are given positive numbers. Let q(n), for $n = 0, 1, \dots, c_l$, be the equilibrium probability of being in state *n*, that is, q(n) satisfies

$$nq(n) = [\epsilon^2/\sigma^2 + (n-1)(1-\epsilon/\sigma^2)]q(n-1), \text{ for } n = 1, 2, \cdots, c_l$$
 (4.11a)

where

$$\sum_{n=0}^{c_l} q(n) = 1. \tag{4.12a}$$

Denote

$$P_{lp}(c_l;\epsilon;\sigma^2) = \sum_{n=c_l-r_p+1}^{c_l} q(n) = 1 - \sum_{n=0}^{c_l-r_p} q(n).$$
(4.13a)

The right-hand side of (4.13a) represents the probability that the birth-death process n is in a state $> c_l - r_p$. This corresponds to blocking since there is no room left in the capacity of link l for accommodating any arriving voice call (which requires r_p channels).

When $c_l = \infty$, q(n) has the Pascal distribution, and the means and variance of the birthdeath process are given by ϵ and σ^2 , respectively (see [6]). The Pascal approximation uses the same approach but for finite values of c_l necessitating the solution of Eq. (4.11a), as described below.

If instead of q(n) we use the normalized version

$$q'(n) = q(n)/q(0)$$

the recursion of (4.11a) becomes

$$q'(n) = \frac{1}{n} [\epsilon^2 / \sigma^2 + (n-1)(1-\epsilon / \sigma^2)] q'(n-1), \text{ for } n = 1, 2, \cdots, c_i$$
(4.11b)

with initial condition

$$q'(0) = 1.$$
 (4.12b)

This is easier to evaluate [it does not require knowledge of q(0)], and results in

$$\left[\sum_{n=0}^{c_l}q'(n)\right]^{-1}=q(0).$$

Using this result to modify (4.13a) yields

$$P_{lp}(c_l;\epsilon;\sigma^2) = 1 - \frac{\sum_{n=0}^{c_l-r_p} q'(n)}{\sum_{n=0}^{c_l} q'(n)}.$$
(4.13b)

Next consider the stochastic knapsack model, discussed in the previous subsection, for the isolated link l. When $c_l = \infty$, the mean and variance of the number of busy circuits [under the knapsack approximation for the steady-state voice distribution] are obtained after some manipulations by

$$\epsilon_l = \sum_{q \in \mathcal{P}_l} r_q \left[\sum_{n=0}^{\infty} n P(n_q^v = n) \right] = \sum_{q \in \mathcal{P}_l} r_q \rho_q^v \tag{4.14}$$

and

$$\sigma_l^2 = \sum_{q \in \mathcal{P}_l} r_q^2 \left[\sum_{n=0}^{\infty} (n - \rho_q^v)^2 P(n_q^v = n) \right] = \sum_{q \in \mathcal{P}_l} r_q^2 \rho_q^v$$
(4.15)

respectively. Thus, the infinite-capacity stochastic knapsack gives the same mean and variance for the number of busy circuits as the birth-death process with parameters $\epsilon = \epsilon_l$ and $\sigma^2 = \sigma_l^2$. It is, therefore, natural to approximate the probability of blocking a classp connection arriving to the stochastic knapsack with finite capacity (i.e., $c_l < \infty$) by $P_{lp}[c_l; \epsilon_l; \sigma_l^2]$.

For the multi-rate network with multiple links we use the same notation as in Section 4.2 and make again the link independence assumption, so that class-q connections arrive to link l according to a Poisson process with the rate given by (4.7). By invoking the Pascal approximation, we obtain the probability that the capacity of link l is not available for a call of class p as

$$\hat{L}_{lp} = P_{lp} \left[c_l; \sum_{q \in \mathcal{P}_l} r_q \rho_q^v \prod_{\ell \in q, \ell \neq l} (1 - \hat{L}_{\ell q}); \sum_{q \in \mathcal{P}_l} r_q^2 \rho_q^v \prod_{\ell \in q, \ell \neq l} (1 - L_{\ell q}) \right], \quad q \in \mathcal{P}_l, \quad l \in \mathcal{L}$$

$$(4.16)$$

where the function $P_{lp}(\dots, \dots, \dots)$ is given by (4.13) and we use \hat{L}_{lp} to denote the approximation to the probability of blocking of voice path p on link l for the Pascal method; the same quantity for the knapsack method was denoted by L_{lp} . In general $\hat{L}_{lp} \neq L_{lp}$; however, in the important single-rate case $(r_p = 1 \text{ for all } p \in \mathcal{P})$, we show in Appendix F that $\hat{L}_{lp} = L_{lp}$ $(p \in \mathcal{P}_l, l \in \mathcal{L})$.

As with the knapsack approximation of Section 4.2, there exists a nonunique solution to the fixed point equation (4.16). Once having found such a solution, B_p , the probability of blocking a class-*p* connection (or a voice call along path *p*) is approximated by (4.9); similarly, B_l , the probability of blocking any voice call on link *l* is approximated by (4.10).

The computational complexity and the accuracy of the Pascal approximation are very comparable with those of the knapsack approximation. Moreover, the Pascal approximation technique was shown in [6] to be asymptotically accurate (correct) under the same limiting regime as Kelly's and the knapsack approximations. Through our comparisons it has also been established that the Pascal approximation is accurate over a broader range of load parameters than Kelly's approximation.

4.4 Mitra's Approximation

The approximation technique suggested by Mitra [10] relies on a Taylor series expansion of the normalization constant of (3.10). Although it has the advantage that we can control its accuracy by increasing the number of terms in the Taylor series expansion, it has the disadvantage of converging only for a rather limited number of scenarios. Actually, we were not initially aware of that disadvantage, since [10] remains silent about it.

In [10], the approximation technique was applied to a tree network. However, in that paper there was not any stated constraint on the traffic loads and number of channels (of the circuits), without which convergence was unattainable. We simulated Mitra's results and found out that he only presented in his paper the scenarios for which convergence was attained. By increasing the loads we found out that the convergence was no longer guaranteed.

Consequently, although we were able to extend the applicability of Matra's approximation method (a) to more general topologies than trees, (b) from single-rate networks to multi-rate networks, and (c) to the probability of data queueing (besides the probability of voice blocking), the severely limited range of convergence precludes the application of this method in most cases. Since all our simulation results using this method showed poor accuracy (except for the results presented in Mitra's paper about the tree network and for only the traffic loads reported there), we opted not to present the mathematical details of the application of this approximation method even for our novel work (i.e., the extensions to general topologies and to multi-rate networks).

4.5 Knapsack and Pascal Approximations for the Radio Network Model of Section 2.2

In Appendix G we show that for networks with single-rate traffic the knapsack and Pascal methods provide identical approximations to the probabilities of voice blocking and data queueing. Consequently for the radio model of Sections 2.1.2 and 2.2.2, which is characterized by $r_p = 1$ for all paths $p \in \mathcal{P}$, we use only the knapsack approximation in deriving the numerical results for the voice blocking probability in Section 12.

The application of the knapsack approximation method to the evaluation of the probability of voice blocking for the radio model of Sections 2.1.2 and 2.2.2 is a straightforward modification of the approach described in detail in Section 4.2. In particular, we replace l(links) by n (nodes), \mathcal{L} by \mathcal{N} , \mathcal{P}_l (set of paths using link l) by \mathcal{P}_n (set of paths intersecting at node n), c_l (link capacities) by T_n (number of transceivers at nodes), and $r_p = 1$ (for all $p \in \mathcal{P}$). We are then able to use the results of subsection 4.3 directly.
5. KNAPSACK APPROXIMATION TO THE PROBABILITY OF DATA QUEUEING

In this section we describe in detail the application of the knapsack approximation to the probability of data queueing. This development appears for the first time in this report. The approximation is first developed for general multi-rate networks in Section 5.1 and then modified in Section 5.2 to be suitable for the multi-hop radio network of Sections 2.1.2 and 2.2.2.

5.1 General Multi-Hop Multi-Rate Networks

The key idea behind the application of the knapsack approximation to the probability of data queueing is to use the technique described in Section 4.2 to approximate the external sum in (3.13), while treating the internal sum (the sum with respect to n_l^d) as a single entity for given values of $(\underline{N}^v, \underline{N}^d)$ and use Proposition 3.1. This approach is described in the following in detail.

As for the case of voice traffic (Section 4.2) in order to derive the knapsack approximation to the probability of data queueing in link l, Q_l , we first consider the probability of queueing for a single-link multi-rate system. As we saw in the comments after Proposition 3.1 of Section 3, the probability of data queueing for link l can be expressed as in (3.13) which we may modify as

$$Q_{l} = 1 - \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega_{s}} \left[\sum_{n_{l}^{d} = 0}^{c_{l}^{\prime} - 1} I[(n_{l}^{v}, n_{l}^{s}) \neq (c_{l}, 0)] P(n_{l}^{d} | \underline{N}^{v}, \underline{N}^{s}) \right] P(\underline{N}^{v}, \underline{N}^{s})$$
$$= 1 - \sum_{k=0}^{c_{l}} \sum_{m=0}^{k} I[(k, m) \neq (c_{l}, 0)] \cdot \left[\sum_{n_{l}^{d} = 0}^{c_{l}^{\prime} - 1} P(n_{l}^{d} | c_{l}^{\prime}) \right] P(k, m)$$
(5.1)

where

$$\Omega(k,m) = \left\{ \left(\underline{N}^{v}, \underline{N}^{s}\right) \middle| 0 \le n_{p}^{s} \le n_{p}^{v}, \quad 0 \le r_{p}n_{p}^{v} \le c_{l}, \quad p \in \mathcal{P}_{l}; \sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{s} = m; \quad \sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{v} = k \right\}$$

$$(5.2)$$

while

$$P(k,m) = \sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(k,m)} P(\underline{N}^{v},\underline{N}^{s})$$
(5.3)

with

$$\sum_{k=0}^{c_{i}} \sum_{m=0}^{k} P(k,m) = 1$$
(5.4)

and

$$c_l' = c_l - k + m \tag{5.5}$$

is the number of channels available for data, where c_i is the total number of channels in the link, k is the number of voice calls in progress, m of which are presently in silence mode, and $P(n_l^d | \underline{N^v}, \underline{N^s})$ is the steady-state probability of an $M/M/c'_l$ or an $M/D/c'_l$ system. In (5.1) we simplified the notation for $P(n_l^d | \underline{N^v}, \underline{N^s})$ to $P(n_l^d | c'_l)$ because it depends on $(\underline{N^v}, \underline{N^s})$ only through c'_l ,

The parameters k and m in (5.1)-(5.5) represent the total bandwidth (channel capacity) occupied by the number of total ongoing voice calls and of voice calls in silent mode, respectively, on which the expression for queueing probability under the M/D/cdata model is conditioned. The importance of this conditioning should not be underestimated; the fact that we can obtain efficient recursive expressions for the probability distribution (mass function) of (k,m) (see Appendix C) enables us to perform the final averaging with respect to k and m and evaluate the unconditional expressions for the performance measures of interest.

Regarding the evaluation of

$$\bar{P}(c_l', \rho_l^d) = \sum_{n_l^d=0}^{c_l'-1} P(n_l^d | c_l')$$
(5.6)

which is a factor in Eq. (5.1), we proceed as follows depending on the model used for the data traffic

(a) For an $M/M/c'_l$ data model, $\bar{P}(c'_l, \rho^d_l)$ is given by a well known formula [1] (also see

 $1 - \Pi_w$ in Appendix B.1), which can be put in the recursive form

$$\bar{P}(c_l', \rho_l^d) = \frac{1}{1 + \left(\frac{c_l' - \rho_l^d}{\rho_l^d}\right) \cdot \frac{c_l' - 1 - \rho_l^d \cdot \bar{P}(c_l' - 1, \rho_l^d)}{(c_l' - 1 - \rho_l^d) \cdot \bar{P}(c_l' - 1, \rho_l^d)}}$$
(5.7*a*)

$$\bar{P}(0,\rho_l^d) = 1 \tag{5.7b}$$

Consequently, $\tilde{P}(c'_l, \rho^d_l)$ can be evaluated iteratively.

(b) For an $M/D/c'_l$ data model, rather than evaluating $\bar{P}(c'_l, \rho^d)$ directly, we evaluate $P(n_l^d|c'_l)$ iteratively via a special-purpose overrelaxation method (see Tijms [13] and Appendix B), and sum up the resulting $P(n_l^d|c'_l)$ to obtain $\bar{P}(c'_l, \rho^d_l)$.

The quantity in (5.6) represents the total probability that the residual data capacity c'_l of the *l*-th link [given by (5.5), which depends on the number of active voice calls] is not exceeded by the number of data messages using link *l*. Thus, it corresponds to the conditional probability of no queueing of data (or alternatively one minus the probability of finding the system totally occupied) when conditioned on the voice state.

In order to evaluate P(k,m) we define

$$P'(k,m) = P(k,m)/P(0,0)$$
(5.8a)

as the normalized version of P(k, m). This results in

$$\left[\sum_{k=0}^{c_l}\sum_{m=0}^k P'(k,m)\right]^{-1} = P(0,0).$$
 (5.8b)

Then P'(k,m) can be obtained iteratively from Proposition 5.1 that follows, and the probability of data queueing Q_l for link l (and a single-link network) can be approximated by

$$Q_{l}(c_{l};\rho_{p}^{v},p\in\mathcal{P}_{l}) = 1 - \sum_{k=0}^{c_{l}}\sum_{m=0}^{k}I[(k,m)\neq(c_{l},0)]\cdot\bar{P}(c_{l}',\rho_{l}^{d})P(k,m)$$
$$= 1 - \frac{\sum_{k=0}^{c_{l}}\sum_{m=0}^{k}I(c_{l}'\neq0)\cdot\bar{P}(c_{l}',\rho_{l}^{d})\cdot P'(k,m)}{\sum_{k=0}^{c_{l}}\sum_{m=0}^{k}P'(k,m)}.$$
(5.9)

In (5.9) we used the notation $Q_l(c_l; \rho_p^v, p \in \mathcal{P}_l)$ in order to emphasize the dependence of the probability of data queueing on the voice loads $(\rho_p^v, p \in \mathcal{P}_l)$. The importance of this notation will become clear below when the approximation for a multi-link network is obtained.

Proposition 5.1

P'(k,m) satisfies

$$P'(k,m) = \begin{cases} \frac{1}{m} \frac{\alpha}{\alpha+\beta} \sum_{p \in \mathcal{P}_l} r_p \rho_p^v P'(k-r_p,m-r_p), & \text{if } r_p \leq m \leq k \leq c_l \\\\ \frac{1}{k-m} \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_l} r_p \rho_p^v P'(k-r_p,m), & \text{if } 0 \leq m < r_p \leq k \leq c_l \\\\ 1, & \text{if } k = 0, m = 0 \\\\ 0, & \text{if } k \text{ or } m \text{ are not positive multiples of } r_p \end{cases}$$

$$P'(0,0) = 1 \tag{5.10b}$$

integer

(5.10a)

$$P'(k,m) = 0 \text{ if } \begin{cases} m > k, \text{ or } k < 0, \text{ or } m < 0, \\ \text{or if } (k,m) \text{ can not be represented as the linear combination of } r_p, p \in \mathcal{P}_k \end{cases}$$

$$(5.10c)$$

Proof: It is provided in Appendix C.

Finally, to evaluate the probability of data queueing for a multi-link network we proceed as follows. We use the thinned voice load (under the interlink independence assumption)

$$[\rho_p^{v}]' = \rho_p^{v} \prod_{\ell \in p, \ell \neq l} (1 - L_{\ell p})$$
(5.11)

offered to link l from voice path p in place of the original ρ_p^v in the functional form of (5.9). The resulting approximation to the probability of queueing at link l (and a multi-link network) is

$$Q_{l}\left(c_{l};\rho_{p}^{\nu}\prod_{\ell\in p,\ell\neq l}(1-L_{\ell p}),p\in\mathcal{P}_{l}\right)$$
(5.12)

where $L_{\ell p}$ is the probability of blocking voice calls taking path p at link ℓ , which is obtained via the knapsack approximation on the voice part (Section 4.2).

A comment is in order here regarding the theoretical accuracy of the approximation; the practical accuracy is very satisfactory as demonstrated in Section 11 where the various approximations are compared with the results of accurate Monte-Carlo summation. Assuming that the computation of the data portion $\bar{P}(c_l, \rho_l^d)$ of (5.1) is a curate, we claim that the same limiting regime (i.e., when both voice load and number of channels in each link are large but their ratio is held fixed) that yields the probability of voice blocking B_l asymptotically accurate also yields the probability of data queueing Q_l asymptotically accurate. Our justification is at this point only a conjecture (it has not been proved mathematically) and is based on the facts that (a) the same inter-link voice independence assumption was used as in the voice blocking probability calculation, and (b) in the evaluation of Q_l , the knapsack approximation was applied to the voice portion (i.e., to determine the expected residual data capacity) for which it has already been shown to be accurate. We anticipate to be able to establish this claim rigorously in the future.

5.2 Radio Network Model of Sections 2.1.2, 2.2.2, and 2.4.2

As discussed in Sections 2.1.2 and 2.2.2 the radio network model considered in this report assumes that the data traffic over link $l = (n_1, n_2) \in \mathcal{L}$ connecting the network nodes $n_1, n_2 \in \mathcal{N}$ follows an $M/D/c'_l$ system model independently from the other links of the network. The residual capacity available for data is given by

$$c'_{l} = \min\left\{T_{n_{1}} - \sum_{p \in \mathcal{P}_{n_{1}}} (n_{p}^{v} - n_{p}^{s}), T_{n_{2}} - \sum_{p \in \mathcal{P}_{n_{2}}} (n_{p}^{v} - n_{p}^{s})\right\}, \ l \in \mathcal{L}, \ n_{1}, n_{2} \in \underline{N}$$
(5.13)

where

$$\sum_{p\in\mathcal{P}_n}n_p^v\leq T_n\quad n\in\mathcal{N}.$$

It is assumed that the data are queued at buffers (of infinite capacity) available at the node transceivers. The above equation for the data link capacity expresses the dependence of the M/D/c data model on the current voice traffic at nodes n and m. However, it does not address the way in which a node's unused transceivers are allocated to support data links with each of its neighbors, as was discussed in Section 2.4.2.

To simplify the notation we consider the link $l = (n_1, n_2) = (1, 2)$ connecting the network nodes $1, 2 \in \mathcal{N}$. The above data capacity can be written as

$$c'_{l} = \min \{T_{1} - k_{1} + m_{1}, \ T_{2} - k_{2} + m_{2}\}$$
(5.14)

where

$$0 \le k_i = \sum_{p \in \mathcal{P}_i} n_p^v \le T_i, \qquad i = 1, 2$$
 (5.15a)

$$0 \le m_i = \sum_{p \in \mathcal{P}_i} n_p^s \le k_i, \qquad i = 1, 2$$
 (5.15b)

and \mathcal{P}_i is the set of paths containing node *i*, for i = 1, 2.

To employ the knapsack approximation in this case for a single-link network involves defining

$$P(k_1, m_1, k_2, m_2) = \sum_{(\underline{N}^{\nu}, \underline{N}^{\bullet}) \in \Omega(k_1, m_1, k_2, m_2)} P(\underline{N}^{\nu}, \underline{N}^{\bullet})$$
(5.16)

where

$$\Omega(k_1, m_1, k_2, m_2) = \left\{ (\underline{N}^v, \underline{N}^s) \mid 0 \le n_p^s \le n_p^v, \ p \in \mathcal{P}; \sum_{p \in \mathcal{P}_i} n_p^v = k_i, \sum_{p \in \mathcal{P}_i} n_p^s = m_i, \ i = 1, 2 \right\}$$

$$\mathcal{P} = \mathcal{P}_1 \cup \mathcal{P}_2$$
(5.17)

$$\sum_{k_1=0}^{T_1} \sum_{m_1=0}^{k_1} \sum_{k_2=0}^{T_2} \sum_{m_2=0}^{k_2} P(k_1, m_1, k_2, m_2) = 1$$

and writing

$$Q_{l} = 1 - \sum_{k_{1}=0}^{T_{1}} \sum_{m_{1}=0}^{k_{1}} \sum_{k_{2}=0}^{T_{2}} \sum_{m_{2}=0}^{k_{2}} I(c_{l}' \neq 0) \cdot \bar{P}(c_{l}', \rho_{l}^{d}) \cdot P(k_{1}, m_{1}, k_{2}, m_{2})$$
(5.18)

where

$$\bar{P}(c_l', \rho_l^d) = \sum_{n_l^d=0}^{c_l'-1} P(n_l^d | c_l')$$

is provided by the M/D/c data model (Appendix B).

After the appropriate normalization

$$P'(k_1, m_1, k_2, m_2) = P(k_1, m_1, k_2, m_2) / P(0, 0, 0, 0)$$
(5.19)

and using the approach of Section 5.1, the above single-link approximation is put in the form

$$Q_{l}(T_{1}, T_{2}; \rho_{p}^{v} p \in \mathcal{P}_{1} \cup \mathcal{P}_{2}) = 1 - \frac{\sum_{k_{1}=0}^{T_{1}} \sum_{m_{1}=0}^{k_{1}} \sum_{k_{2}=0}^{T_{2}} \sum_{m_{2}=0}^{k_{2}} I(c_{l}' \neq 0) \bar{P}(c_{l}', \rho_{l}^{d}) P'(k_{1}, m_{1}, k_{2}, m_{2})}{\sum_{k_{1}=0}^{T_{1}} \sum_{m_{1}=0}^{k_{1}} \sum_{m_{2}=0}^{T_{2}} \sum_{m_{2}=0}^{k_{2}} P'(k_{1}, m_{1}, k_{2}, m_{2})}$$

$$(5.20)$$

with

$$c'_{l} = \min \{T_{1} - k_{1} + m_{1}, T_{2} - k_{2} + m_{2}\}.$$
(5.14)

First we consider the case $\mathcal{P}_1 = \mathcal{P}_2$, which means that the same set of paths passes through node 1 and node 2. In this case, $k_1 = k_2$, $m_1 = m_2$, and Q_i simplifies to

$$Q_{l} = 1 - \frac{\sum_{k=0}^{T} \sum_{m=0}^{k} I(c_{l}' \neq 0) \cdot \bar{P}(c_{l}', \rho_{l}^{d}) \cdot P'(k, m)}{\sum_{k=0}^{T} \sum_{m=0}^{k} P'(k, m)}$$
(5.21*a*)

where

$$T = \min\{T_1, T_2\}$$
(5.21b)

$$c'_{l} = \min \{T_{1}, T_{2}\} - k + m = T - k + m$$
(5.21c)

The recursion for P'(k,m) is given by (5.10).

Next we consider the case $\mathcal{P}_1 \neq \mathcal{P}_2$, in which case calls of one or more types are supported by only one of the nodes of interest. The following Proposition holds **Proposition 5.2**

 $P'(k_1, m_1, k_2, m_2)$ satisfies the recursion

$$\begin{split} P'(k_1, m_1, k_2, m_2) &= \\ & \left\{ \begin{array}{ll} \frac{1}{m_1} \cdot \frac{\alpha}{\alpha + \beta} \left[\sum_{p \in \mathcal{P}_1 \cap \mathcal{P}_2^c} \rho_p^{v} P'(k_1 - 1, m_1 - 1; k_2, m_2) + \sum_{p \in \mathcal{P}_1 \cap \mathcal{P}_2^c} \rho_p^{v} P'(k_1 - 1, m_1 - 1; k_2 - 1, m_2 - 1) \right]; \\ & \text{if } 1 \leq m_1 \leq k_1 \leq T_1, 1 \leq m_2 \leq k_2 \leq T_2; \\ \frac{1}{m_1} \cdot \frac{\alpha}{\alpha + \beta} \left[\sum_{p \in \mathcal{P}_1 \cap \mathcal{P}_2^c} r_p^{(1)} \rho_p^{v} P'(k_1 - 1, m_1 - 1; k_2, m_2) \right]; \\ & \text{if } 1 \leq m_1 \leq k_1 \leq T_1, m_2 = 0 \leq k_2 \leq T_2; \\ \frac{1}{m_2} \cdot \frac{\alpha}{\alpha + \beta} \left[\sum_{p \in \mathcal{P}_1 \cap \mathcal{P}_2^c} \rho_p^{v} P'(k_1, m_1; k_2 - 1, m_2 - 1) \right]; \\ & \text{if } m_1 = 0 \leq k_1 \leq T_1, 1 \leq m_2 \leq k_2 \leq T_2 \\ \frac{1}{k_1 - m_1} \cdot \frac{\beta}{\alpha + \beta} \left[\sum_{p \in \mathcal{P}_2 \cap \mathcal{P}_1^c} \rho_p^{v} P'(k_1 - 1, m_1; k_2, m_2) + \sum_{p \in \mathcal{P}_1 \cap \mathcal{P}_2} \rho_p^{v} P'(k_1 - 1, m_1; k_2 - 1, m_2) \right]; \\ & \text{if } m_1 = 0 < k_1 \leq T_1, m_2 = 0 < k_2 \leq T_2 \\ 1; \\ 1; \\ 0; \end{array} \right] \end{split}$$

(5.22)

where \mathcal{P}_2^c denotes the complement of the set \mathcal{P}_2 .

Proof: Refer to Appendix D where a more general proof involving constraints of the form

$$0 \le k_i = \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^v \le Z_i, \qquad i = 1, 2$$
$$0 \le m_i = \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^s \le k_i, \qquad i = 1, 2$$

is provided. For the approximation described in this section we considered only the special case

$$r_p^{(1)} = r_p^{(2)} = 1 \quad p \in \mathcal{P}_1 \cup \mathcal{P}_2$$

and

$$Z_i = T_i, \quad i = 1, 2$$

For the multi-link network, the knapsack approximation for node $l = (n_1, n_2)$ is obtained with the help of the thinned load method, thus we use the function of (5.20) (where $1 = n_1$ and $2 = n_2$) with voice loads $\rho_p^v \prod_{n \in p, n \neq n_1, n_2} (1 - L_{np})$ instead of ρ_p^v to get

$$Q_l\left(T_{n_1}, T_{n_2} ; \rho_p^v \prod_{n \in p, n \neq n_1, n_2} (1 - L_{np}), p \in \mathcal{P}_1 \cup \mathcal{P}_2\right), \quad l = \{n_1, n_2\} \in \mathcal{L}$$
(5.23)

for a fixed value of ρ_l^d and

$$c'_{l} = \min \left\{ T_{n_{1}} - k_{1} + m_{1}, T_{n_{2}} - k_{2} + m_{2} \right\}.$$

6. PASCAL APPROXIMATION TO THE PROBABILITY OF DATA QUEUEING

In this section we describe in detail the application of the Pascal approximation to the evaluation of the probability of data queueing, a development that appears in this report for the first time. This approximation is first developed for general multi-rate networks in Section 6.1; in Section 6.2, we describe how it can be modified to suit the multi-hop radio network of Sections 2.1.2 and 2.2.2.

6.1 General Multi-Rate Networks

In Section 4.3 the Pascal approximation used a one dimensional birth-death process whose equilibrium probability mass function is the Pascal distribution to approximate the number of busy circuits (paths with ongoing voice calls). In order to approximate the probability of data queueing given by (3.13), which involves conditioning on the total number of voice calls and the on the number of silent (inactive) calls, we need to extend the Pascal approximation to two dimensions.

In this context, we consider the two-dimensional birth-death process (k-m,m) (where k-m roughly represents the total rate used by all active calls and m the rate of all inactive calls) on the state space $\{(k,m) \mid 0 \le k \le c, 0 \le m \le k\}$ with the transition diagram of Figure 1, where

$$\lambda_{j,i} = \left[\frac{\epsilon_j^2}{\sigma_j^2} + i\left(1 - \frac{\epsilon_j}{\sigma_j^2}\right)\right], \quad j = 1, 2, \ i = 0, 1, 2, \dots, c \tag{6.1}$$

and ϵ_j^2, σ_j^2 , for j = 1, 2, are given positive numbers. Let q(k-m, m) denote the equilibrium probability of being in state (k-m, m). Furthermore, consider the associated process (k, m) and denote by $\hat{P}(k, m)$ the equilibrium probability of being at state (k, m).

From the local balance equations for the birth-death process (k-m,m) we know that q(k-m,m) must satisfy the conditions

$$(k-m) \cdot q(k-m,m) = \lambda_{1,k-1-m} \cdot q(k-1-m,m)$$
(6.2)

and

$$m \cdot q(k-m,m) = \lambda_{2,m-1} \cdot q(k-m,m-1).$$
 (6.3)

The resulting conditions for $\hat{P}(k,m)$ are

$$(k-m) \cdot \hat{P}(k,m) = \lambda_{1,k-1-m} \cdot \hat{P}(k-1,m)$$
 (6.4)

and

$$m \cdot \hat{P}(k,m) = \lambda_{2,m-1} \cdot \hat{P}(k-1,m-1).$$
 (6.5)

Using the above equations we can evaluate $\hat{P}(k,m)$ from the following recursion

$$\hat{P}(k,m) = \begin{cases} \frac{\lambda_{1,k-1}}{k} \cdot \hat{P}(k-1,0), & m=0; \ 1 \le k \le c\\ \frac{\lambda_{2,m-1}}{m} \cdot \hat{P}(k-1,m-1), & 1 \le m \le k; \ 1 \le k \le c \end{cases}$$
(6.7*a*)

where

$$\sum_{k=0}^{c} \sum_{m=0}^{k} \hat{P}(k,m) = 1.$$
 (6.8*a*)

If instead of $\hat{P}(k,m)$ we use the normalized version

$$\hat{P}'(k,m) = \hat{P}(k,m)/\hat{P}(0,0)$$

then the recursion (6.7a)-(6.8a) becomes

$$\hat{P}'(k,m) = \begin{cases} \frac{\lambda_{1,k-1}}{k} \cdot \hat{P}'(k-1,0), & m=0; \ 1 \le k \le c\\ \frac{\lambda_{2,m-1}}{m} \cdot \hat{P}'(k-1,m-1), & 1 \le m \le k; \ 1 \le k \le c \end{cases}$$
(6.7b)

with initial condition

$$\dot{P}'(0,0) = 1 \tag{6.8b}$$

which is easier to evaluate [it does not require knowledge of P(0,0)] and results in

$$\left[\sum_{k=0}^{c}\sum_{m=0}^{k}\hat{P}'(k,m)\right]^{-1}=\hat{P}(0,0).$$

Proposition 6.1

When $c = \infty$, $\hat{P}(k,m)$ is two-dimensional Pascal-distributed with the following parameters:

$$E[k-m] = \epsilon_1 \tag{6.9}$$

$$E[m] = \epsilon_2 \tag{6.10}$$

$$var(k-m) = \sigma_1^2 \tag{6.11}$$

$$var(m) = \sigma_2^2 \tag{6.12}$$

$$cov(k-m,m) = 0.$$
 (6.13)

Proof: It is provided in Appendix D.

The application of the Pascal distribution to the probability of data queueing involves (as in the case of the knapsack approximation) considering first a single-link multi-rate network, applying the Pascal approximation, and then obtaining the result for the multilink case using thinned voice loads in place of the initial voice loads. We start with the following proposition

Proposition 6.2

For the single-link multi-rate network, when the link capacity $c = \infty$, denote by n_v the number of channels occupied by voice calls in the link and denote by n_s the number of channels occupied by silence voice calls in the link, then

$$E[n_v] = \sum_{p \in \mathcal{P}} r_p \rho_p^v \tag{6.14}$$

$$E[n_{s}] = \frac{\alpha}{\alpha + \beta} \cdot \sum_{p \in \mathcal{P}} r_{p} \rho_{p}^{v}$$
(6.15)

$$E[n_v - n_s] = \frac{\beta}{\alpha + \beta} \cdot \sum_{p \in \mathcal{P}} r_p \rho_p^v$$
(6.16)

$$var(n_v) = \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v \tag{6.17}$$

$$var(n_s) = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v$$
(6.18)

$$var(n_v - n_s) = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v$$
(6.19)

$$cov(n_v - n_s, n_s) = 0. \tag{6.20}$$

Proof: It is provided in Appendix E.

From the above two propositions we observe that the infinite capacity single-link multi-rate model employs the same means and variances for $(n_v - n_s)$ and n_s as the two-dimensional birth-death process with parameters $\epsilon_1 = E[n_v - n_s]$, $\epsilon_2 = E[n_s]$, $\sigma_1^2 = E[n_s]$

 $var(n_v - n_s)$, and $\sigma_2^2 = var(n_s)$. Note that the covariances $cov\{n_v - n_s, n_s\}$ and $cov\{k - m, m\}$ are both zero. Therefore, as in the voice case, the probability of data queueing in (3.13) for the finite-capacity single-link multi-rate model can approximated by

$$\hat{Q}_{l}(c_{l};\rho_{p}^{v},p\in\mathcal{P}_{l}) = 1 - \sum_{k=0}^{c_{l}}\sum_{m=0}^{k}I[(k,m)\neq(c_{l},0)]\cdot\bar{P}(c_{l}',\rho_{l}^{d})\cdot\hat{P}(k,m)$$
$$= 1 - \frac{\sum_{k=0}^{c_{l}}\sum_{m=0}^{k}I[(k,m)\neq(c_{l},0)]\cdot\bar{P}(c_{l}',\rho_{l}^{d})\cdot\hat{P}'(k,m)}{\sum_{k=0}^{c_{l}}\sum_{m=0}^{k}\hat{P}'(k,m)}$$
(6.21)

where $\bar{P}(c'_l, \rho^d_l)$ is given by (5.7a)-(5.7b) for an M/M/ c'_l data model and can be evaluated from Appendix B for an M/D/ c'_l data model, and $\hat{P}'(k,m)$ is the equilibrium probability of the above birth-death process with parameters

$$\epsilon_1 = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} r_p \rho_p^v \tag{6.22}$$

$$\epsilon_2 = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} r_p \rho_p^v \tag{6.23}$$

$$\sigma_1^2 = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}_i} r_p^2 \rho_p^v$$
(6.24)

$$\sigma_2^2 = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} r_p^2 \rho_p^v \tag{6.25}$$

which can be evaluated iteratively via (6.7b)-(6.8b). We used the notation $\hat{Q}_l(c_l; \rho_p^v, p \in \mathcal{P}_l)$ in order to emphasize the functional dependence of the probability of data queueing on the voice loads $(\rho_p^v, p \in \mathcal{P}_l)$. This is necessary for the extension to the multi-link case that follows.

For the multi-link multi-rate network we use as the approximation to the probability of data queueing at link l the expression

$$\hat{Q}_{l}\left(c_{l};\rho_{p}^{v}\prod_{\ell\in p,\ell\neq l}(1-\hat{L}_{\ell p}),p\in\mathcal{P}_{l}\right)$$
(6.26)

where $\hat{L}_{\ell p}$ is the (approximate) probability of voice call taking path p being blocked at link ℓ , which is obtained via the Pascal approximation as in Section 4.3. As it was done in

Section 5, we again used the interlink independence approximation for voice and thinned the voice loads ρ_p^v to $\rho_p^v \prod_{\ell \in p, \ell \neq l} (1 - \hat{L}_{\ell p})$ for $p \in \mathcal{P}_l$ in (6.26).

The same comments about the practical accuracy and the asymptotic accuracy of this approximation with those made at the end of Section 5 are valid here.

6.2 Radio Network Model of Sections 2.1.2, 2.2.2, and 2.4.2

The general approach described in Section 6.1 above can be extended to derive a four-dimensional Pascal approximation for the approximation of probability of queueing data of the radio network model of Sections 2.1.2 and 2.2.2. The basic steps are similar to those detailed in Section 5.2 for the knapsack approximation. The key step is again the derivation of recursive expressions for the quantity $\hat{P}(k_1, m_1, k_2, m_2)$ defined as in (5.15) but evaluated via the Pascal method for the multi-rate case. We omit the details here which are similar to those provided in Appendix D. For the single-rate case characterizing the radio network model of this report the knapsack and Pascal approximations yield identical results (see Appendix F) and thus there is no need for additional work.

7. EXTENSION OF THE APPROXIMATIONS TO M/D/c DATA MOD-ELS AND THE AVERAGE QUEUEING DELAY AS THE PERFORMANCE MEASURE

As discussed in Section 2 the performance measures of interest for the data traffic are

 $Q_l \triangleq$ probability of data queued at link $l \ (l \in \mathcal{L})$,

 $W_l \stackrel{\Delta}{=}$ average waiting time (not including service time) of data in queue at link l $(l \in \mathcal{L})$,

and their averages \bar{Q} and \bar{W} [refer to (2.1b)-(2.1c)] with respect to the data loads of the links. In this section we first show (Section 7.1) that the approximations to the probability of data queueing of Sections 5 and 6 are applicable not only to the M/M/c data model but also to the M/D/c data model. By contrast, for the average data queueing delay these approximations are shown (Section 7.2) to apply only to the M/D/c data model. Finally, in Section 7.3 a modified network model that guarantees finite data delays by dedicating a portion of the link capacity (or the number of node transceivers) to exclusive data use is provided.

7.1 Approximations to the Probability of Data Queueing for M/D/c Data Models

Recall that for the M/M/c data model and a multi-rate system, the probability of data queueing Q_i takes the form

$$Q_{l} = Pr(n_{l}^{d} \ge c_{l} - k_{l}^{v} + k_{l}^{s}) = 1 - Pr\left[c_{l} - k_{l}^{v} + k_{l}^{s} > n_{l}^{d}, \text{ where } (k_{l}^{v}, k_{l}^{s}) \neq (c_{l}, 0)\right]$$
$$= 1 - \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega_{s}} \left[\sum_{n_{l}^{d} = 0}^{c_{l}^{\prime} - 1} P(n_{l}^{d} | c_{l}^{\prime})\right] P(\underline{N}^{v}, \underline{N}^{s})$$
(7.1)

where

$$k_l^v = \sum_{p \in \mathcal{P}_l} r_p n_p^v \tag{7.2}$$

$$k_l^s = \sum_{p \in \mathcal{P}_l} r_p n_p^s \tag{7.3}$$

and

$$c'_{l} = c_{l} - k^{v}_{l} + k^{s}_{l}. \tag{7.4}$$

 $P(\underline{N}^{v}, \underline{N}^{s})$ is the steady state probability of the voice calls evaluated in Section 3, and $P(n_{l}^{d}|c_{l}')$ is the steady state probability of the number of data packets in an $M/M/c_{l}'$ system. The application of the knapsack and Pascal methods for obtaining approximations to the probability of data queueing Q_{l} was described in Sections 5 and 6, respectively, for an M/M/c data model.

The corresponding formula for the M/D/c data model is obtained by simply replacing the $P(n_l^d|c_l)$ of an $M/M/c_l'$ system by that of an $M/D/c_l'$ system (see Appendix B). Consequently, the application of the knapsack and Pascal approximation methods to Q_l and an M/D/c data model is a straightforward extension of the results of Sections 5 and 6.

7.2 Approximations to the Average Queueing Delay of Data

The average data queueing delay W_l for the M/M/c data model and the M/D/c data model takes (upon application of Little's formula) the form

$$W_{l} = \frac{\text{average number of data packets in the queue at link }l}{\text{average data rate to link }l} = \frac{N_{l}^{Q}}{F_{l}^{d}}$$
(7.5)

where

$$N_l^Q = \sum_{(\underline{N}^v, \underline{N}^s) \in \Omega_s} \left[\sum_{n=0}^{\infty} n P(n_l^d = c_l' + n | c_l') \right] P(\underline{N}^v, \underline{N}^s)$$
(7.6)

$$F_l^d = \rho_l^d \cdot \mu_l^d \tag{7.7}$$

and $P(n_l^d | c_l')$ and $P(\underline{N^v}, \underline{N^s})$ are as described above for the two models. Therefore, W_l can be evaluated by applying the various approximations on N_l^Q and treating $\sum_{n=0}^{\infty} nP(n_l^d = c_l' + n \mid c_l')$ in the same manner as $\sum_{n_l^d=0}^{c_l'-1} P(n_l^d \mid c_l')$ was treated in the original performance measure, i.e., in the probability of data queueing. Finally, $\sum_{n=0}^{\infty} nP(n_l^d = c_l' + n \mid c_l')$ can be evaluated in closed form (omitted here) for an $M/M/c_l'$ system and via Tijms' algorithm (see [13] and Appendix B) for an $M/D/c_l'$ system.

The application of the approximation methods of Sections 5 and 6 to the queueing delay for data W_l and a M/D/c model is described in more detail next.

7.2.1 Knapsack Approximation

As discussed above we apply the approximation methods directly to the evaluation of the average number of data packets in the queue (N_l^Q) required for the evaluation of the average queueing delay $W_l = N_l^Q/F_l^d$. As in Section 5 we start with a single-link multiple-rate scenario and write for link l

$$N_{l}^{Q} = \sum_{k=0}^{c_{l}} \sum_{m=0}^{k} I[(k,m) \neq (c_{l},0)] \cdot \left[\sum_{n=0}^{\infty} nP(n_{l}^{d} = c_{l}' + n \mid c_{l}') \right] \cdot P(k,m)$$
(7.8)

where

$$c_l' = c_l - k + m \tag{7.9}$$

$$P(k,m) = \sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(k,m)} P(\underline{N}^{v},\underline{N}^{s})$$
(7.10)

and

$$\Omega(k,m) = \left\{ (\underline{N}^{v}, \underline{N}^{s}) \mid 0 \le n_{p}^{s} \le n_{p}^{v}; \ 0 \le r_{p}n_{p}^{v} \le c_{p}; \ p \in \mathcal{P}_{l}, \sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{s} = m; \sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{v} = k \right\}.$$

$$(7.11)$$

Let

$$\bar{N}(c'_{l}, \rho^{d}_{l}) = \sum_{n=0}^{\infty} n P(n^{d}_{l} = c'_{l} + n \mid c'_{l})$$
(7.12)

which is the average number of data packets in the queue for an $M/D/c'_l$ system, and is isolated from other terms in N_l^Q and can thus be evaluated separately. The evaluation of $\bar{N}(c'_l, \rho^d_l)$ is straightforward from Tijms' algorithm (see [13]) after the steady-state probability for $M/D/c'_l$ data has been obtained.

Then, following Section 5 we obtain an approximation to N_l^Q as

$$N_l^Q\left(c_l; \ \rho_p^v \prod_{\ell \in \mathcal{P}_l, \ell \neq l} (1 - L_{\ell p}), p \in \mathcal{P}_l\right)$$
(7.13)

where $L_{\ell p}$ is the approximate voice blocking probability for link ℓ across voice path p evaluated in Section 4.2 (via the knapsack approximation) and $N_l^Q(c_l; \rho_p^v, p \in \mathcal{P}_l)$ is a function of c_l and $(\rho_p^v, p \in \mathcal{P}_l)$ given by

$$N_{l}^{Q}[c_{l}; \ \rho_{p}^{v}, \ p \in \mathcal{P}_{l}] = \frac{\sum_{k=0}^{c_{l}} \sum_{m=0}^{k} I[(k,m) \neq (c_{l},0)] \cdot \bar{N}(c_{l}',\rho_{l}^{d}) \cdot P'(k,m)}{\sum_{k=0}^{c_{l}} \sum_{m=0}^{k} P'(k,m)}$$
(7.14)

and P'(k,m) is given by the same equations (5.10a)-(5.10c) of Section 5.

7.2.2 Pascal Approximation

For the Pascal approximation, we define $\bar{N}(c_l, \rho_l^d)$ as in (7.12) and still use the expressions (7.8)-(7.11) as in the knapsack method above. However, for the average length of data queue of link l (and a single-link network) we now use

$$\hat{N}_{l}^{Q}[c_{l}; \ \rho_{p}^{v}, \ p \in \mathcal{P}_{l}] = \frac{\sum_{k=0}^{c_{l}} \sum_{m=0}^{k} I[(k,m) \neq (c_{l},0)] \cdot \bar{N}(c_{l}',\rho_{l}^{d}) \cdot \hat{P}'(k,m)}{\sum_{k=0}^{c_{l}} \sum_{m=0}^{k} \hat{P}'(k,m)}$$
(7.15)

instead of (7.14), where $\hat{P}'(k,m)$ is now obtained from (6.7b) as described in Section 6. Subsequently, the approximation to N_l^Q (for a multi-link network) is obtained again as

$$\hat{N}_{l}^{Q}\left(c_{l}; \ \rho_{p}^{v}\prod_{\ell\in\mathcal{P}_{l}, \ell\neq l} (1-\hat{L}_{\ell p}), p\in\mathcal{P}_{l}\right)$$
(7.16)

where $\hat{L}_{\ell p}$ is the approximate voice blocking probability for link ℓ across voice path p evaluated in Section 4.3 (via the Pascal approximation). Finally, the desired performance measure W_l is obtained as $W_l = \frac{\dot{N}_l^Q}{F_l^Q}$.

7.3 Network Model Modification for Guaranteeing Finite Data Delays

Since the link capacity available for data given by (7.4) can take the value 0 we can not guarantee finite data delays unless we assume that a (small) fraction of the link capacity is always reserved for data use.

For the multi-rate network model of Sections 2.1.1 and 2.2.1 the above requirement implies that

$$c'_{l} = c^{d}_{l} + c^{v}_{l} - \sum_{p \in \mathcal{P}_{l}} r_{p} n^{v}_{p} + \sum_{p \in \mathcal{P}_{l}} r_{p} n^{s}_{p}$$
(7.17)

where c_l^d and c_l^v are the portions of the link capacity reserved for data and voice traffic respectively; if c_l^v is not used by the current voice traffic it can be used by the data traffic as (7.17) indicates.

For the single-rate radio network model of Sections 2.1.2, 2.2.2, and 2.4.2 the above requirement implies that the capacity available for data at link $l = (n_1, n_2)$ is

$$c_{l}' = \min\left\{T_{n_{1}}^{d}, T_{n_{2}}^{d}\right\} + \min\left\{T_{n_{1}}^{v} - \sum_{p \in \mathcal{P}_{n_{1}}} (n_{p}^{v} - n_{p}^{s}), T_{n_{2}}^{v} - \sum_{p \in \mathcal{P}_{n_{2}}} (n_{p}^{v} - n_{p}^{s})\right\}, l \in \mathcal{L}, \ n_{1}, n_{2} \in \mathcal{N}$$
(7.18)

where T_n^d represents the number of transceivers at node *n* that are dedicated to data and T_n^v represents the number of transceivers which are primarily used by voice; the portion of T_n^v not occupied by voice calls can by used by data but voice maintains preemptive priority over data over this portion. The total number of transceivers at node *n* is

$$T_n = T_n^d + T_n^v, \quad n \in \mathcal{N}.$$
(7.19)

Similarly to the comment following eq. (5.13) and the discussion in Section 2.4.2, the above equation (7.18) does not address the way in which a node's unused transceivers are allocated to support data links with each of its neighbors.

Finally, note that, even if a portion of the link capacity $(c_l^d \text{ or } T_n^d)$ is set aside for exclusive data use, the data delay remains finite only for average data loads smaller than c_l^d (i.e., $\rho_l^d < c_l^d$); refer to the relevant comment (Comment 2) following Proposition 3.1 in Section 3.

8. EVALUATION OF PERFORMANCE MEASURES VIA THE MONTE-CARLO SUMMATION METHOD

For the purpose of comparison and testing the accuracy of the the knapsack and Pascal approximations to the performance measures of interest we also evaluate the probability of voice blocking, the probability of data queueing, and the queuing data delay via the Monte-Carlo summation method. In the next subsection this method is described in some detail and then in the following subsections it is explained how it is applied to the various performance measures.

The key difference between the approximation method described in this section and the approximation methods of the previous sections (Sections 4, 5, 6, and 7) is that the Monte-Carlo summation method can provide an estimate (approximation) to the performance measure of interest within any desired confidence interval at the cost of an increase in the number of calls to random number generators performed. In this sense the Monte-Carlo summation method constitutes our baseline for the true values of the performance measures of interest in this report, and, where exact values are not available, the accuracy of all other approximations is compared to the values generated by the Monte-Carlo simulation of the appropriate sums.

It should be emphasized that in contrast to the familiar Monte-Carlo simulation method the Monte-Carlo summation approach does not involve a simulation of the dynamic behavior of the system under study. As is explained in the following section, quantities related to system performance are evaluated by using known properties of their distributions in conjunction with a random generation process.

8.1 The Monte-Carlo Summation Method

It is now generally accepted that, in the absence of a special structure, multidimensional integration (or summation) is best performed by Monte Carlo methods [7], [15]. This method has been shown to be efficient and accurate in several applications involving expressions of the form

$$G = \sum_{\underline{n}\in\Omega} \prod_{k=1}^{K} \hat{q}_k(n_k)$$
(8.1)

where Ω denotes the state space of an underlying K-dimensional vector stochastic process $\underline{n} = (n_1, \ldots, n_K)$ and $q_k(\cdot), k = 1, \ldots, K$, are known functions. For the cases of interest to

our study, G represents the normalization constant of (3.10) for product-form stochastic networks with voice traffic. However, as we have seen in Section 3 [eqs. (3.12a)-(3.12b) and (3.13)] and Section 7 [eqs. (7.5)-(7.7)], the performance measures of interest can actually be expressed as nonlinear functions of normalization constants. Therefore, expressions of the form

$$\Phi = \frac{\sum_{\underline{n} \in \Lambda} f(\underline{n})q(\underline{n})}{\sum_{\underline{n} \in \Lambda} q(\underline{n})}$$
(8.2)

where $f(\cdot)$ is a known function are even more necessary than expressions of the form (8.1) in the evaluation procedure. The method for the evaluation of (8.1) is also applicable to (8.2). Thus, here, we will first describe briefly the method for evaluating (8.1) and then, at the end of this subsection we will describe the evaluation of (8.2).

The starting point is to let

$$q(\underline{n}) = I(\underline{n} \in \Omega) \prod_{k=1}^{K} q_k(n_k)$$
(8.3)

where $I(\cdot)$ is the indicator function and rewrite (8.1) as follows:

$$G = \sum_{n_1=0}^{N_1} \sum_{n_2=0}^{N_2} \dots \sum_{n_K=0}^{N_K} q(\underline{n})$$
(8.4)

where $N_k = \max\{n_k : \underline{n} \in \Omega\}$. Thus calculating G involves a multi-dimensional summation.

First, we let $\underline{V}^i = (V_1^i, V_2^i, \dots, V_K^i)$, $i = 1, 2, \dots, n$, be a sequence of n i.i.d. random vectors, where each V^i takes values in $\Lambda = \{0, \dots, N_1\} \times \{0, \dots, N_2\} \times \dots \times \{0, \dots, N_K\}$. Next, we define $P_s(\underline{n}) = P(\underline{V}^i = \underline{n})$, for $\underline{n} \in \Lambda$, which is a sampling distribution that can be specified in a way that optimizes the efficiency of the Monte Carlo method, and set

$$Z^{i} = \frac{q(\underline{V}^{i})}{P_{s}(\underline{V}^{i})}.$$
(8.5)

Then the quantity

$$\overline{Z}_n = \frac{1}{n} \sum_{i=1}^n Z^i \tag{8.6}$$

provides an unbiased estimator for G (i.e., $E[\overline{Z}_n] = G$). Moreover, the Central Limit Theorem implies that, for large n,

$$P\left(|\overline{Z}_n - G| \le \frac{c(\eta)\sigma_n(Z)}{\sqrt{n}}\right) = 1 - \frac{\eta}{2}$$
(8.7)

where $c(\eta)$ is the critical value of the standard normal distribution N(x), that is, it satisfies

$$1 - \frac{\eta}{2} = \int_{-\infty}^{c(\eta)} N(x) dx \tag{8.8}$$

and $\sigma_n^2(Z)$ is the sample variance of Z^i , for i = 1, ..., n, i.e.,

$$\sigma_n^2(Z) = \frac{1}{n-1} \sum_{i=1}^n (Z^i - \overline{Z}_n)^2.$$
(8.9)

Notice that, for any fixed n (simulation size), \overline{Z}_n is an estimate for G, whose accuracy can be assessed for the confidence interval $100(1 - \eta)\%$ by

$$\left[\overline{Z}_n - \frac{c(\eta)\sigma_n(Z)}{\sqrt{n}}, \overline{Z}_n + \frac{c(\eta)\sigma_n(Z)}{\sqrt{n}}\right]$$
(8.10)

induced by (8.7). As the samples are being drawn, the sample variance can be calculated and the confidence intervals can be given explicitly. Furthermore, if greater accuracy is desired, more samples can be drawn, thereby decreasing the width of the confidence interval. This method is particularly well suited for optimization, as only rough estimates are needed for performance metaures and gradients when the current solution is not close to optimal. Ross [7] actually shows that the gradients of the performance measures pertaining to voice can be obtained with little additional effort.

From (8.7) it is clear that the effectiveness of the Monte Carlo summation method depends on

- 1. the effort required to generate \underline{V}^i from the distribution $P_{\bullet}(\underline{n}), \underline{n} \in \Lambda$;
- 2. the effort required to evaluate the ratio $q(\cdot)/P_s(\cdot)$ during the sampling procedure;
- 3. the size of σ^2 , the variance of Z^i .

To improve the efficiency of the method the following steps are usually taken. First, if the random variables $V_1^i, V_1^i, \ldots, V_K^i$ are independent (i.e., $F_{\bullet}(\underline{n}) =$

 $P_1(n_1)P_2(n_2)\ldots P_K(n_K)$, the \underline{V}_i can be generated in a total of O(K) time with the alias algorithm (e.g., see [16]); this computational effort is independent of the number of values the stochastic process (which in our example is link occupancy) can take on. This means that the method can handle networks with large link capacities. Second, selecting the appropriate sampling distribution $P_s(\underline{n})$, for $\underline{n} \in \Lambda$, can significantly reduce the variance σ^2 . In particular, it is desirable to sample more frequently the points \underline{n} , at which $q(\underline{n})$ is important, which is typically done by considering functions $P_s(\cdot)$ that are similar to $q(\cdot)$. Ideally, one would like $q(\cdot)/P_s(\cdot)$ to be nearly constant; however, there exists a tradeoff between this similarity and the effort required to sample from $P_s(\cdot)$.

As we already mentioned, many performance measures of interest are given by nonlinear functions of normalization constants, and thus quantities of the form (8.2) must be evaluated. A natural estimate for Φ based on an *n* sample simulation is

$$\Phi_n = \frac{\sum_{i=1}^n Y^i}{\sum_{i=1}^n Z^i}$$
(8.1)

where

$$Y^{i} = f(\underline{V}^{i})q(\underline{V}^{i})/P_{s}(\underline{V}^{i})$$
(8.12)

and

$$Z^{i} = q(\underline{V}^{i})/P_{\bullet}(\underline{V}^{i}).$$

$$(8.13)$$

Although Φ_n converges (almost surely) to Φ of (8.3), Φ_n has the undesirable property of being biased. Fortunately, this bias diminishes as n becomes large. It is also known [7] that the ratio estimator Φ_n can be made free of bias to order 1/n with a modification that requires an insignificant amount of additional CPU time. Moreover, the confidence interval for Φ can again be constructed (see [7]) as the sampling proceeds (i.e., on line) as follows. Let \overline{Y}_n and $\sigma_n^2(Y)$ be the sample mean and variance associated with Y_i , $i = 1, \ldots, n$ [defined analogously to \overline{Z}_n of (8.6) and $\sigma_n^2(Z)$ of (8.9)]. Furthermore let

$$\sigma_n^2(Y,Z) = \frac{1}{n-1} \sum_{i=1}^n (Y^i - \overline{Y}_n)(Z^i - \overline{Z}_n)$$
(8.14)

be the sample covariance associated with the two sets of random variables. Then the $(1 - \eta)100\%$ confidence interval for Φ_n is

$$\left(\frac{\overline{Y}_{n}\overline{Z}_{n}-\frac{c^{2}(\eta)}{n}\sigma_{n}^{2}(Y,Z)-r_{n}}{\overline{Z}_{n}^{2}-\frac{c^{2}(\eta)}{n}\sigma_{n}^{2}(Z)},\frac{\overline{Y}_{n}\overline{Z}_{n}-\frac{c^{2}(\eta)}{n}\sigma_{n}^{2}(Y,Z)+r_{n}}{\overline{Z}_{n}^{2}-\frac{c^{2}(\alpha)}{n}\sigma_{n}^{2}(Z)}\right)$$
(8.15)

where $c(\eta)$ is as defined above and r_n is given by

$$r_n = \sqrt{\left[\overline{Y}_n \overline{Z}_n - \frac{c^2(\eta)}{n} \sigma_n^2(Y, Z)\right]^2} - \left[\overline{Z}_n^2 - \frac{c^2(\eta)}{n} \sigma_n^2(Z)\right] \left[\overline{Y}_n^2 - \frac{c^2(\eta)}{n} \sigma_n^2(Y)\right]$$
(8.16)

Note that the width of the confidence interval is $O(1/\sqrt{n})$.

8.2 Monte-Carlo Summation for the Probability of Voice Blocking

The method described in Section 8.1 for the estimator (8.5)-(8.6) of (8.1) and the ratio estimator (8.11)-(8.13) of (8.2) is directly applicable to the evaluation of the voice blocking probabilities of (3.12a) and (3.12b). We need only set $K = |\mathcal{P}|, q(\underline{n}) = P(\underline{N}^v)$ of (3.9), k = p for $p \in \mathcal{P}, N_k = c_p/r_p$, and $q_k(n) = (\rho_p^v)^n/n!$ and estimate first the normalization constant G of (3.10). The key is of course the choice of the sampling function $P(\underline{n}) = P_s(\underline{n})$. In [7] it is suggested that we select

$$P_s(\underline{n}) = \frac{1}{G_s} \prod_{p \in \mathcal{P}} \frac{\gamma_p^{n_p^{\nu}}}{n_p^{\nu}!}$$
(8.17)

where

$$G_s = \prod_{p \in \mathcal{P}} \sum_{m=0}^{c_p/r_p} \frac{\gamma_p^m}{m!}.$$
(8.18)

The γ_p for $p \in \mathcal{P}$ are the important sampling parameters. These can be set as $\gamma_p = \rho_p^v$ or they can be selected according to a more tedious procedure (see [7]), which yields narrower confidence intervals. The estimator of (8.5)-(8.6) then takes the form

$$\hat{G} = \overline{Z}_n = \frac{1}{n} \sum_{i=1}^n \frac{q(\underline{V}^i)}{P_s(\underline{V}^i)} = \frac{G_s}{n} \sum_{i=1}^n \alpha^i I(\underline{V}^i \in \Omega^v)$$
(8.19)

where G_s is given by (8.18), Ω^v is defined by (3.11), and

$$\alpha^{i} = \prod_{p \in \mathcal{P}} \left(\frac{\rho_{p}^{v}}{\gamma_{p}} \right)^{V_{p}^{i}}$$
(8.20)

Once \hat{G} is obtained (as a function of the vector of link capacities <u>c</u>), the blocking probabilities B_l of (3.12a) and B_p of (3.12b) are obtained from those expressions where \hat{G} is used instead of G. Is is also possible to estimate B_p directly using the estimator (8.11)-(8.13) as

$$\hat{P}_{p} = \Phi_{n} = \frac{\sum_{i=1}^{n} Y^{i}}{\sum_{i=1}^{n} Z^{i}} = \frac{\sum_{i=1}^{n} \alpha^{i} I(\underline{V}^{i} \in \Omega_{p}^{v})}{\sum_{i=1}^{n} \alpha^{i} I(\underline{V}^{i} \in \Omega^{v})}$$
(8.21)

where Ω^{v} is given by (3.11) and Ω_{p}^{v} is obtained from (3.11) by replacing the $1 \times |\mathcal{L}|$ vector of link capacities <u>c</u> by the vector $\underline{c} - \underline{A}_{p}^{T}$ defined below (3.12b). The voice blocking probability P_{l} of (3.12a) can be estimated directly in the same manner. We may use (8.21) again with Ω_{l}^{v} instead of Ω_{p}^{v} in the numerator; Ω_{l}^{v} is obtained from Ω^{v} of (3.11) by replacing c_{l} by $c_{l} - 1$.

8.3 Monte-Carlo Summation for the Probability of Data Queueing

We rewrite $1 - Q_l$, where Q_l is the probability of data queued at link $l \ (l \in \mathcal{L})$ as follows,

$$1 - Q_l = \frac{\sum_{(\underline{N}^v, \underline{N}^s) \in \Lambda} f(\underline{N}^v, \underline{N}^s) \cdot q(\underline{N}^v, \underline{N}^s)}{\sum_{(\underline{N}^v, \underline{N}^s) \in \Lambda} q(\underline{N}^v, \underline{N}^s)}$$
(8.22)

where

$$q(\underline{N}^{v}, \underline{N}^{s}) = \prod_{p \in \mathcal{P}} \frac{(\rho_{1p}^{v})^{n_{p}^{v} - n_{p}^{s}} (\rho_{2p}^{v})^{n_{p}^{s}}}{(n_{p}^{v} - n_{p}^{s})! n_{p}^{s}!} \cdot I[0 \le \sum_{p \in \mathcal{P}_{l}} r_{p} n_{p}^{v} \le c_{l}, l \in \mathcal{L}]$$
(8.23)

$$f(\underline{N}^{v}, \underline{N}^{s}) = \sum_{n_{l}^{d}=0}^{c_{l}^{\prime}} P(n_{l}^{d}|c_{l}^{\prime}) \cdot I[(\sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{v}, \sum_{p \in \mathcal{P}_{l}} r_{p}n_{p}^{s}) \neq (c_{l}, 0)]$$

$$\Lambda = \{(\underline{N}^{v}, \underline{N}^{s})|0 \leq r_{p}n_{p}^{v} \leq c_{p}, 0 \leq n_{p}^{s} \leq n_{p}^{v}, p \in \mathcal{P}\}$$

$$(8.24)$$

and

$$\rho_{1p}^{\nu} = \frac{\beta}{\alpha + \beta} \cdot \rho_{p}^{\nu} \quad , \quad \rho_{2p}^{\nu} = \frac{\alpha}{\alpha + \beta} \cdot \rho_{p}^{\nu} \tag{8.25}$$

Then $1 - Q_l$ can be estimated by the ratio estimator of (8.11) as

$$\Phi_n = \frac{\sum_{i=1}^n Y^i}{\sum_{i=1}^n Z^i}$$

where

$$Y^{i} = \frac{f(\underline{V}^{i}, \underline{U}^{i}) \cdot q(\underline{V}^{i}, \underline{U}^{i})}{P_{s}(\underline{V}^{i}, \underline{U}^{i})}$$
(8.26)

$$Z^{i} = \frac{q(\underline{V}^{i}, \underline{U}^{i})}{P_{\bullet}(\underline{V}^{i}, \underline{U}^{i})}$$
(8.27)

and $(\underline{V}^i, \underline{U}^i), i = 1, ..., n$ is a sequence of i.i.d. random vectors, where each $(\underline{V}^i, \underline{U}^i)$ takes values in Λ ; $P_s(\underline{V}^i, \underline{U}^i)$ are importance sampling functions of the form

$$P_{s}(\underline{V}^{i} = \underline{N}^{v}, \underline{U}^{i} = \underline{N}^{s}) = \frac{1}{G_{s}} \cdot \prod_{p \in \mathcal{P}} \frac{\gamma_{1p}^{n_{p}^{v} - n_{p}^{s}} \gamma_{2p}^{n_{p}^{s}}}{(n_{p}^{v} - n_{p}^{s})! n_{p}^{s}!}, \quad (\underline{N}^{v}, \underline{N}^{s}) \in \Lambda$$
(8.28)

where

$$G_{s} = \prod_{p \in \mathcal{P}} \sum_{\substack{n_{p}^{v} = 0 \\ n_{p}^{v} = 0}}^{c_{p}/r_{p}} \sum_{\substack{n_{p}^{s} = 0 \\ n_{p}^{v} = 0}}^{n_{p}^{v} - n_{p}^{v}} \frac{\gamma_{1p}^{n_{p}^{v} - n_{p}^{v}} \gamma_{2p}^{n_{p}^{s}}}{(n_{p}^{v} - n_{p}^{s})! n_{p}^{s!}} = \prod_{p \in \mathcal{P}} \sum_{\substack{n_{p}^{v} = 0 \\ n_{p}^{v} = 0}}^{c_{p}/r_{p}} \frac{\gamma_{p}^{v}}{n_{p}^{v}!}$$
(8.29)

and the importance sampling parameters γ_{1p} and γ_{2p} are given by

$$\gamma_{1p} = \frac{\beta}{\alpha + \beta} \cdot \gamma_p \quad , \quad \gamma_{2p} = \frac{\alpha}{\alpha + \beta} \cdot \gamma_p. \tag{8.30}$$

With the above definitions and the definitions of \overline{Y}_n , \overline{Z}_n , $\sigma_n^2(Y)$, $\sigma_n^2(Z)$, and $\sigma_n^2(Y,Z)$ of (8.6), (8.9), and (8.14), respectively, we obtain from (8.15) that the $1 - \eta$ confidence interval for $1 - \Phi_n$ the estimate of $1 - Q_l$ is

$$\left(1 - \frac{\overline{Y}_n \overline{Z}_n - \frac{c^2(\eta)}{n} \sigma_n^2(Y, Z) + r_n}{\overline{Z}_n^2 - \frac{c^2(\eta)}{n} \sigma_n^2(Z)}, 1 - \frac{\overline{Y}_n \overline{Z}_n - \frac{c^2(\eta)}{n} \sigma_n^2(Y, Z) - r_n}{\overline{Z}_n^2 - \frac{c^2(\eta)}{n} \sigma_n^2(Z)}\right)$$
(8.31)

where $c(\eta)$ and r_n are as defined in (8.8) and (8.16).

Notice that $(V_1^i, U_1^i), (V_2^i, U_2^i), ..., (V_{|\mathcal{P}|}^i, U_{|\mathcal{P}|}^i)$ are independent, i.e.,

$$P_{s}(\underline{V}^{i} = \underline{N}^{v}, \underline{U}^{i} = \underline{N}^{s}) = \prod_{p \in \mathcal{P}} P_{s}(V_{p}^{i} = n_{p}^{v}, U_{p}^{i} = n_{p}^{s})$$
(8.32)

and

$$P_{s}(V_{p}^{i} = n_{p}^{v}, U_{p}^{i} = n_{p}^{s}) = P_{s}(V_{p}^{i} = n_{p}^{v}) \cdot P_{s}(U_{p}^{i} = n_{p}^{s}|V_{p}^{i} = n_{p}^{v}) \quad , \tag{8.33}$$

where

$$P_{s}(V_{p}^{i} = n_{p}^{v}) = \frac{\frac{\gamma_{p}^{n_{p}^{v}}}{n_{p}^{v!}}}{\sum_{\substack{r_{p} \neq 0 \\ n_{p}^{v} \neq 0}}^{c_{p}/r_{p}} \frac{\gamma_{p}^{n_{p}^{v}}}{n_{p}^{v!}}}, \quad 0 \le r_{p}n_{p}^{v} \le c_{p}$$
(8.34)

and

$$P_{s}(U_{p}^{i} = n_{p}^{s}|V_{p}^{i} = n_{p}^{v}) = \frac{\frac{\beta^{n_{p}^{i} - n_{p}^{i} \alpha n_{p}^{i}}{(\alpha + \beta)^{n_{p}^{i}}} \cdot n_{p}^{v}!}{(n_{p}^{v} - n_{p}^{s})!n_{p}^{s}!}, \quad 0 \le n_{p}^{s} \le n_{p}^{v}$$
(8.35)

Therefore, $|\mathcal{P}|$ of (V_p^i, U_p^i) can be generated independently by first generating V_p^i from $P_s(V_p^i)$ and then generating U_p^i from $P_s(U_p^i|V_p^i)$. The alias algorithm is used for both generating processes of V_p^i and U_p^i .

In our numerical results we simulate for a 95% confidence interval with n = 300,000based on the importance sampling scenario $\gamma_p = \rho_p^v, p \in \mathcal{P}$.

8.4 Monte-Carlo Summation for the Average Queueing Data Delay

This is the same as for the probability of data queueing except for the different function $f(\underline{N}^{v}, \underline{N}^{s})$. The new function is given by (7.5)-(7.6) in Section 7.

9. ADMISSION CONTROL VIA REDUCED-LOAD APPROXIMATIONS

Admission control strategies can play an important role in integrated voice/data networks because, by controlling in a coordinated manner the admission of new calls in all network circuits, the overall probability of voice blocking can be reduced and so may the probability of queueing data and the data packet delay. In this paper admission control strategies based on (a) thresholds on the individual path traffic and (b) linear combinations of voice traffic over selected sets of paths are considered. The justification for this is two-fold: First, such admission control strategies have been shown in [11]-[12] to work well for voice-only multi-hop radio networks of small size. Second, this type of admission control strategies enables the extension of the approximations of this report from problems without control to problems with control at the expense of only minimum additional computational complexity. Therefore, since our approximations can handle mid-size and even large-size multi-hop radio networks, near-optimal control strategies can be derived for networks of any size based on the approximate performance measures evaluated below.

After describing the relevant models for admission control in Section 9.1, we present in Sections 9.2 and 9.3 two distinct methods for incorporating admission control strategies into the approximation methods outlined in Sections 4 (for the probability of voice blocking) and in 5 and 6 (for the probability of data queueing). The development in these sections appears for the first time in this report. Only the modification of the knapsack approximation is described in detail. The Pascal approximation can be also modified in a similar way but this is omitted from the report.

9.1 Models for Admission Control

The problem of admission control is formulated in the context of our exposition in the previous sections, as follows. Together with the resource (bandwidth) constraints

$$\sum_{p \in \mathcal{P}_l} r_p n_p^v \le c_l, \qquad l \in \mathcal{L}$$
(9.1)

[where $r_p = 1$ for all $p \in \mathcal{P}$ and $c_l, \mathcal{P}_l, l \in \mathcal{L}$, must be replaced by T_n, \mathcal{P}_n for $n \in \mathcal{N}$ for the radio network of Section 2.1] which have played an important role in the previous analysis, we also consider the additional control constraints

$$\sum_{\boldsymbol{p}\in\mathcal{P}_{\boldsymbol{s}}}r_{\boldsymbol{p}}^{(\boldsymbol{s})}n_{\boldsymbol{p}}^{\boldsymbol{v}}\leq Y_{\boldsymbol{s}},\qquad \boldsymbol{s}\in S.$$
(9.2)

These control constraints result in the blocking of voice calls, even when network resources are available to support them. We refer to a system without such control constraints (i.e., one in which only the resource constraints of Eq. (9.1) are applicable) as an "uncontrolled" system. In Eq. (9.2) S is the set of all such constraints, Y_{\bullet} is the threshold for the s-th constraint, and \mathcal{P}_{\bullet} is the set of paths involved in the constraint. These control constraints couple the voice traffic over selected sets of paths. Moreover, direct threshold constraints of the form

$$n_p^v \le X_p, \qquad p \in \mathcal{P} \tag{9.3}$$

can be considered as special cases of the more general constraint of (9.2). Also notice that the quantities $r_p^{(s)}$ for $p \in \mathcal{P}_s$ involved in the s-th constraint need not be equal to (the data rate) r_p nor do they need to be equal to each other. Clearly, with this model any linearcombination type of constraint on the number of calls of classes $p \in \mathcal{P}$ can be considered. The objective is to determine the set of constraints (which collectively constitute a control policy) that results in optimal performance.

9.2 The Knapsack Approximation via Conditioning on Additional Constraints

We now show how to incorporate one additional control constraint into the knapsack approximation, which is obtained under the standard resource constraint of (9.1). The procedure can be extended to apply to two or more control constraints at the expense of an increasing computational effort due to the dimensionality of the problem.

For this section we assume that $S = \{s\}$ (a single-element set), that Y_s is the control threshold [it can actually be $Y_s/r^{(s)}$ for $r_p^{(s)} = r^{(s)}$ ($p \in \mathcal{P}_s$), when all $r_p^{(s)}$ are equal to each other], and that c_l is the capacity of the *l*-th link ($l \in \mathcal{L}$). We rewrite the two constraints (the resource constraint for link *l*, and the control constraint with threshold Y_s) as

$$0 \leq \sum_{p \in \mathcal{P}_l} r_p n_p^v = k_1 \leq c_l \tag{9.4}$$

$$0 \le \sum_{p \in \mathcal{P}_s} r_p^{(s)} n_p^v = k_2 \le Y_s \tag{9.5}$$

in terms of the quantities k_1 and k_2 representing the sums in (9.4) and (9.5), respectively.

9.2.1 Probability of Voice Blocking

Application of the knapsack approximation to the above formulation yields

$$K_{lp} = 1 - \sum_{k_1=0}^{c_l - r_p} \sum_{k_2=0}^{Y_s - r_p^{(s)}} P(k_1, k_2)$$
(9.6)

(for $p \in \mathcal{P}_l \cap \mathcal{P}_s$) as the voice blocking probability where

$$P(k_1, k_2) = \sum_{(\underline{N}^*, \underline{N}^*) \in \Omega(k_1, k_2)} P(\underline{N}^*, \underline{N}^*)$$
(9.7)

$$\Omega(k_1, k_2) = \left\{ (\underline{N}^{\boldsymbol{v}}, \underline{N}^{\boldsymbol{s}}) | 0 \le n_p^{\boldsymbol{s}} \le n_p^{\boldsymbol{v}}, p \in \mathcal{P}_l \cup \mathcal{P}_{\boldsymbol{s}}; 0 \le \sum_{\boldsymbol{p} \in \mathcal{P}_l} r_{\boldsymbol{p}} n_{\boldsymbol{p}}^{\boldsymbol{v}} = k_1 \le c_l; 0 \le \sum_{\boldsymbol{p} \in \mathcal{P}_s} r_{\boldsymbol{p}}^{(\boldsymbol{s})} n_{\boldsymbol{p}}^{\boldsymbol{v}} = k_2 \le Y_s \right\}$$

$$(9.8)$$

and $\sum_{k_1=0}^{c_l} \sum_{k_2=0}^{Y_*} P(k_1, k_2) = 1$. After the normalization $P'(k_1, k_2) = P(k_1, k_2)/P(0, 0)$, (9.6) becomes

$$K_{lp} = \begin{cases} 1 - \frac{\sum_{\substack{k_1 = 0 \\ k_1 = 0 \\ k_2 = 0 \\ k_1 = 0 \\ k_2 = 0 \\ p'(k_1, k_2) \\ k_1 = 0 \\ k_2 = 0 \\ p'(k_1, k_2) \\ k_1 = 0 \\ k_2 = 0 \\ p'(k_1, k_2) \\ k_1 = 0 \\ k_2 = 0 \\ p'(k_1, k_2) \\ k_1 = 0 \\ k_2 = 0 \\ p'(k_1, k_2) \\ k_1 = 0 \\ p \in \mathcal{P}_l \cap \mathcal{P}_s^c \end{cases}$$
(9.9)

Proposition 9.1

 $P'(k_1, k_2)$ satisfies the recursion

$$P'(k_1,k_2) = \begin{cases} \frac{1}{k_1} \cdot \frac{\beta}{\alpha+\beta} \left[\sum_{p \in \mathcal{P}_l \cap \mathcal{P}_s} r_p \rho_p^v P'(k_1 - r_p, k_2) + \sum_{p \in \mathcal{P}_l \cap \mathcal{P}_s} r_p \rho_p^v P'(k_1 - r_p, k_2 - r_p^{(s)}) \right]; \\ & \text{if } r_p \leq k_1 \leq c_l, r_p^{(s)} \leq k_2 \leq Y_s \\ \\ \frac{1}{k_1} \cdot \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_l \cap \mathcal{P}_s} r_p \rho_p^v P'(k_1 - r_p, k_2); \\ & \text{if } r_p \leq k_1 \leq c_l, k_2 = 0 \\ \\ \frac{1}{k_2} \cdot \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_s \cap \mathcal{P}_l^c} r_p^{(s)} \rho_p^v P'(k_1, k_2 - r_p^{(s)}) \\ & \text{if } k_1 = 0, r_p^{(s)} \leq k_2 \leq Y_s \\ \\ 1; \\ & \text{if } k_1 = 0, \ k_2 = 0 \\ \\ 0; \\ & \text{if } k_1 \text{ is not an integer multiple of } r_p \\ & \text{or } k_2 \text{ is not an integer multiple of } r_p^{(s)} \end{cases}$$

(9.10)

where $r_p \leq k_1 \leq c_l$ and $r_p^{(s)} \leq k_2 \leq T_s$.

Proof: It follows easily as a special case of of Appendix B; we only need to set $r_p^{(1)} = r_p$, $r_p^{(2)} = r_p^{(s)}$, $\mathcal{P}_1 = \mathcal{P}_l$, $\mathcal{P}_2 = \mathcal{P}_s$, $Z_1 = c_l$, $Z_2 = Y_s$, and use only the constraints on k_i ; we set $m_i = 0$ for i = 1, 2.

9.2.2 Probability of Data Queueing

The application of the knapsack approximation to the probability of queueing data when conditions (9.1) and (9.2) are present requires an extension of the method of Section 5.2 to incorporate additional dimensions. In particular, instead of the k_i , m_i for 1,2 defined and used in Section 5.2 to represent the necessary integer entities for nodes $n_1 = 1$ and $n_2 = 2$ (connected by link l) and which reflect only the bandwidth (or transceiver) conditions, we must introduce additional such entities to represent the threshold condition of (9.2). Actually, we need a total of six integers $(k_1, m_1; k_2, m_2; k_3, m_3)$ to represent all necessary entities: (k_1, m_1) for the transceiver constraint on the set $\mathcal{P}_1 \cap \mathcal{P}_s$, (k_2, m_2) for the transceiver constraint on the set $\mathcal{P}_2 \cap \mathcal{P}_s$, and (k_3, m_3) for the threshold constraint on the set \mathcal{P}_s ; compare with (9.4)-(9.5) and (5.15). This requires the definition of a sixdimensional entity $P(k_1, m_1; k_2, m_2; k_3, m_3)$ similar to that of (5.16) and the corresponding recursion. The details are complicated but straightforward and are omitted. The rest of the knapsack approximation for this case proceeds as in Section 5.2.

9.2.3 Extension to Multiple Control Constraints

The formulation, the knapsack approximations, and the recursions for the probabilities of voice blocking and data queueing can be extended to include additional control constraints of the form (9.2) for several distinct $s \in S$. For a small cardinality number |S|the computational effort, although increasing exponentially (as $2^{|S|}$) with the total number of control constraints, remains reasonable; however, for large |S| it becomes prohibitive. This is the reason for our consideration in Section 9.3 of a different approach that requires a substantially reduced computational effort.

9.3 The Knapsack Approximation via the Introduction of Fictitious Links

The second method for applying the knapsack approximation to integrated voice/data networks with admission control is very straightforward and does not suffer from the computational problems of the first. Actually, the required computational effort grows only linearly with the number of control constraints. The key idea here is to introduce fictitious links l_s with link capacity Y_s that represent each of the control constraints ($s \in S$) of (9.2). On these fictitious links the paths carry information at a rate $r_p^{(s)}$ (which may not be equal to r_p). In this manner all voice calls transmitted over the paths (circuits) which satisfy a particular control constraint (i.e., $p \in \mathcal{P}_s$) use this fictitious link, that is $l_s \in p$ for all $p \in \mathcal{P}_s$. This means that besides the natural links of the network (part of the given architecture) new links are added which do not represent any physical connection but rather they signify satisfaction of control constraints. After the set of links \mathcal{L} has been expanded to include the new links, that is $\mathcal{L}' = \mathcal{L} \cup \mathcal{S}$, we apply the usual procedure for the knapsack approximation to the new expanded set of links \mathcal{L}' . For example for the voice blocking probability we use the recursions

$$L_{l'q} = K_{l'q} \left(c_{l'}; \rho_p^v \prod_{\ell' \in p, \ell' \neq l'} (1 - L_{\ell'p}), \ p \in \mathcal{P}_{l'}, \ l' \in \mathcal{L}' = \mathcal{L} \cup \mathcal{S} \right)$$

for $q \in \mathcal{P}_{l'}$, where $\mathcal{P}_{l'} = \{p \in \mathcal{P} | l' \in p, l' \in \mathcal{L}'\}$ and finally $B_q = 1 - \prod_{l' \in q} (1 - L_{l'q})$. In computing $K_{l'q}$ [for example via (4.5)-(4.6)] we must be careful to use (c_l, r_p) for the real links (when $l' \in \mathcal{L}$) and $(Y_s, r_p^{(s)})$ for the fictitious links (when $l' \in S$). The above approach is also applicable to the approximation of the probability of data queueing and the average data delay (Section 5) after the expansion of \mathcal{L} to \mathcal{L}' has taken place.

10. REVENUE SENSITIVITY VIA REDUCED-LOAD APPROXIMATIONS

In several practical problems of performance evaluation and optimization in networks such as optimal data routing and the allocation of additional resources in response to increases in traffic demand, the rates of change or derivatives (termed sensitivities) of certain performance measures (termed revenue) with respect to network resources (e.g., link capacities) and traffic loads (e.g., average offered voice or data traffic) play an important role. If accurate approximations to these sensitivities that require modest computational effort for their evaluation can be derived, then they can be used in a variety of optimization problems to derive near-optimal control or allocation strategies. This is exactly what is accomplished in this section.

Specifically, we first present certain popular measures of revenue and then derive their (approximate) sensitivities with respect to link capacities, voice loads, and data loads. This is first accomplished in Section 10.1 for the general multi-rate network model (of Sections 2.1.1 and 2.2.1); in particular, Section 10.1.1 deals with suitable sensitivities for voice revenue and Section 10.1.2 deals with suitable sensitivities for data revenue. Then in Section 10.2 we outline how to modify and apply these results to the r ulti-hop radio network model of Sections 2.1.2 and 2.2.2.

10.1 Revenue Sensitivities for General Multi-Rate Networks

Suitable choices for the long-run average voice and data revenues for the multi-rate network of Sections 2.1.1 and 2.2.1 are

$$W^{\nu}(\underline{\rho}^{\nu};\underline{c}) = \sum_{p \in \mathcal{P}} \gamma_{p}^{\nu} \rho_{p}^{\nu} (1 - B_{p})$$
(10.1)

$$W^{d}(\underline{\rho}^{d}, \underline{\rho}^{v}; \underline{c}) = \sum_{l \in \mathcal{L}} \gamma_{l}^{d} \rho_{l}^{d} (1 - Q_{l}), \qquad (10.2)$$

where γ_p^v $(p \in \mathcal{P})$ and γ_l^d $(l \in \mathcal{L})$ is the revenue rate when a path-*p* call is accepted and link- ℓ data traffic is served, respectively. Thus (10.1) and (10.2) provide natural definitions of the (long-term) average revenue generated when voice calls are accepted (not blocked) and data packets are served (not queued), respectively. In the above notation $\underline{\rho}^v$ and $\underline{\rho}^d$ denote the vectors of voice and data loads $(\rho_p^v, p \in \mathcal{P} \text{ and } \rho_l^d, l \in \mathcal{L})$ and \underline{c} the vector of link capacities $(c_l, l \in \mathcal{L})$. In this section it is assumed that the probabilities of voice blocking B_p and data queueing Q_l are evaluated via the knapsack approximation.

10.1.1 Sensitivity of Voice Revenue

The sensitivity of the voice revenue with respect to link capacities and voice loads is evaluated in [6] (see also [7]) via the knapsack and Pascal approximation methods. Here we repeat the results for the sake of completeness and to introduce the necessary notation.

For the voice revenue the following sensitivity measures are defined with respect to link capacity and voice loads:

$$c_{lp}^{\nu} \simeq W^{\nu} \left(\underline{\rho}^{\nu}; \underline{c} - r_{p} \underline{e}_{p} \right) - W^{\nu} (\underline{\rho}^{\nu}; \underline{c})$$
(10.3)

$$\frac{dW^{v}}{d\rho_{p}^{v}} \simeq (1 - B_{p}) \cdot (\gamma_{p}^{v} - \sum_{l \in p} c_{lp}^{v})$$

$$(10.4)$$

where \underline{e}_p is an $|\mathcal{L}| \times 1$ vector with unit value at entries $l \in p$ and zero value at all others.

The quantities c_{lp}^{v} in (10.3) have been called shadow prices in [6] and represent the expected revenue lost (number of blocked voice calls) when we remove r_{p} circuits from link l (or accept a call of path p at link l) for one unit of time. If we make the approximation that class-q calls arrive at link l according to a Poisson process with offered load

$$\bar{\rho}_q^v = \rho_q^v \prod_{\ell \in q, \ell \neq l} (1 - L_{\ell q})$$

(termed thinned load, with L_{lq} as defined in Section 4.2), then the expected loss in revenue from class-q connections being blocked at link l due to the removal of r_p circuits for one unit of time is $\bar{\rho}_q^v \cdot h_{lqp} \cdot \gamma_q$, where

$$h_{lqp} = K_{lq} \left(c_l - r_p, \rho_{q'}^v \prod_{\ell \in q', \ell \neq l} (1 - L_{\ell q}), q' \in \mathcal{P}_l \right) - K_{lq} \left(c_l, \rho_{q'}^v \prod_{\ell \in q', \ell \neq l} (1 - L_{\ell q'}), q' \in \mathcal{P}_l \right)$$
(10.5)

where K_{lq} is given by (4.5)-(4.6). However, each class-q connection blocked on link l (if it had been accepted), would have used r_q circuits on each link $l \in q, l \neq l$. Thus the expected gain in revenue from additional connections being accepted on links $l \in q, l \neq l$ due to a class-q connection being blocked at link l for this one unit of time is $\bar{p}_q^v \cdot h_{lqp} \cdot \sum_{j \in q, j \neq l} c_{jq}^v$. Subtracting the gain from the loss and summing over $q \in \mathcal{P}_l$ yields the following system of linear equations:

$$c_{lp}^{v} \simeq \sum_{q \in \mathcal{P}_{l}} \left[\bar{\rho}_{q}^{v} K_{lq} \left(c_{l} - r_{p}, \bar{\rho}_{q'}^{v}, q' \in \mathcal{P}_{l} \right) - \bar{\rho}_{q}^{v} \bar{K}_{lq} \left(c_{l}, \bar{\rho}_{q'}^{v}, q' \in \mathcal{P}_{l} \right) \right] \cdot \left(\gamma_{q} - \sum_{\ell \in q, \ell \neq l} c_{\ell q}^{v} \right),$$

$$\simeq \sum_{q \in \mathcal{P}_l} \left[\rho_q^v \left(1 - B_q(\underline{c}) \right) - \rho_q^v \left(1 - B_q \left(\underline{c} - r_p \underline{e}_p \right) \right) \right] \cdot \left(\gamma_q - \sum_{\ell \in q, \ell \neq l} c_{\ell q}^v \right).$$
(10.6)

In deriving (10.6) we first use the results of Section 4.2 to show that the relationship between the load ρ_q^v and the thinned load $\bar{\rho}_q^v$ on link l is

$$\bar{\rho}_q^v(1-L_{lq})=\rho_q^v(1-B_q)$$

and from this it follows that

$$\bar{\rho}_q^{v} K_{lq}\left(c_l, \bar{\rho}_{q'}^{v}, q' \in \mathcal{P}_l\right) = \bar{\rho}_q^{v} L_{lq} = \bar{\rho}_q^{v} - \rho_q^{v}\left(1 - B_q(\underline{c})\right).$$

Finally, the right hand side term in (10.4) can be interpreted as follows. An additional call offered to route p will be accepted with probability $(1 - B_p)$; if accepted it will earn γ_p^v revenue, but at a cost c_{lp} for each link $l \in p$.

10.1.2. Sensitivity of Data Revenue

In contrast to the previous section that reviewed the results of [6] for the sensitivities of the voice revenue, the derivation of sensitivities for the data revenue of integrated voice/data multi-rate networks appears for the first time in this section. The following sensitivity measures with respect to link capacities and data loads, respectively, are suggested for the data revenue of (10.2)

$$c_l^d \simeq W^d(\underline{\rho}^v, \underline{\rho}^d; \underline{c}) - W^d(\underline{\rho}^v, \underline{\rho}^d; \underline{c} - \underline{e}_l)$$
(10.7)

$$\frac{\partial W^d}{\partial \rho_l^d} \simeq (1 - Q_l) \cdot (\gamma_l^d - c_l^d) \tag{10.8}$$

where \underline{e}_l is an $|\mathcal{L}| \times 1$ vector with unit value at the entry l (the link in question) and zero value at all other entries.

Since link data behavior is assumed to be mutually independent (refer to Section 2.2.1), the term in (10.7) can be obtained in the form

$$c_l^d \simeq \left[\rho_l^d (1 - Q_l(\underline{c})) - \rho_l^d (1 - Q_l(\underline{c} - \underline{e}_l))\right] \cdot \gamma_l^d \tag{10.9}$$

following similar reasoning to that which resulted in (10.6) above for the voice revenue shadow prices. However, because of the inter-link independence assumption the approximation in (10.9) can not be fully trusted.

The term in (10.8) again can be obtained intuitively by noting that additional data offered to link l will not be queued with probability $(1 - Q_l)$; if accepted, it will earn γ_l revenue but at a cost c_l^d .

Instead of obtaining $\partial W^d / \partial \rho_\ell^d$ from (10.8) for which (10.9) is necessary, an alternative way of computing $\partial W^d / \partial \rho_\ell^d$ is used in this report, according to which

$$\frac{\partial W^d}{\partial \rho_\ell^d} \simeq \gamma_l^d (1 - Q_l) + \gamma_l^d \rho_l^d \frac{\Delta (1 - Q_l)}{\Delta \rho_l^d}$$
(10.10)

where $\Delta(1-Q_l)/\Delta\rho_l^d$ denotes the ratio of finite differences for $\Delta\rho_l^d = .001$.

The evaluation of the data revenue sensitivity with respect to voice load is considerably more complicated. The result is given by the following Proposition:

Proposition 10.1

The data revenue sensitivity with respect to voice loads takes the form

$$\frac{\partial W^{d}}{\partial \rho_{p}^{v}} = (1 - B_{p}) \left\{ \frac{\beta}{\alpha + \beta} \cdot W^{d}(\underline{c} - r_{p}\underline{e}_{p}) - W^{d}(\underline{c}) + \frac{\alpha}{\alpha + \beta} \cdot \frac{1}{G(\underline{c} - r_{p}\underline{e}_{p})} \sum_{(\underline{N}^{v}, \underline{N}^{s}), \in \Omega(\underline{c} - r_{p}\underline{e}_{p})} \prod_{p \in \mathcal{P}} \rho_{p}^{n_{p}^{v}} \frac{\frac{\alpha^{n_{p}^{s}} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v}}}}{n_{p}^{s!}(n_{p}^{v} - n_{p}^{s})!} \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \right\}$$

$$= (1 - B_{p}) \left[\frac{\beta}{\alpha + \beta} \cdot W^{d}(\underline{c} - r_{p}\underline{e}_{p}) - W^{d}(\underline{c}) + \frac{\alpha}{\alpha + \beta} \cdot \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p}\underline{e}_{p})} P_{(\underline{c} - r_{p}\underline{e}_{p})}(\underline{N}^{v}, \underline{N}^{s}) \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \right]. \quad (10.11)$$

where G is given by (3.3), $\bar{P}(c'_l, \rho^d_l)$ is given by (5.6), the residual capacity for data is $c'_l = c_l - \sum_{p \in \mathcal{P}_l} r_p(n^v_p - n^s_p)$, and

$$w^{d}(\underline{\gamma}^{d},\underline{\rho}^{d},\underline{c}) = \sum_{l \in \mathcal{L}} \gamma_{l}^{d} \rho_{l}^{d} \cdot \bar{P}(c_{l}^{\prime},\rho_{l}^{d}) \cdot I(c_{l}^{\prime}>0).$$
(10.12)

The terms in (10.11) and (10.12) can be approximated by applying the knapsack or Pascal methods. In particular, the knapsack approximation is applied to the Q_l terms in

$$W^{d}(\underline{c} - r_{p}\underline{e}_{p}) = \sum_{l \in \mathcal{L}} \gamma_{l}^{d} \rho_{l}^{d} (1 - Q_{l}(\underline{c} - r_{p}\underline{e}_{p})), \qquad (10.13)$$

$$W^{d}(\underline{c}) = \sum_{l \in \mathcal{L}} \gamma_{l}^{d} \rho_{l}^{d} (1 - Q_{l}(\underline{c})), \qquad (10.14)$$

and to the term

$$\sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(\underline{c}-r_{p}\underline{c}_{p})} P_{(\underline{c}-r_{p}\underline{c}_{p})}(\underline{N}^{v},\underline{N}^{s})w^{d}(\underline{\gamma}^{d},\underline{\rho}^{d},\underline{c})$$

$$=\sum_{l\in\mathcal{L}}\gamma_{l}^{d}\rho_{l}^{d}\sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(\underline{c}-r_{p}\underline{c}_{p})} P_{(\underline{c}-r_{p}\underline{c}_{p})}(\underline{N}^{v},\underline{N}^{s}) \cdot \sum_{n_{l}^{d}=0}^{c_{l}^{\prime}-1} P(n_{\ell}^{d}|c_{\ell}^{\prime}) \cdot I(c_{\ell}^{\prime}>0). \quad (10.15)$$

Proof: Provided in Appendix H

10.2. Revenue Sensitivities for the Multi-Hop Radio Network of Sections 2.1.2 and 2.2.2

Recall that for the multi-hop radio network of Section 2.1.2 and 2.2.2, the link capacity vector \underline{c} should be replaced by the node transceiver vector \underline{T} and the rate of voice calls in all circuits is $r_p = 1$, $p \in \mathcal{P}$.

To obtain the sensitivities of the voice revenue we must replace \underline{c} by \underline{T} , c_l by T_n , $l \in \mathcal{L}$ (links) by $n \in \mathcal{N}$ (nodes), \mathcal{P}_l (set of all circuits or paths using link l) by \mathcal{P}_n (set of all paths passing by node n), and $\underline{c} - r_p \underline{e}_p$ by $\underline{T} - \underline{e}_p$ (where \underline{e}_p is an $|\mathcal{N}| \times 1$ vector with unit value at entries $n \in p$ and zero value at all others) in (10.3), (10.4), (10.6) and all other expressions of Section 10.1.1.

To obtain the sensitivities of the data revenue we must carry out the same substitutions in (10.10) and in (10.11)-(10.14) as we did for the voice revenue sensitivities in the previous paragraph. In (10.11)-(10.14) the following additional substitution is necessary: the residual (data) capacity

$$c'_l = c_l - \sum_{p \in \mathcal{P}_l} r_p (n_p^v - n_p^s)$$

of link $l \in \mathcal{L}$ connecting nodes $n_1, n_2 \in \mathcal{N}$ [i.e., $l = (n_1, n_2)$] must be replaced by

$$c'_{l} = \min\{T_{n_{1}} - \sum_{p \in \mathcal{P}_{n_{1}}} (n^{v}_{p} - n^{s}_{p}) , T_{n_{2}} - \sum_{p \in \mathcal{P}_{n_{2}}} (n^{v}_{p} - n^{s}_{p})\}.$$
11. NUMERICAL RESULTS AND PERFORMANCE COMPARISONS

The presentation of the numerical results in this section is organized as follows. In Section 11.1 the network paradigms used for general multi-rate wired networks and for a radio multi-hop network are described in detail in terms of architectures, network parameters, and traffic parameters. In Section 11.2 comparisons of the approximations to and the exact values of the performance measures of interest are carried out for the radio network paradigm with no admission control. This is repeated in Section 11.3 for the general multi-rate multi-hop wired network paradigm. Admission control schemes for voice traffic (based on thresholds) in the radio network paradigm are described in Section 11.4. Section 11.5 presents the approximations to the voice and data revenue sensitivities for the radio and wired network paradigms of Section 11.1. Finally, Section 11.6 discusses the required computational effort for the various approximations.

11.1 Network and Traffic Models for the Paradigms Used

The two networks of interest are shown in Figures 2a-2b and 3a-3b. The network of Figure 2a is the paradigm shown in [11]-[12]. It is a ten-node multi-hop radio network, in which the main resource of interest is the number of transceivers at each of the network nodes. In most of our examples involving this network, \underline{T} (which denotes the vector of transceivers at the ten network nodes) takes the value $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$; we also study $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4, 4)$ to demonstrate the quality of the approximations when applied to a smaller system. In Figure 2b, the five circuits used by voice calls in the network of Figure 2a are shown. The voice calls follow the model of Sections 2.2.1 and 2.2.2 and have in all examples (unless specified otherwise) activity factor $\frac{\beta}{\alpha+\beta} = .4$ (half-duplex); however; limited results are also presented for voice activity factors of 0.8 (full-duplex) and 1.0 (corresponding to a voice model with no silence periods). The network data traffic is transmitted over the same nine links, that are used by the aforementioned five voice circuits. As discussed in Section 2.2.2, an M/D/c model is used for data traffic over the above nine links, with capacities defined by Eq. 2.0. In this model, the data can only use transceivers that are not occupied by voice calls. In the model of Section 7.3, some transceivers at every node are a priori dedicated to data traffic and the data link capacity is defined by (7.18); we will come back to this model when we show results for the average data delay. The network of Figures 2a-2b is characterized by a single data rate, which assumes a common value for both voice and data traffic.

Figures 3a and 3b, show the multi-rate star network of [6], which is used in this section as a paradigm of a general-purpose integrated voice/data multi-rate wired network. The main network resource is the vector of link capacities \underline{c} . Initially we reproduce the example of [6] by setting $\underline{c} = (90, 100, 110, 120)$ for the four links of the network; later we demonstrate the accuracy of the approximations for smaller values of the link capacities as well. There are twelve voice circuits in the network of Figure 3a; and Figure 3b lists the links used and the transmission bandwidth required for each voice circuit. Data traffic is assumed to always require unit transmission bandwidth. The voice model is described in Section 2.2.1 and the activity factor for the voice calls is $\frac{\beta}{\alpha+\beta} = .4$. An M/D/c model is again used for data traffic with link capacity given by (3.7b). In this model, data traffic is allowed to use only the capacity left unused in the four network links after the voice call requests have been accommodated. In the model of Section 7.3, some portion of the capacity of each link is a priori dedicated to data traffic and the data link capacity is given by (7.17); we will come back to this model when we show results for the average data delay.

11.2 Comparisons of Approximations for the Radio Network of Figure 2 without Admission Control

This group of numerical results pertains to the radio network model of Figure 2 in the absence of admission control, i.e., a voice call is admitted if and only if a transceiver is available at every node along the predetermined path. Networks with eight transceivers per node are evaluated in Section 11.2.1 and networks with four transceivers per node are evaluated in Section 11.2.2. In these two subsections, the entire number of transceivers at each node is available for use by the voice traffic; data traffic is served only if some transceivers remain unoccupied. However, in Section 11.2.3 we also evaluate networks for which one pair of transceivers is dedicated to data traffic for each of the nine data links in the network of Figure 2.

11.2.1 Results for a Radio Network with Eight Transceivers per Node

We first consider the case of $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, i.e., each node has eight transceivers. Figures 4 and 5 show the exact value, knapsack approximation, and Monte Carlo summation (the midpoint of the confidence interval is shown) for the average probabilities of voice blocking and data queueing, versus the offered voice and data loads, respectively. The "exact value" solutions for the probability of voice blocking are based on the product-form solution. Those for data packet queueing are based on the M/D/c model

and Kleinrock's independence assumption, thus in the latter case the results are not truly exact but rather they are based on closed-form approximations. In both figures, the probabilities of voice blocking and data queueing represent the averages of such probabilities over all voice circuits and data links in the radio network, respectively, that is,

$$\bar{B} = \frac{\sum_{p \in \mathcal{P}} \rho_p^v B_p}{\sum_{p \in \mathcal{P}} \rho_p^v}$$

and

$$\bar{Q} = \frac{\sum_{l \in \mathcal{L}} \rho_l^d Q_l}{\sum_{l \in \mathcal{L}} \rho_l^d}.$$

In Figure 4, the value of the data load is irrelevant (since voice has priority over data) and the offered voice load $\rho_p^v r_p$ ranges from 0 to 15. In this case, we have a single-rate network with $r_p = 1$ for all paths, and ρ_p^v takes the same value for all five paths (voice circuits). Notice that the knapsack approximation, the Monte Carlo summation method, and the exact expression for the probability of voice blocking yield results that are very close to each other for all values of the offered voice load. Recall that the knapsack and Pascal approximations yield identical results for single-rate networks.

In deriving the approximation via the Monte Carlo summation method, we used 300,000 sample points (calls to the random number generator computer routines), which resulted in a confidence interval of 95%. Importance sampling scenarios according to which $\gamma_p = \rho_p^v (p \in \mathcal{P})$ for the probability of voice blocking (refer to Section 8.2) and $\gamma_{1p} = \rho_p^v \beta/(\alpha + \beta)$, $\gamma_{2p} = \rho_p^v \alpha/(\alpha + \beta)$ ($p \in \mathcal{P}$) for the probability of data queueing (refer to Section 8.3) were used. Recall from the discussion in Section 8 [following equations (8.7)-(8.10) and (8.14)-(8.15)] that the confidence interval parameter η and the number of samples (calls of the random number generator) n are related in a complicated manner [e.g., see (8.10)], which involves the sample variance $\sigma_n(Z)$; therefore, the appropriate value of n that guarantees the desirable confidence interval is found through trial and error. The number 300,000 given above represents the result of several such attempts to find a number that is sufficiently large to work for most situations of interest (for all different values of the sample variances for the numerical examples considered). The variation of the 95% confidence interval did not exceed 2% of its mean value during this search.

Similar results are shown in Figure 5, in which the offered voice load is $\rho_p^v r_p = 5.5$ for all voice circuits, and the offered data load ρ_l^d ranges from 0 to 4. The data loads of

all nine links are assumed equal. Again, the knapsack approximation is very close to the exact value of the probability of data queueing.

Tables 1-4 amplify the results of Figures 4 and 5 by providing detailed comparisons of exact values, knapsack approximation, and Monte Carlo summation for the probabilities of voice blocking and data queueing for a broad range of offered voice and data loads. The column denoted "percent error" refers to the percent relative error generated via a comparison of the results obtained using the knapsack approximation to the exact values [= 100 x (knapsack-exact)/(exact)]; the sign of the relative error has been maintained in these calculations. The results are organized in a manner that first presents the results for the average quantities (averaged over the traffic loads of all paths or links) and then the individual results for each path or link. This organization plan is followed for basically all the numerical results presented in tabular form in this report. Table 1 shows the average probability of voice blocking \overline{B} (over all paths) for different average voice loads $\overline{\rho}^{\nu}$ defined as

$$\bar{\rho}^{v} = \frac{1}{|\mathcal{P}|} \sum_{p \in \mathcal{P}} r_{p} \rho_{p}^{v}.$$

Then Table 2 shows the individual probabilities of voice blocking B_p for each of the paths $p \ (p \in \mathcal{P})$ for three values of the path voice loads ρ_p^v ; recall that $r_p = 1$ for the radio network of Figure 2.

Similarly, Table 3 shows the average probability of data queueing \bar{Q} (over all links) for different average voice loads $\bar{\rho}^v$, average data loads $\bar{\rho}^d$ defined as

$$\bar{\rho}^d = \frac{1}{|\mathcal{L}|} \sum_{l \in \mathcal{L}} \rho_l^d$$

and for the three voice activity factor values 0.4, 0.8, and 1.0.

Then Tables 4a, 4b, and 4c show the individual probabilities of data queueing Q_l for each of the links $l \ (l \in \mathcal{L})$ for various data loads ρ_l^d , for voice activity factors 0.4, 0.8, and 1.0, and for voice loads 0.1, 1.0, and 2.0, respectively.

In the case of the single-rate radio network of Figure 2, since $r_p = 1$ for all $p \in \mathcal{P}$, if the utilizations ρ_p^v are equal for all paths, then $\bar{\rho}^v = \rho_p^v$ as well. Similarly, equal values of ρ_l^d for all data links imply that $\bar{\rho}^d = \rho_l^d$ as well. Therefore, since in all tables of this subsection the ρ_p^v 's are equal to each other for all voice paths and the ρ_l^d 's are equal to each other for all data links, no distinction is made in the text between the values of ρ_p^u and $\bar{\rho}^v$ or between the values of ρ_l^d and $\bar{\rho}^d$. In any event the caption of each table contains all necessary information about the voice and data loads.

In particular, in Table 1, the approximations and the exact value of average probability of voice blocking are compared for different values of the average offered voice load ($\bar{\rho}^v = \rho_p^v$, $p \in \mathcal{P}$). It is observed that the agreement between the approximations and the exact value is from excellent to very satisfactory over the entire range of values of the offered voice load considered.

We take the opportunity here to quantify the terms excellent, very satisfactory, and satisfactory (or fair) pertaining to approximation accuracy and used throughout this section. By excellent accuracy we mean that relative error between the approximation and the exact value is smaller than 1%; by very satisfactory we mean that the relative approximation error is smaller than 5%; and by satisfactory (or fair) we mean that the relative approximation error is most of the time smaller than 10% and occasionally between 10% and 20%. In all of our examples, the exact value always falls within the confidence interval provided by the Monte-Carlo summation method.

In Table 2, the results of Table 1 are shown in greater detail for the voice-blocking probabilities of each of the five voice circuits of the radio network and for offered voice loads equal to 1.0, 5.5, and 10.0 for all paths. Again, the approximations are very accurate for all voice circuits, although a variation in accuracy is observed from circuit to circuit.

Similarly, in Table 3, the knapsack approximation and the exact value of the average probability of data queueing are compared for different values of the offered voice and data load. For each of these tables the results are shown for data loads ($\bar{\rho}^d = \rho_l^d$, $l \in \mathcal{L}$) of 0.5, 2.5, and 4.0 (corresponding roughly to situations of light, moderate, and heavy data traffic) and for several values of the voice loads ($\bar{\rho}^v = \rho_p^v$, $p \in \mathcal{P}$). The accuracy of the knapsack approximation remains very satisfactory over the entire range of traffic parameters of interest.

Tables 4a - 4c show certain results of Table 3 in greater detail for the probabilities of data queueing at each of the nine data links of the radio network of Figure 2. The results are compared for the three voice activity factor values 0.4, 0.8, and 1.0 for voice loads 0.1 (Table 4a) and 1.0 (Table 4b) and data loads 0.5, 2.5, and 4.0. Results for percent error are included only for queueing probabilities greater than 0.0005. Again, the knapsack

maintains from excellent to satisfactory accuracy over the entire range of traffic parameters of interest.

The results of Tables 2 and 3 (for probability of voice blocking) and Tables 3 and 4 (for probability of data queueing) demonstrate that the knapsack approximation is accurate not only when averages are taken over all circuits (paths) or links, but also when the network is examined at a more detailed "microscopic" level.

11.2.2 Results for a Network with Fewer Transceivers per Node

Table 6 shows the knapsack approximation and the exact value of the average probability of data queueing for different voice and data loads. Similar trends to those observed for the average voice-blocking probability are observed here as well. Table 7 shows in detail certain results of Table 6 for the probabilities of data queueing of each of the nine data links of the radio network and, in particular, for the values 0.2, 1.0, and 2.0 of offered data load and the values 0.1 (in Table 7a), 1.0 (in Table 7b), 5.0 (in Table 7c), and 10.0 (in Table 7d) of the voice load. Again, the knapsack provides satisfactory performance, although the agreement with the exact values is not as good as for the larger system of Section 11.2.1.

11.2.3 Results for a System with Transceivers Reserved for Data

The last group of Tables in this subsection (Tables 8, 9, and 10) pertains to the radio network model of Figure 2 modified according to requirements of Section 7.3 so that finite data delays can be guaranteed. According to this model, a pair of transceivers for each of the nine data links is dedicated to data traffic; that is, the data link capacity

vector $\underline{c}^d = (1, 1, 1, 1, 1, 1, 1, 1, 1)$ is guaranteed a priori. Thus, instead of the original node transceiver vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, the five voice circuits of Figure 2b can only use the transceiver vector $\underline{T}^v = (5, 7, 7, 7, 4, 7, 4, 7, 7, 7)$ that remains after the data transceivers have been assigned. The voice activity factor is $\beta/(\alpha + \beta) = 0.4$ for all results in this subsection.

In Table 8, the Monte-Carlo summation and knapsack approximations are compared to the exact value of the blocking probability for each of the five voice circuits and for the average value for a voice load of 2.5. The accuracy of the approximations remains satisfactory but it is inferior to that of the Tables 1 and 2 for which the voice transceiver vector was $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$; the smaller number of transceivers available for voice traffic degrades somewhat the accuracy of the approximations. In Tables 9 and 10, the average data delay

$$\bar{W} = \frac{\sum_{l \in \mathcal{L}} \rho_l^d W_l}{\sum_{l \in \mathcal{L}} \rho_l^d}$$

(measured in terms of packet duration prior to starting transmission) and the delays of each of the nine data links W_l $(l \in \mathcal{L})$, respectively, are shown in terms of the exact value and the knapsack approximation for values 2.5 and 10.0 of the offered voice load and 0.7, 0.9, and .999 of the offered data load. Again, as in the case of the probability of queueing data, the "exact value" for the queueing delay is actually a closed-form approximation based on the M/D/c model and Kleinrock's independence assumption. Satisfactory accuracy is observed in all cases.

11.3 Comparisons of Approximations for the Multi-Rate Network of Figure 3 without Admission Control

Figures 6 and 7 and Tables 11 to 14 pertain to performance results and comparisons of approximations for the multi-rate network of Figure 3, in the absence of admission control. The twelve voice circuits of this network (see Figure 3b) have rates $\underline{r}^{v} = (1,1,1,1,1,1,5,5,5,5,5,5,5,5)$ and the four data links have rates $\underline{r}^{d} = (1,1,1,1)$. Figures 6 and 7 pertain to the multi-rate network of Figures 3a and 3b with the link capacity vector $\underline{c} = (18, 20, 22, 24)$. The voice activity factor is $\beta/(\alpha + \beta) = 0.4$ for all results in this subsection.

In Figure 6, the average voice blocking probability is depicted versus the offered voice load $\rho_p^v r_p$ over the range 0 to 3. The load $\rho_p^v r_p$ takes the same value for all twelve voice

circuits. Whenever $\rho_p^v r_p$ $(p \in \mathcal{P})$ assumes a particular value, e.g., 1.5, the corresponding value of ρ_p is obtained from $1.5/r_p$ for each $p \in \mathcal{P}$, where r_p takes values 1 or 5 according to the vector of voice path (circuit) rates above.

For all tables in this subsection the offered loads (ρ_p^v) of the first six voice paths with $r_p = 1, p = 1, 2, ..., 6$ are equal to each other, and the offered loads of the remaining six voice paths with $r_p = 5, p = 7, 8, ..., 12$ are also equal to each other and equal to one fifth of the common value of the first six. This implies that the average voice load $\bar{\rho}^v = \rho_1^v = 5\rho_7^v$. Once these relationships are established the specific values of the loads are only shown in the caption of each table and they are not elaborated upon in the accompanying narrative.

As observed in Figure 6, the difference between the knapsack and Pascal approximations and the middle of the confidence interval of the Monte-Carlo summation is very small for the entire range of values of the offered voice load. Similarly, in Figure 7, the average probability of data queueing is illustrated as a function of the offered data load ρ_l^d . The data loads of all four links are assumed equal and the offered voice loads $\rho_p^v r_p$ are all equal to 2.0. Again excellent agreement is observed between the three approximations. The exact values for the probabilities of voice blocking and of data queueing are extremely time-consuming to compute for the multi-rate network in question, so they were not generated. However, since the Monte-Carlo summation method provides a confidence interval for the value of the performance measure of interest, we can judge the accuracy of the approximations even without having their exact values.

In Table 11, the average voice blocking probability is shown for the multi-rate network of Figure 2, different link capacity allocations, and different values of the offered voice loads. The four link capacity allocations considered are $\underline{c} = (90, 100, 110, 120)$, $\underline{c} = (18, 20, 22, 24)$, $\underline{c} = (9, 10, 11, 12)$, and $\underline{c} = (5, 5, 6, 6)$ and they are termed allocations 1, 2, 3, and 4, respectively. Allocation 1 is the example studied in [6]. The others are obtained by reducing the link capacities proportionally (or nearly so). The offered voice loads depicted in Table 11 have been selected so that they correspond to conditions of very light traffic (voice-blocking probability smaller than 0.1%), light to moderate traffic (voice-blocking probability around 1.0%), and moderate to heavy traffic (voice-blocking probability larger than 10.0%), for each of the four capacity allocation scenarios. Note that as the number of channels per link decreases, the offered voice load that corresponds to any particular blocking probability decreases at a much faster rate. For example, let us compare the results for capacity allocations 1 and 3. Although the link capacities are decreased by a factor of 10, the traffic levels corresponding to a specific blocking probability are decreased by a considerably larger factor, especially at low blocking probabilities. Such a behavior is expected; the availability of a large number of channels permits operation at higher throughput levels, because we can take advantage of the law of large numbers.

The knapsack and Pascal approximations remain relatively close to (but generally above) the upper edge of the confidence interval obtained via the Monte Carlo summation method. However, they appear to move farther away from that upper edge (corresponding to larger error) as the number of channels in the capacity allocation schemes decreases. Although we cannot claim that the knapsack approximation is uniformly more accurate than the Pascal approximation, it appears that the former ought to be trusted over a broader range of traffic (offered voice loads) and network (link capacities) parameters than the latter.

Tables 12a and 12b show in greater detail the voice-blocking probabilities of each of the twelve voice circuits of the network of Figure 3. In particular, in Table 12a the detailed results (knapsack, Pascal, and Monte Carlo approximations) are shown for offered voice loads of 8.0, 10.0, and 15.0 and for capacity allocation 1. The loads of the individual voice circuits are such that the average voice load takes the values 8.0, 10.0, or 15.0 once the rates of the different circuits have been accounted for. Again, the knapsack approximation appears to be superior to the Pascal approximation. Table 12b shows the probabilities of blocking for all twelve voice circuits for capacity allocation 2 and for offered voice loads 0.3, 0.7, and 1.7. The knapsack approximation appears to be superior to the Pascal approximation over a broad range of traffic and network parameters. The accuracy of the approximations exhibits a slight degradation, as the number of link channels in the capacity allocations decreases from that of capacity allocation 1 (90, 100, 110, 120) shown in Table 12a to that of capacity allocation 2 (18, 20, 22, 24) shown in Table 12b.

Tables 13 and 14 parallel the results of Tables 11 and 12 for the probability of data queueing. In particular, Table 13 shows the average probability of data queueing for the four capacity allocations and the different voice and data loads. The specific values of the offered voice and data loads have been selected so that they result in probabilities of data queueing of the order of 0.1%, 1.0%, and 10.0%, respectively, corresponding to light, moderate, and moderately heavy traffic. The knapsack and Pascal approximations appear

to be either within or close to the edges of the confidence interval obtained via the Monte Carlo summation method. Again, the knapsack approximation appears to be superior to the Pascal approximation over a broad range of traffic and network parameters. Table 14 complements the results of Table 13 by showing the probabilities of data queueing at each of the four data links of the network of Figure 3 for capacity allocation 2 and for different voice and data loads. The voice loads, for which results are shown, are 0.3, 0.7, and 1.7; the data loads are selected in the manner discussed in Table 13. Observations similar to those made for Table 13 are valid here as well.

11.4 Threshold-Based Admission Control for the Radio Network of Figure 2

The group of Tables 15-20 presents a comparison of the voice blocking probabilities and probabilities of data queueing for the radio network of Figure 2, when admission control is used. In particular, three types of admission control are employed for deriving the results of Table 15: (i) Equal thresholds $X_p = 6$ on the voice traffic of all five circuits of Figure 2b, (ii) Optimal threshold-only policies such that $X_p \leq 6$ on the voice traffic of all five circuits, and (iii) Optimal full admission control policies that consist of thresholds X_p on individual circuit traffic and thresholds Y_s on linear combinations of circuit traffic over selected sets of circuits. The threshold constraints on individual circuit traffic take the form

$$n_p^v \leq X_p, \quad p = 1, 2, 3, 4, 5$$

The linear constraints on selected sets of voice circuits are

 $\begin{array}{l} n_1^v + n_3^v \leq Y_1 \quad (\text{control constraints from node 5}) \\ n_1^v + n_4^v \leq Y_2 \quad (\text{control constraints from node 7}) \\ n_1^v + n_5^v \leq Y_3 \quad (\text{control constraints from node 5}) \\ n_3^v + n_5^v \leq Y_4 \quad (\text{control constraints from node 5}) \\ n_4^v + n_5^v \leq Y_5 \quad (\text{control constraints from node 7}) \end{array}$

These are identical to the admission control schemes considered in [11] and [12]. To the above control constraints one should add the resource constraints of the form

$$\sum_{n \in \mathcal{P}_n} n_p^v \le T_n, \quad n \in \mathcal{N}$$

where T_n is the number of transceivers at node n and \mathcal{P}_n is the set of paths intersecting at node n.

Together with the results of Table 15, one should also go back to Table 1, where no admission control was used. In Table 15, the optimal thresholds obtained for control policies (ii) and (iii) are shown for different values of the voice load; in the former case they are of the form $(X_1, X_2, X_3, X_4, X_5)$, in the latter case they are of the form $(X_1, X_2, X_3, X_4, X_5, Y_1, Y_2, Y_3, Y_4, Y_5)$. In this table, the optimal thresholds are selected on the basis of the exact average probability of voice blocking which is obtained from the product-form solution (as in [11]). The Monte Carlo summation and knapsack approximations are also shown for these thresholds, so that the accuracy of the approximations is checked against the exact value when admission control is used. The comparisons are favorable for both approximations, although the accuracy would be better if the number of transceivers at each node were higher. However, the knapsack approximation does not achieve its minimum value under the optimal thresholds of control policies (ii) and (iii); these optimal thresholds were obtained by using the exact value of the blocking probability. Consequently, we can not repeat the comparisons of [11] for the *c*.fferent control policies based on the knapsack approximation. However, as established with the results of Table 20 below, meaningful comparisons can be made for thresholds obtained from the knapsack approximation.

Table 16 shows the exact value and the knapsack approximation of the voice blocking probabilities for each of the five voice circuits of Figure 2b, voice loads $\rho_p^v = 2.5$ for all circuits, and several different thresholds of the admission control policy (ii). The knapsack approximation shows satisfactory accuracy for all situations examined. Table 17 presents similar results as Table 16, but for different thresholds of the full admission control policy (iii). Again, the knapsack approximation shows satisfactory accuracy for all situations examined, but the accuracy is inferior to that observed in Table 16. In both tables, the accuracy of the knapsack approximation can be improved at the expense of significant computational complexity, if the method of Section 9.2 is used instead of that of Section 9.3, which was used to derive the numerical results presented.

Tables 18 and 19 complement Tables 15, 16, and 17 by presenting results for the probabilities of data queueing in the presence of voice admission control. The voice activity factor is $\beta/(\alpha + \beta) = 0.4$ for all results in this subsection. Table 18 shows the average

probability of data queueing for a data load of 2.5, for all nine data links and for different voice loads and admission control thresholds. The thresholds are the optimal thresholds for control policy (ii) in Table 15. Again, the observed accuracy of the knapsack approximation remains satisfactory, but inferior to that of networks without admission control (see Table 3). Similarly, Table 19 shows detailed results of the probabilities of data queueing at each of the nine data links of the network in question for a data load of 2.5 and different voice loads and admission control thresholds. Observations similar to those made for Table 18 are valid; moreover, the accuracy of the knapsack approximations appears to decrease as the offered voice load increases.

Table 20 compares the exact value and the knapsack approximation for the average voice blocking probability of the radio network of Figure 2 for different voice loads and three admission control scenarios: (a) no control, (b) threshold control on individual voice circuits based on thresholds selected by minimizing the knapsack-evaluated average blocking probability, and (c) threshold admission control based on optimal thresholds obtained by minimizing the exact average voice blocking probability [controls of type (ii) in Table 15]. The percentage improvement (decrease) of the exact blocking probability of the uncontrolled system when knapsack-based thresholds and optimal thresholds are used is also shown in separate columns.

As shown in Table 20, the optimal thresholds derived under policies (b) and (c) differ in one or more voice circuits; however, the performance obtained using the knapsackoptimized thresholds is almost as good as that obtained using the truly-optimal thresholds. For example for an offered voice load of 5.5, the optimal knapsack-optimized thresholds are (1,8,8,8,8) and incur an exact blocking probability of 0.453, whereas the truly optimal thresholds are (1,8,8,8,2) and incur an exact blocking probability of 0.448; the corresponding exact value for the uncontrolled system is 0.472. This corresponds to a 4% improvement (decrease) of the voice blocking probability when the knapsack-optimized thresholds are employed, versus the 5% decrease that can be achieved with the truly optimal thresholds. This improvement resulting from admission control is more drastic for large offered voice loads and for voice blocking probabilities. Up to a voice load of 6.5 the improvement increases as the voice load increases; however, for higher voice loads the improvement decreases with increasing offered voice load.

The results of Table 20 above together with those of Table 23 below (which estab-

lish the significant computational advantage of the knapsack approximation over the exact expressions) provide major motivation for applying the knapsack approximation to problems of control and optimization. Thus the knapsack approximation is useful not only for performance evaluation (as shown in Tables 1 to 14), but for control and optimization as well.

11.5 Voice and Data Revenue Sensitivities for Networks of Figures 2 and 3

The next group of tables (Tables 21a-21c and 22a-22c) pertains to the voice and data revenue sensitivities of the multi-rate network of Figure 3 and to the radio network of Figure 2. These sensitivities were derived in Section 10 and find diverse applications to problems of addition of network resources (e.g., additional transceivers), optimal data routing, joint voice control and data routing etc. In evaluating the revenue sensitivities we need the voice-revenue rates γ_p^v for all voice circuits, and the data revenue rates γ_l^d for all data links of the network. In Table 21, the revenue sensitivities are obtained for the network of Figure 3, an average voice load of 10.0 with circuit loads (10,10,10,10,10,10,2,2,2,2,2,2) and rates (1,1,1,1,1,5,5,5,5,5,5), and offered data loads of 50 for each of the data links in the network. The capacity allocation is $\underline{c} = (90, 100, 110, 120)$, the vector of voice revenue rates is arbitrarily chosen to be (1.0,1.2,1.4,1.6,1.8,2.0,3.0,3.6,4.2,4.8,5.4,6.0), and the vector of data revenue rates is (1.0,1.2,1.4,1.6). The voice activity factor is $\beta/(\alpha + \beta) = 0.4$ for all results in this subsection.

Table 21a shows the Monte-Carlo summation, knapsack, and Pascal approximations to the average voice revenue sensitivity with respect to the loads of all twelve voice circuits. The two approximations are either within or relatively close to the edges of the confidence interval obtained via the Monte Carlo summation method. Table 21b shows the average data revenue sensitivity with respect to the loads of each of the voice circuits. Again, the approximations and the confidence interval are in relatively satisfactory agreement with each other. Table 21c shows the average data revenue sensitivity with respect to the loads of the four data links of the network in question. Similar observations for the approximations like those for Table 21b are valid. It is significant that in all these tables the sign of the exact values of the sensitivities and of the approximate sensitivities is always the same. This guarantees that the the use of the approximate sensitivities (instead of the computationally inefficient exact expressions) in optimization routines will still move the optimization algorithm (e.g., the gradient descent algorithm) to the right direction for optimizing the objective function (performance measure) of interest.

In Table 22, the voice and data revenue sensitivities for the radio network of Figure 2 are presented for a vector of voice revenue rates (1.0, 1.2, 1.4, 1.6, 1.8) and for a vector of data revenue rates (1.0,1.2,1.4,1.6,1.8,2.0,2.2,2.4,2.6). In Table 22a, the exact value and the knapsack approximation to the average voice revenue sensitivity with respect to the loads of each of the five voice circuits are shown; the voice loads of all circuits are equal to 5.5. There is satisfactory agreement between the knapsack approximation and the exact value; the relative error remains below 10%, for most circuits. In Table 22b, the average data revenue sensitivity with respect to the loads of all five voice circuits of the same network are shown for a load of 5.5 for all voice circuits and for two different data load scenarios: one for which the load of all nine data links is equal to 4.0 and another for which it is 2.0. The accuracy of the knapsack approximation here is inferior to that observed in Table 22a, or in Table 21b. This is because the radio network has fewer resources and the knapsack approximation's accuracy decreases as the number of channels in each link decreases. In Table 22c, the average data revenue sensitivity with respect to the loads of each of the nine data links of the network in question is shown for a load of 5.5 on all five voice circuits and the two data load scenarios of Table 22b. The knapsack approximation here appears to be more accurate than that of Table 22b. Similar comments about the common sign of the approximate sensitivities and the exact expressions as for Tables 21a-21b are valid here.

Overall, the accuracy of the knapsack approximation for the voice and data revenue sensitivities appears reasonable and suggests that it can be used instead of the exact values (whose evaluation is computationally prohibitive) in problems of data routing, addition of resources, and so on. A detailed study of these issues will be addressed in the future.

11.6 Required Computational Effort for Various Approximations

Table 23 provides a comparison of the computational effort required for generating the various approximations and exact values of the probabilities of voice blocking and data queueing for the radio network of Figure 2 and the multi-rate network of Figure 3. In the framework of this comparison, the radio network of Figure 2 is typical of a small-size network and that of Figure 3 is typical of a medium- to large-size network. SUN SPARC Station II workstations were employed for all necessary computations. The computer code was written in C and was not optimized.

From the table we see that the computational advantage of the knapsack and Pascal

approximations over the exact expression is tenfold and over the Monte Carlo summation is thirtyfold for the small-size network of Figure 2. For the mid-size network of Figure 3, the advantage in speed of the knapsack and Pascal approximations over the Monte Carlo summation becomes six-hundred-fold, whereas the time necessary for the evaluation of the exact value is prohibitive for most computers. With the latter statement we mean that the required computation time is estimated to be of the order of 6 to 7 days for that paradigm and thus useless for optimization (control or resource allocation) purposes.

The computation time for the probabilities of voice blocking and data queueing appears to be identical for both network paradigms, despite the fact that the probability of data queueing is inherently more complicated (additional conditioning and averaging are involved), for the following reason. We store the pre-computed (recursively) values of the conditional M/D/c data queueing probability (conditioned on the voice state) and then use them in the knapsack approximation (refer to Section 5 for the appropriate equations indicating the conditioning and final averaging). Of course, the memory (storage) requirements for the evaluation of the probability of data queueing are considerably larger than those of the probability of voice blocking.

Table 23 establishes the unquestionable computational advantage of the knapsack and Pascal approximations over the exact expressions for the performance measures of interest and for networks of any size. These two approximations appear to require a computational effort that remains relatively insensitive to the network size. The Monte Carlo summation method is computationally efficient only for small- and mid-size networks. These facts, together with the very satisfactory accuracy of the approximations over the entire range of traffic and network parameters of interest, establish these approximations as excellent candidates for the time-efficient and accurate performance evaluation of networks of arbitrary size and resources (link capacities), as well as for the derivation of near-optimal control schemes that optimize the above performance measures.

12. CONCLUSIONS

In this section we recapitulate the most important features of the approximations developed in this report and draw conclusions about their applicability to problems of network control and optimization, sensitivity analysis, and performance evaluation.

12.1 Accuracy and Computational Advantage of Approximations

In this report, we developed reduced-load approximations (based on the stochastic knapsack and the Pascal distribution) and Monte-Carlo summation techniques, which enable the accurate and computationally efficient evaluation of the probability of blocking of voice calls, the probability of data link queueing, and the average data link delay in integrated multi-hop radio networks. Accurate and time-efficient approximations are necessary, because the computational effort for the evaluation of the exact expressions for these performance measures is prohibitive for networks of even moderate size. By contrast, the approximations of this report (refer to Table 23) are very time-efficient.

12.2 Application of New Approximation Techniques to a Broad Range of Network Architectures and Traffic Types

Although the knapsack, Pascal, and Monte-Carlo Summation methods had already been applied to multi-rate voice-only circuit-switched networks, they had never been applied before either to the performance evaluation of multi-hop radio networks or to networks with integrated voice and data traffic. And, even for multi-rate voice networks, they were thought to be accurate only in the limiting regime characterized by large link capacities and large offered voice loads while their ratio maintained a fixed value.

In this report, besides extending the applicability of the aforementioned approximations to **radio networks** (with transceivers at the nodes being the network resource rather than the link capacities) and to **integrated voice and data traffic**, we showed that these approximations exhibit excellent to very satisfactory accuracy, for the **entire range of network parameters** (e.g., size, link capacities, number of transceivers at nodes) and traffic parameters (e.g., voice and data loads, voice activity and silence periods) of interest. Moreover, the accuracy of the approximations is very satisfactory not only for the **average** performance measures (i.e., averaged over all circuits and links of the network) but also for the performance measures characterizing **individual** circuits or links. These approximation methods are applicable to single-rate networks characterized by a single bandwidth assignment for all voice circuits, as well as to multi-rate networks, in which calls with different bandwidth requirements are accepted; in both cases, the data may be transmitted at a lower rate than that of the voice traffic. Moreover, our methods can handle voice models with periods of activity and silence, in which data is transmitted during the silent periods of ongoing voice calls. Finally, our methods are applicable to variable-rate traffic voice or video; in these situations the rate of the source can assume a number of different values (from a fixed set) as time varies. Our preliminary results actually show that our approach (based on the knapsack approximation) is applicable to sophisticated Markov-Modulated Poisson Process (MMPP) models for the variable-rate voice or video traffic with only a small increase of computational complexity. In this context, besides variable-rate voice, video telephony as well as full-motion video sources can be modeled and networks with such traffic can be analyzed.

12.3 Application to Control and Optimization Problems

Besides employing these approximation methods for the accurate and timeefficient performance evaluation of integrated multi-hop radio networks (and of more general multi-rate networks) we can use them for control and optimization purposes, because of their computational efficiency. The recommended general methodology here is to use an accurate and computationally time-efficient approximation (e.g., knapsack) to evaluate the specific performance measure(s) (single measure or multiple measures in a weighted sum) of interest and carry out the minimization (optimization in general) with respect to the most promising classes of controls. The controls derived will be near-optimal in the sense that their performance will be very close to those of the really optimal ones that are derived from the optimization of the exact performance measures.

In particular, we have shown (refer to Table 20) that threshold policies for voiceadmission control that minimize the knapsack-evaluated probability of voice blocking improve substantially the performance of the controlled network over that of the network without admission control and actually come very close to the performance of the optimal thresholds (which minimize the actual voice blocking probability). Without these computationally efficient accurate approximations to the performance measures of interest, the derivation of near-optimal control schemes is not feasible and one has to resort to ad-hoc designs.

Moreover, the sensitivities of suitably defined voice and data revenue measures with respect to link capacities (or number of node transceivers), voice loads, and data loads were evaluated via the knapsack approximation and shown to be very close to the actual values (refer to Tables 21 and 22). Again, these approximate sensitivities are much more computationally efficient than the cumbersome (and usually prohibitive) exact expressions. Actually, as our results establish, there will be almost negligible loss in revenue when these approximate sensitivities are used in place of the exact ones. Consequently, important practical problems of allocating additional network resources in response to increasing voice and/or data network traffic demand can be easily handled with our approach, as well as problems of data routing in which the derivatives of the data delay (or the probability of queueing) are used by standard optimal routing algorithms.

12.4 Application to High-Speed Networks

The approximations to the probabilities of voice blocking, data queueing, and the average data delay developed in this report can be also applied to important problems in the area of high-speed networks, with proper modification. These include:

- (i) algorithms for the set up of virtual paths for video, voice, and high rate data sources in asynchronous transfer mode (ATM) networks.
- (ii) schemes for admission control of different classes of traffic in ATM networks, and
- (iii) specific formulations of the problem of multicasting hierarchically encoded data to destinations that can receive subsets of the transmitted signal according to their bandwidth and access constraints.

In all such high-speed network applications, the approximations of this report are expected (after suitable modification) to provide both accurate and time-efficient performance evaluations and to be used for deriving near-optimal control schemes and resource allocations.

ACKNOWLEDGEMENT

The authors would like to express their sincere thanks to Dr. Jeffrey Wieselthier of the Naval Research Laboratory for the insightful advice and the stimulating comments that he has offered throughout the preparation of this report.

REFERENCES

- L. Kleinrock. Communication Nets. Stochastic Message Flow and Delay. New York: McGraw-Hill, 1964.
- [2] D. Bertsekas and R. Gallager. Data Networks. Prentice Hall International, 1987.
- [3] J. Y. Hui. Switching and Traffic Theory for Integrated Broadband Networks. Boston: Kluwer Academic Publishers, 1990.
- [4] F. Kelly. "Blocking Probability in Large Circuit-Switched Networks." Advances in Applied Probability, Vol. 18, pp. 473-505, 1986.
- [5] F. Kelly. "Adaptive Routing in Circuit-Switched Networks." Advances in Applied Probability, Vol. 20, pp. 112-144, 1988.
- [6] S. P. Chung and K. W. Ross. "Reduced Load Approximations for Multirate Loss Networks." Preprint, 1991.
- [7] K. W. Ross and J. Wang. "Monte Carlo Summation Applied to Product-Form Loss Networks." Preprint, 1991.
- [8] K. W. Ross and D. D. Yao. "Monotonicity Propertied of the Stochastic Knapsack." IEEE Transactions on Information Theory, Vol. IT-36, pp. 1173-1179, 1990.
- [9] K. W. Ross and D. Tsang. "Teletraffic Engineering for Product-Form Circuit-Switched Networks." Advances in Applied Probability, Vol. 22, pp. 657-675, 1990.
- [10] D. Mitra. "Asymptotic Analysis and Computational Methods for a Class of Simple, Circuit-Switched Networks with Blocking." Advances in Applied Probability, Vol. 19, pp. 219-239, 1987.
- [11] C. M. Barnhart, J. E. Wieselthier, and A. Ephremides. "An Approach to Voice Admission Control in Multi-Hop Wireless Networks." Proceedings of IEEE INFOCOM'93, March 1993, pp. 246-255.
- [12] C. M. Barnhart, J. E. Wieselthier, and A. Ephremides. "Admission Control in Integrated Voice/Data Multi-Hop Radio Networks: Analysis." NRL Memorandum Report, NRL/MR/5521-93-7196.
- [13] L. P. Seelen and H. C. Tijms. "Approximations to the Waiting Time Percentiles in the M/G/c Queue." Teletraffic Issues, Vol. ITC-11, pp. 53-57, 1985.

- [14] J. E. Wieselthier and A. Ephremides. "A Movable-Boundary Channel-Access Scheme for Integrated Voice/Data Networks." Proceedings of the IEEE INFOCOM '91, pp. 721-731, April 1991.
- [15] R. Y. Rubinstein. Simulation and the Monte Carlo Method. New York: John Wiley and Sons, 1981.
- [16] P. Bratley, B. L. Fox, and L. E. Schrage. A Guide to Simulation. New York: Springer-Verlag, 1987.
- [17] D. Y. Burman, J. P. Lehoczky, and Y. Lim. "Insensitivity of Blocking Probabilities in a Circuit-Switching Network." J. Appl. Prob. Vol. 21, pp. 850-859, 1984.

APPENDIX A Proof of Part (a) of Proposition 3.1 (System Steady-State Distribution)

The starting point is the chain rule for the steady-state probability of the voice state $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s})$, namely

$$P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}, n_{l}^{d}) = P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}) \cdot P(n_{l}^{d} | \underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}).$$
(A-1)

Since voice has preemptive priority over data, the analysis of the voice component can be isolated from that of the data component. Consider the voice state vector $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s})$; denote

$$\underline{N_{\mathcal{P}p}^{v+}} = (n_1^v, \dots, n_{p-1}^v, n_p^v + 1, n_{p+1}^v, \dots, n_{|\mathcal{P}|}^v)$$

$$\underline{N_{\mathcal{P}p}^{v-}} = (n_1^v, \dots, n_{p-1}^v, n_p^v - 1, n_{p+1}^v, \dots, n_{|\mathcal{P}|}^v)$$

$$\underline{N_{\mathcal{P}p}^{s+}} = (n_1^s, \dots, n_{p-1}^s, n_p^s + 1, n_{p+1}^s, \dots, n_{|\mathcal{P}|}^s)$$

$$\underline{N_{\mathcal{P}p}^{s-}} = (n_1^s, \dots, n_{p-1}^s, n_p^s - 1, n_{p+1}^s, \dots, n_{|\mathcal{P}|}^s).$$

The global balance equation for $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s})$ is

$$\begin{split} P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}) \cdot \sum_{p \in \mathcal{P}} \left[n_{p}^{s} \mu_{p}^{v} + (n_{p}^{v} - n_{p}^{s}) \mu_{p}^{v} + (n_{p}^{v} - n_{p}^{s}) \alpha + n_{p}^{s} \beta + F_{p}^{v} \frac{\alpha}{\alpha + \beta} + F_{p}^{v} \frac{\beta}{\alpha - \beta} \right] = \\ &= \sum_{p \in \mathcal{P}} \left[P(\underline{N}_{\mathcal{P}p}^{v-}, \underline{N}_{\mathcal{P}p}^{s-}) \cdot F_{p}^{v} \frac{\alpha}{\alpha + \beta} + P(\underline{N}_{\mathcal{P}p}^{v-}, \underline{N}_{\mathcal{P}}^{s}) \cdot F_{p}^{v} \frac{\beta}{\alpha + \beta} \right. \\ &+ P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}p}^{s+}) \cdot (n_{p}^{s} + 1) \beta + P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}p}^{s-}) \cdot (n_{p}^{v} - n_{p}^{s} + 1) \alpha \\ &+ P(\underline{N}_{\mathcal{P}p}^{v+}, \underline{N}_{\mathcal{P}p}^{s+}) \cdot (n_{p}^{s} + 1) \mu_{p}^{v} + P(\underline{N}_{\mathcal{P}p}^{v+}, \underline{N}_{\mathcal{P}p}^{s}) \cdot (n_{p}^{v} - n_{p}^{s} + 1) \mu_{p}^{v} \right] \tag{A-2}$$

for all $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}), (\underline{N}_{\mathcal{P}p}^{v\pm}, \underline{N}_{\mathcal{P}p}^{s\pm}) \in \Omega_{s}.$

The local balance equations for $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s})$ are

$$P(\underline{N_{\mathcal{P}}^{v}}, \underline{N_{\mathcal{P}}^{s}}) \cdot n_{p}^{s} \mu_{p}^{v} = P(\underline{N_{\mathcal{P}p}^{v-}}, \underline{N_{\mathcal{P}p}^{s-}}) \cdot F_{p}^{v} \frac{\alpha}{\alpha + \beta}$$
(A-3)

$$P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}) \cdot (n_{p}^{v} - n_{p}^{s})\mu_{p}^{v} = P(\underline{N}_{\mathcal{P}_{p}}^{v-}, \underline{N}_{\mathcal{P}}^{s}) \cdot F_{p}^{v} \frac{\beta}{\alpha + \beta}$$
(A-4)

$$P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}) \cdot (n_{p}^{v} - n_{p}^{s})\alpha = P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}p}^{s+}) \cdot (n_{p}^{s} + 1)\beta \qquad (A-5)$$

$$P(\underline{N}^{\nu}_{\mathcal{P}}, \underline{N}^{s}_{\mathcal{P}}) \cdot n^{s}_{p}\beta = P(\underline{N}^{\nu}_{\mathcal{P}}, \underline{N}^{s-}_{\mathcal{P}p})(n^{\nu}_{p} - n^{s}_{p} + 1)\alpha \qquad (A-6)$$

$$P(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}) \cdot F_{p}^{v} \frac{\alpha}{\alpha + \beta} = P(\underline{N}_{\mathcal{P}p}^{v+}, \underline{N}_{\mathcal{P}p}^{s+}) \cdot (n_{p}^{s} + 1)\mu_{p}^{v} \qquad (A - 7)$$

$$P(\underline{N_{\mathcal{P}}^{v}},\underline{N_{\mathcal{P}}^{s}})\cdot F_{p}^{v}\frac{\beta}{\alpha+\beta} = P(\underline{N_{\mathcal{P}p}^{v+}},\underline{N_{\mathcal{P}}^{s}})\cdot (n_{p}^{v}+1-n_{p}^{s})\mu_{p}^{v} \qquad (A-8)$$

for all $(\underline{N}_{\mathcal{P}}^{v}, \underline{N}_{\mathcal{P}}^{s}), (\underline{N}_{\mathcal{P}p}^{v\pm}, \underline{N}_{\mathcal{P}p}^{s\pm}) \in \Omega_{s}.$

It is straightforward to show that the $P(\underline{N_{\mathcal{P}}^{v}}, \underline{N_{\mathcal{P}}^{s}})$ provided in Part (a) of Proposition 1 is a solution to the above local balance equations. Any solution to the local balance equation must also be a solution to the global balance equation and, as $P(\underline{N_{\mathcal{P}}^{v}}, \underline{N_{\mathcal{P}}^{s}})$ in Part (a) satisfies the balance equation, it is the steady state probability.

APPENDIX B

M/M/c and M/D/c Data Models

This appendix reviews all important formulas of the M/M/c and M/D/c data models and sketches the relaxation method proposed by Tijms [13] for the evaluation of the steady state probabilities and other quantities of interest for the M/D/c model.

B.1 Useful Expressions for M/M/c Data Model

The steady-state probabilities satisfy the recursion

$$\lambda p_{j-1} = \min(j,c) \mu p_j, \quad j = 1, 2, \dots$$

and are given by the expressions

$$p_j = \begin{cases} \frac{(c\rho)^j}{j!} p_0, & j = 0, 1, \dots, c-1, \\ \frac{(c\rho)^j}{c! c^{j-c}} p_0, & j \ge c \end{cases}$$

where

$$p_0 = \left\{ \sum_{k=0}^{c-1} \frac{(c\rho)^k}{k!} + \frac{(c\rho)^c}{c!(1-\rho)} \right\}^{-1}.$$

The probability of finding the system busy (blocking probability) $\Pi_w = P\{j \ge c\} = \sum_{j=c}^{\infty} p_j$ is given by

$$\Pi_{w} = \frac{(c\rho)^{c}}{c!(1-\rho)} \left\{ \sum_{k=0}^{c-1} \frac{(c\rho)^{k}}{k!} + \frac{(c\rho)^{c}}{c!(1-\rho)} \right\}^{-1}$$

The average number of customers in the queue $E\{N_q\} = \sum_{j=c}^{\infty} (j-c)p_j$ is given by

$$E(N_q) = \frac{(c\rho)^c \rho}{c!(1-\rho)^2} \left\{ \sum_{k=0}^{c-1} \frac{(c\rho)^k}{k!} + \frac{(c\rho)^c}{c!(1-\rho)} \right\}^{-1}$$

Finally, the probability distribution of queueing delay (i.e., waiting time, not including service time) W_q is given by

$$P\{W_q > x\} = \sum_{j=c}^{\infty} p_j \sum_{k=0}^{j-c} e^{-c\mu x} \frac{(c\mu x)^k}{k!} = \prod_w e^{-c\mu(1-\rho)x}, \quad x \ge 0.$$

and the average queueing delay is given by

$$E(W_q) = \frac{E(N_q)}{\lambda} = \frac{(c\rho)^c}{c!c\mu(1-\rho)^2} \left\{ \sum_{k=0}^{c-1} \frac{(c\rho)^k}{k!} + \frac{(c\rho)^c}{c!(1-\rho)} \right\}.$$

B.2 Useful Expressions for M/D/c Data Model

The steady-state probabilities satisfy the following equilibrium equations

$$p_0(t+D) = \sum_{k=0}^{c} p_k(t) e^{-\lambda D}$$

and

$$p_{j}(t+D) = \sum_{k=0}^{c} p_{k}(t)e^{-\lambda D} \frac{(\lambda D)^{j}}{j!} + \sum_{k=c+1}^{c+j} p_{k}(t)e^{-\lambda D} \frac{(\lambda D)^{j-k+c}}{(j-k+c)!}, \quad j \ge 1$$

where D is the service time.

From those we can derive the recursion

$$p_{j} = e^{-\lambda D} \frac{(\lambda D)^{j}}{j!} \sum_{k=0}^{c} p_{k} + \sum_{k=c+1}^{c+j} p_{k} e^{-\lambda D} \frac{(\lambda D)^{j-k+c}}{(j-k+c)!}, \quad j = 0, 1, \dots$$

where

$$\sum_{j=0}^{\infty} p_j = 1$$

The average number of customers in the queue can be also derived as

$$E(N_q) = \frac{(c\rho)^2 - c(c-1) + \sum_{j=0}^{c-1} \{c(c-1) - j(j-1)\} p_j}{2c(1-\rho)}$$

and its second moment as

$$E(N_q(N_q-1)) = \frac{(c\rho)^3 - c(c-1)(c-2) + \sum_{j=0}^{c-1} \{c(c-1)(c-2) - j(j-1)(j-2)\}p_j}{3c(1-\rho)} - \frac{(c-1-c\rho^2)}{1-\rho} E(N_q).$$

B.3 Relaxation Algorithm for Evaluating Steady State Probabilities of M/D/c Data Model

The above recursion for the p_j 's can not be used directly to find the p_j 's of the M/D/c system because it is equivalent to an infinite system of linear equations. Thus we first truncate it to a finite system by using a sufficiently large integer L chosen so that

$$\sum_{j=L}^{\infty} p_j \le \sum_{j=L}^{\infty} p_j^{\exp} \le 10^{-9}$$

where p_j^{exp} for j = 0, 1, ... denote the steady state probabilities of the M/M/c system, which are available in closed form (see Section B.1 above); the inequality reflects the intuitive fact that the M/D/c queue involves less variability than the M/M/c queue.

It turns out (see [13]) that for light traffic it is preferable to use the truncation to L terms described above. However, for non-light traffic, unacceptably large values of L may be needed. In this case it is advantageous to use the theoretical fact [13] that

$$\frac{p_{j-1}}{p_j} \approx \tau \quad \text{for all sufficiently large j}$$

where

$$\tau = 1 + \delta/\lambda$$

and δ is determined as the unique positive solution to the equation

$$\lambda(e^{\delta D/c} - 1) = \delta$$

This asymptotic result is exploited in the following manner: for all j such that $0 \le j \le N$ we use p_j and try to determine their value from the system of linear equations; for all j > N we use

$$p_j \approx \tau^{N-j} p_N \quad \text{for} \quad j > N$$

where N > c. It turns out (see [13]) that the necessary value of N is smaller than the necessary value of L (of the previous paragraph). The procedure for obtaining the p_j 's is sketched next (following [13]).

After truncating the linear system of equations involving the p_j 's to an $(N+1)\times(N+1)$ linear system and replacing the probabilities p_j by $p_N \tau^{N-j}$, for $j \ge N$, we obtain the following system of linear equations:

$$p_j = \sum_{\substack{k=0\\k\neq j}}^N a_{jk} p_k, \qquad j = 0, 1, \dots, N$$

with

$$\sum_{j=0}^{N} p_j + \frac{\tau}{\tau - 1} p_N = 1$$
 (16)

where

$$a_{jk} = \begin{cases} \frac{a(\min(j,j-k+c))}{1-a(\min(j,c))}; & 0 \le j \le N-1, \ 0 \le k \le \min(c+j,N-1) \\ \frac{\sum_{k=N}^{c+j} \tau^{N-k} a(j-k+c)}{1-a(\min(j,c))}; & N-c \le j \le i, -1, \ k = N \\ \frac{a(\min(j,j-k+c))}{1-\sum_{k=N}^{c+j} \tau^{N-k} a(j-k+c)}; & j = N, \ 0 \le k \le N-1 \\ 0; & \text{otherwise} \end{cases}$$

Here,

$$a(\ell) = e^{-\lambda D} (\lambda D)^{\ell} / \ell!, \text{ for } \ell \ge 0.$$

The above system of linear equations is solved via the modified successive overrelaxation method of Tijms [13].

According to the standard successive overrelaxation method the operator B_{ω} associated with a relaxation factor ω transforms each vector $x = (x_0, x_1, \ldots, x_N)$ into the vector $B_{\omega}x$, whose components $(B_{\omega}x)_i$ are defined recursively by

$$(B_{\omega}x)_i = (1-\omega)x_i + \omega \left[\sum_{j=0}^{i-1} a_{ij}(B_{\omega}x)_j + \sum_{j=i+1}^N a_{ij}x_j\right] \quad i = 0, 1, \dots, N.$$

Assuming that the integer N is sufficiently large—so that the reduced system of linear equations has a solution—then this solution is an eigenvector of B_{ω} with associated eigenvalue 1. Letting $\lambda_1(\omega)$ be the eigenvalue having the largest absolute value among the eigenvalues of B_{ω} unequal to 1, the standard successive overrelaxation method with a fixed relaxation factor ω converges only if $|\lambda_1(\omega)| < 1$. Moreover, the standard overrelaxation method has the best convergence rate for that value of ω for which $|\lambda_1(\omega)| < 1$ is smallest. It should be noted that the optimal value of ω may be rather sensitive to the parameters of the specific problem considered and, in some cases, will be close to 1.

The problems of the standard overrelaxation method are avoided with Tijms' algorithm [13]. In this algorithm ω is always kept between 1 and 2. In case $\lambda_1(\omega)$ is real, the algorithm estimates $\lambda_1(\omega)$ after some iteration of the overrelaxation method (see the parameter r^h in the algorithm below). This estimate provides a method to formulate a successive overrelaxation algorithm, in which the relaxation factor is dynamically adjusted in order to search for that value of ω , for which $|\lambda_1(\omega)| < 1$ is smallest. The steps of Tijms' algorithm for the calculation of the state probabilities in the M/D/c queue are provided below.

Special-Purpose Overrelaxation Method for the M/D/c Queue

Step 0. Choose N > c and $x^0 \ge 0$ with

$$\sum_{i=0}^{N} x_i^0 + \tau (\tau - 1)^{-1} x_N^0 = 1$$

Also, h := 0 and $\omega := 1.20$. **Step 1.** $\omega^{old} := 0$, $\lambda(\omega^{old}) := 1$, $f^h := r^h := \infty$. **Step 2.** h := h + 1. Compute the vectors

$$\tilde{x}^h = B_\omega x^{h-1}$$

$$h_{i}^{h} = \left[\sum_{i=0}^{N} |\tilde{x}_{i}^{h}| + \frac{\tau}{\tau - 1} |\tilde{x}_{N}^{h}|\right]^{-1} \tilde{x}^{h}$$

and the scalar

$$f^h := \frac{1}{N} \sum_{i=0}^{N} \left| \frac{\tilde{x}_i^h - \tilde{x}_i^{h-1}}{\tilde{x}_i^h} \right|$$

If $f^h < \epsilon_N$, with ϵ_N a prespecified accuracy number, then go to Step 4. Otherwise

$$r^h := \frac{f^h}{f^{h-1}}$$

If $r^h \ge 1$ or $h \ge 10$, then ω is likely too large and decrease ω , as $\omega := 1 + \frac{1}{2}(\omega - 1)$, put $x^0 := x^h$ and h := 0, and go to Step 1. If $r^h < 1$ and r^h has sufficiently converged, according to $|(r^h - r^{h-1})/r^h| < 0.025$, then go to step 3; otherwise, return to Step 2. **Step 3.** $\lambda(\omega) := r^h$. Test for one of the following four possibilities: (a) $\omega > \omega^{old}$ and $\lambda(\omega) > \lambda(\omega^{old})$; (b) $\omega > \omega^{old}$ and $\lambda(\omega) \le \lambda(\omega^{old})$; (c) $\omega < \omega^{old}$ and $\lambda(\omega) > \lambda(\omega^{old})$; (d) $\omega < \omega^{old}$ and $\lambda(\omega) \le \lambda(\omega^{old})$. For the cases (a) and (d),

$$\omega^{old} := \omega, \quad \lambda(\omega^{old}) := \lambda(\omega), \quad \omega := 1 + 0.85(\omega - 1)$$

whereas, for the cases (b) and (c),

$$\omega^{old} := -\omega, \quad \lambda(\omega^{old}) := \lambda(\omega), \quad \omega := 1 + 1.25(\omega - 1)$$

Next, $x^0 := x^h$, h := 0, $f^h := r^h := \infty$, and then go to Step 2. Step 4. If

$$|\sum_{i=0}^{N} \tilde{x}_{i}^{h} + \frac{\tau}{\tau - 1} \tilde{x}_{h}^{N} - 1| < 10^{-5}$$

and

$$\left|\sum_{i=0}^{c-1} i x_i^h + c \left(1 - \sum_{i=0}^{c-1} x_i^h\right) - c \rho\right| < 10^{-5}$$

then the algorithm is stopped and the state probabilities p_i are obtained from

$$p_i = x_i^h$$
 for $0 \le i \le N$ and $p_i = \tau^{N-1} x_N^h$ for $i > N$.

The above stopping criteria use the fact that the probabilities sum to 1 and that the average number of busy servers equals $c\rho$. Otherwise,

$$x_i^0 := x_i^h \text{ for } 0 \le i \le N \text{ and } x_i^0 := \frac{1}{\tau} x_{i-1} \text{ for } N \le i \le N+10$$

N := N + 10, h := 0, and then go to Step 1.

B.4 Recursive Algorithm for Evaluating the Probability Distribution of Queueing Delay under the M/D/c Data Model

To compute the probability distribution of the queueing delay W_q , that is $P\{W_q \le x\}$, we first find an integer m and the remainder u $(0 \le u < D)$ such that

$$x=mD+u.$$

Then we obtain the desired quantity from

$$P\{W_q \le x\} = b_{mc+c-1}(u)$$

where the $b_l(u)$ satisfy the recursion

$$\sum_{i=0}^{j} p_{i} = \sum_{k=0}^{j} b_{j-k}(u) e^{-\lambda u} \frac{(\lambda u)^{k}}{k!}, \quad \text{for} \quad j = 0, 1, \dots$$

with initial value

$$b_0(u)=e^{\lambda u}p_0.$$

Actually, for computational purposes the following recursion is preferred

$$b_j(u) = e^{\lambda u} \sum_{i=0}^j p_i - \sum_{k=0}^{j-1} b_k(u) \frac{(-\lambda u)^{j-k}}{(j-k)!}, \quad \text{for} \quad j = 1, 2, \dots$$

or equivalently

$$b_j(u) = e^{\lambda u} p_j - \sum_{k=0}^{j-1} b_k(u) \frac{(-\lambda u)^{j-1-k}}{(j-1-k)!} \left(\frac{\lambda u}{j-k} - 1\right), \quad \text{for} \quad j = 1, 2, \dots$$

The above recursion should be used in combination with the approximation

$$P\{W_q > x\} \approx \alpha e^{-\delta x} \text{ for } x \geq D/\sqrt{c}.$$

In the above approximation δ is again determined as the solution to the equation

$$\lambda(e^{\delta D/c} - 1) = \delta$$

and α is given by

$$\alpha = \frac{\delta\eta}{\lambda(\tau-1)^2\tau^{c-1}}$$

where

•

.

$$\tau = 1 + \frac{\delta}{\lambda}$$

and

$$\eta = \lim_{j \to \infty} \tau^j p_j = \frac{\sum_{i=0}^{c-1} p_i(\tau^{i-1} - \tau^{c-1})}{c/\tau - \lambda D}$$

.

APPENDIX C

Proof of Proposition 5.1

(Knapsack Approximation)

From the proof of Proposition 3.1 we know that $P(\underline{N}^{v}, \underline{N}^{s})$ satisfies several local balance equations. But actually the first two of them are sufficient for solving $P(\underline{N}^{v}, \underline{N}^{s})$. For the same reason, the following two local balance equations are sufficient for obtaining P(k,m)

$$\frac{\alpha}{\alpha+\beta}\rho_p^{\nu}P(\underline{N_p^{\nu-}},\underline{N_p^{s-}}) = n_p^{s}P(\underline{N^{\nu}},\underline{N^{s}}), \quad n_p^{\nu}, n_p^{s} \ge 1$$
 (C-1)

$$\frac{\beta}{\alpha+\beta}\rho_p^{\nu}P(\underline{N}_p^{\nu-},\underline{N}_p^{s}) = (n_p^{\nu} - n_p^{s})P(\underline{N}^{\nu},\underline{N}^{s}), \quad n_p^{\nu} \ge 1 \qquad (C-2)$$

where we have dropped the subscript \mathcal{P} in $\underline{N_{\mathcal{P}p}^{v-}}, \underline{N_{\mathcal{P}p}^{s-}}, \underline{N_{\mathcal{P}p}^{s-}}, \underline{n}$ and in the other vectors in order to simplify the notation; as usual, $\rho_p^v = F_p/\mu_p^v$. Summing both sides of equation (C-1) with respect to $(\underline{N^v}, \underline{N^s}) \in \Omega(k, m)$, one can obtain

$$\frac{\alpha}{\alpha+\beta}\rho_p^v \sum_{(\underline{N^v},\underline{N^s})\in\Omega(k,m)\cap\{n_p^v,n_p^s\geq 1\}} P(\underline{N^{v-}},\underline{N^{s-}}) = \sum_{(\underline{N^v},\underline{N^s})\in\Omega(k,m)} n_p^s P(\underline{N^v},\underline{N^s}) \quad (C-3)$$

where

$$\begin{split} \Omega(k,m) \cap \{n_p^v, n_p^s \ge 1\} &= \\ &= \left\{ (\underline{N}^v, \underline{N}^s | 1 \le n_p^s \le n_p^v, \ r_p \le r_p n_p^v \le c_l, \ 0 \le n_q^s \le n_q^v, \ 0 \le r_q n_q^v \le c_l, \ q, p \in \mathcal{P}_l, q \neq p; \right. \\ &\left. \sum_{q \in \mathcal{P}_l, q \neq p} r_q n_q^s + r_p (n_p^s - 1) = m - r_p, \sum_{q \in \mathcal{P}_l, q \neq p} r_q n_q^v + r_p (n_p^v - 1) = k - r_p \right\}. \end{split}$$

For the RHS of (C-3) we used the fact that, if $n_p^s = 0$ then RHS = 0 and if $n_p^v = 0$ then $n_p^s = 0$ and thus RHS = 0. If we define

$$\hat{n}_{q}^{s} = \begin{cases} n_{p}^{s} - 1, & p = q \\ n_{q}^{s}, & p \neq q \end{cases}$$

$$\hat{n}_{q}^{v} = \begin{cases} n_{p}^{v} - 1, & p = q \\ n_{q}^{v}, & p \neq q \end{cases}$$

and replace (n_q^v, n_q^s) by $(\hat{n}_q^v, \hat{n}_q^s)$ in the LHS of (C-3), then $\Omega(k, m) \cap \{n_p^v, n_p^s \ge 1\} = \Omega(k - r_p, m - r_p)$ with the new variables and we can rewrite the LHS of (C-3)

$$\frac{\alpha}{\alpha+\beta}\rho_p^v P(k-r_p,m-r_p).$$

Moreover, we can rewrite the RHS of (C-3) as

$$\sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(k,m)}n_{p}^{s}\frac{P(\underline{N}^{v},\underline{N}^{s})}{P(k,m)}P(k,m).$$

Note that

$$P(\underline{N}^{v}, \underline{N}^{s} \Big| \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{s} = m, \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v} = k) = \begin{cases} \frac{P(N^{v}, N^{s})}{P(k, m)}, & (\underline{N}^{v}, \underline{N}^{s}) \in \Omega(k, m) \\ 0, & \text{otherwise} \end{cases}$$

Denote

$$E[n_p^s|k,m] = \sum_{(\underline{N}^v,\underline{N}^s)\in\Omega(k,m)} n_p^s \cdot P(\underline{N}^v,\underline{N}^s \Big| \sum_{q\in\mathcal{P}_l} r_q n_q^s = m, \sum_{q\in\mathcal{P}_l} r_q n_q^v = k).$$

Then the RHS of (C-3) becomes

$$E[n_p^s|k,m] \cdot P(k,m)$$

Equating the modified LHS and RHS of (C-3) we have

$$\frac{\alpha}{\alpha+\beta}\rho_p^v P(k-r_p,m-r_p) = E[n_p^s|k,m] \cdot P(k,m). \qquad (C-4)$$

Following similar arguments, we obtain

$$\frac{\beta}{\alpha+\beta}\rho_p^{\nu}P(k-r_p,m) = E[n_p^{\nu}-n_p^{s}|k,m] \cdot P(k,m) \qquad (C-5)$$

from equation (C-2). Multiplying equations (C-4) and (C-5) by r_p , summing with respect to $p \in \mathcal{P}_l$, and using the facts that

$$E\left\{\sum_{p\in\mathcal{P}_l}r_pn_p^*\middle|k,m\right\}=m$$

and

$$E\left\{\sum_{p\in\mathcal{P}_l}r_p\left(n_p^v-n_p^s\right)\bigg|k,m\right\}=k-m$$

we obtain

$$\frac{\alpha}{\alpha+\beta}\sum_{p\in\mathcal{P}_l}r_p\rho_p^{\nu}P(k-r_p,m-r_p)=m\cdot P(k,m)\,,\ r_p\leq m\leq k\leq c_l$$

and

$$\frac{\beta}{\alpha+\beta}\sum_{p\in\mathcal{P}_l}r_p\rho_p^{\nu}P(k-r_p,m)=(k-m)\cdot P(k,m), \quad 0=m< r_p\leq k\leq c_l.$$

Finally, by defining

$$P'(k,m) = P(k,m)/P(0,0)$$

we derive (5.10a) and Proposition 5.1 follows.

APPENDIX D

Four-Dimensional Knapsack Approximation

(Proofs of Propositions 5.2 and 9.1)

In this appendix we provide the recursion for the computation of the four-dimensional knapsack approximation used for evaluating the probability of queueing of data and the average queueing delay of data (Sections 5.2 and 7.2) and for the probability of voice blocking with admission control (Section 9). We present the proof only for Proposition 5.2; Proposition 9.1 follows as a special case of 5.2.

Here we provide a recursion for $P(k_1, m_1, k_2, m_2)$ defined as

$$P(k_1, m_1, k_2, m_2) = \sum_{(\underline{N}^v, \underline{N}^s) \in \Omega(k_1, m_1, k_2, m_2)} P(\underline{N}^v, \underline{N}^s)$$

where \mathcal{P}_i for i = 1, 2 are two arbitrary sets of paths, with $\mathcal{P}_1 \neq \mathcal{P}_2$.

$$0 \leq k_i = \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^v \leq Z_i, \qquad i = 1, 2$$

$$0 \leq m_i = \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^s \leq k_i, \qquad i = 1, 2.$$

Notice that these inequalities can represent any constraint involving a linear combination of the number of voice calls n_p^v of all classes $p \in \mathcal{P}_i$. The coefficients $r_p^{(i)}$ (i = 1, 2) may be equal to the rates r_p or may be arbitrary non-negative constants. For example, $Z_1 = c_l$ and $r_p^{(1)} = r_p$ whereas $Z_2 = T_s$ and $r_p^{(2)} = r_p^{(s)}$ corresponds to the scenario of Section 9.1. Also the general case of unequal rates $r_p^{(i)}$ of path p under the two constraints i = 1, 2 (i.e., $r_p^{(1)} \neq r_p^{(2)}$) can be handled by our analysis. Finally, $\Omega(k_1, m_1, k_2, m_2)$ is defined by

$$\Omega(k_1, m_1, k_2, m_2) = \left\{ (\underline{N}^v, \underline{N}^s) | 0 \le n_p^s \le n_p^v, p \in \mathcal{P} = \mathcal{P}_1 \cup \mathcal{P}_2; \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^v = k_i, \right.$$
$$\left. \sum_{p \in \mathcal{P}_i} r_p^{(i)} n_p^s = m_i, i = 1, 2 \right\}$$

$$\sum_{k_1=0}^{Z_1} \sum_{m_1=0}^{k_1} \sum_{k_2\neq 0}^{Z_2} \sum_{m_2=0}^{k_2} P(k_1, m_1, k_2, m_2) = 1.$$

The starting point is to consider the first two of the local balance equations for $P(\underline{N}^{v}, \underline{N}^{s})$ given in Appendix A, namely

$$\frac{\alpha}{\alpha+\beta}\rho_p^{v}P(\underline{N_p^{v-}},\underline{N_p^{s-}}) = n_p^{s}P(\underline{N^{v}},\underline{N^{s}}), \quad n_p^{v}, \ n_p^{s} \ge 1)$$

and

$$\frac{\beta}{\alpha+\beta}\rho_p^v P(\underline{N_p^{v-}},\underline{N_p^s}) = (n_p^v - n_p^s)P(\underline{N^v},\underline{N^s}), \quad n_p^v \ge 1$$

where we have dropped the subscript \mathcal{P} in $\underline{N}_{\mathcal{P}p}^{v-}$, $\underline{N}_{\mathcal{P}p}^{s-}$, and in the other vectors in order to simplify the notation; as usual, $\rho_p^v = F_p/\mu_p^v$. After summing both members of these equations with respect to all $(\underline{N}^v, \underline{N}^s)$ in the set $\Omega(k_1, m_1, k_2, m_2)$ defined above we obtain

$$\frac{\alpha}{\alpha+\beta}\rho_p^v \sum_{(\underline{N}^v,\underline{N}^s)\in\Omega(k_1,m_1,k_2,m_2)\cap\{n_p^v,n_p^s\geq 1\}} P(\underline{N}^{v-},\underline{N}^{s-}) = \sum_{(\underline{N}^v,\underline{N}^s)\in\Omega(k_1,m_1,k_2,m_2)} n_p^s P(\underline{N}^v,\underline{N}^s)$$
(D-1)

and

$$\frac{\beta}{\alpha+\beta}\rho_p^v \sum_{(\underline{N}^v,\underline{N}^s)\in\Omega(k_1,m_1,k_2,m_2)\cap\{n_p^v,n_p^s\geq 1\}} P(\underline{N}^{v-},\underline{N}^s) = \sum_{(\underline{N}^v,\underline{N}^s)\in\Omega(k_1,m_1,k_2,m_2)} (n_p^v-n_p^s)P(\underline{N}^v,\underline{N}^s)$$
(D-2)

where

$$\begin{split} \Omega(k_{1}, m_{1}, k_{2}, m_{2}) &\cap \{n_{p}^{v}, n_{p}^{s} \geq 1\} \\ &= \left\{ (\underline{N}^{v}, \underline{N}^{s}) : 1 \leq n_{p}^{s} \leq n_{p}^{v}, \ 0 \leq n_{q}^{s} \leq n_{q}^{v}; \ q \neq p, \ q \in \mathcal{P}; \\ &\sum_{q \in \mathcal{P}_{i}, q \neq p} r_{q}^{(i)} n_{q}^{s} + r_{p}^{(i)} (n_{p}^{s} - I(p \in \mathcal{P}_{i})) = m_{i} - r_{p}^{(i)} I(p \in \mathcal{P}_{i}) \\ &\sum_{q \in \mathcal{P}_{i}, q \neq p} r_{q}^{(i)} n_{q}^{v} + r_{p}^{(i)} (n_{p}^{v} - I(p \in \mathcal{P}_{i})) = k_{i} - r_{p}^{(i)} I(p \in \mathcal{P}_{i}), \ i = 1, 2 \end{split}$$

The LHS of (D-2) becomes

$$\frac{\alpha}{\alpha+\beta}\rho_p^v P(k_1-r_p^{(1)}I(p\in\mathcal{P}_1),m_1-r_p^{(1)}I(p\in\mathcal{P}_1),k_2-r_p^{(2)}I(p\in\mathcal{P}_2),m_2-r_p^{(2)}I(p\in\mathcal{P}_2))$$

Note that if $m_i = 0$ then $p \notin \mathcal{P}_i$, (since $m_i = 0$ implies that $n_p^s = 0, \forall p \in \mathcal{P}_i$). The RHS of (D-2) can be written as

$$P(k_1,m_1,k_2,m_2)\cdot\sum_{(\underline{N}^{v},\underline{N}^{s})\in\Omega(k_1,m_1,k_2,m_2)}n_{p}^{s}\frac{P(\underline{N}^{v},\underline{N}^{s})}{P(k_1,m_1,k_2,m_2)}$$

where

$$P\left((\underline{N^{v}}, \underline{N^{s}}) \mid \sum_{p \in \mathcal{P}_{i}} r_{p}^{(i)} n_{p}^{v} = k_{i}, \sum_{p \in \mathcal{P}_{i}} r_{p}^{(i)} n_{p}^{s} = m_{i}, i = 1, 2\right)$$
$$= \begin{cases} \frac{P(\underline{N^{v}}, \underline{N^{s}})}{P(k_{1}, m_{1}, k_{2}, m_{2})}; & (\underline{N^{v}}, \underline{N^{s}}) \in \Omega(k_{1}, m_{1}, k_{2}, m_{2})\\ 0; & \text{otherwise} \end{cases}$$

Since the RHS has thus taken effectively the form of the conditional expectation of n_p^s given (k_1, m_1, k_2, m_2) , by combining the RHS and the LHS we conclude that

$$\frac{\alpha}{\alpha+\beta}\rho_p^v P\left(k_1 - r_p^{(1)}I(p \in \mathcal{P}_1), m_1 - r_p^{(1)}I(p \in \mathcal{P}_1), k_2 - r_p^{(2)}I(p \in \mathcal{P}_2), m_2 - r_p^{(2)}I(p \in \mathcal{P}_2)\right) = E[n_p^s|(k_1, m_1, k_2, m_2)] \cdot P(k_1, m_1, k_2, m_2) \qquad (D-3)$$

From equation (D-2) we can obtain in a similar way the equation

$$\frac{\beta}{\alpha+\beta}\rho_p^{\nu}P\left(k_1 - r_p^{(1)}I(p \in \mathcal{P}_1), m_1, k_2 - r_p^{(2)}I(p \in \mathcal{P}_2), m_2\right) = E[n_p^{\nu} - n_p^{s}](k_1, m_1, k_2, m_2)] \cdot P(k_1, m_1, k_2, m_2) \qquad (D-4)$$

If we multiply equations (D-3) and (D-4) by $r_p^{(i)}$ and sum with respect to $p \in \mathcal{P}_i$, for i = 1, 2, we obtain
$$\begin{split} P(k_{1},m_{1},k_{2},m_{2}) = \\ & \left\{ \begin{array}{l} \frac{1}{m_{1}} \frac{\alpha}{\alpha+\beta} \Big[\sum_{p \in \mathcal{P}_{1} \cap \mathcal{P}_{2}^{c}} r_{p}^{(1)} \rho_{p}^{v} P(k_{1}-r_{p}^{(1)},m_{1}-r_{p}^{(1)},k_{2},m_{2}) + \sum_{p \in \mathcal{P}_{1} \cap \mathcal{P}_{2}^{c}} r_{p}^{(1)} \rho_{p}^{v} P(k_{1}-r_{p}^{(1)},m_{1}-r_{p}^{(1)},k_{2}-r_{p}^{(2)},m_{2}-r_{p}^{(2)}) \Big] \\ & \text{if } r_{p}^{(1)} \leq m_{1} \leq k_{1} \leq Z_{1}, r_{p}^{(2)} \leq m_{2} \leq k_{2} \leq Z_{2}; \\ \frac{1}{m_{1}} \cdot \frac{\alpha}{\alpha+\beta} \Big[\sum_{p \in \mathcal{P}_{1} \cap \mathcal{P}_{2}^{c}} r_{p}^{(1)} \rho_{p}^{v} P(k_{1}-r_{p}^{(1)},m_{1}-r_{p}^{(1)};k_{2},m_{2}) \Big], \\ & \text{if } r_{p}^{(1)} \leq m_{1} \leq k_{1} \leq Z_{1}, 0 = m_{2} \leq k_{2} \leq Z_{2}; \\ \frac{1}{m_{2}} \cdot \frac{\alpha}{\alpha+\beta} \Big[\sum_{p \in \mathcal{P}_{1} \cap \mathcal{P}_{2}^{c}} r_{p}^{(2)} \rho_{p}^{v} P(k_{1},m_{1};k_{2}-r_{p}^{(2)},m_{2}-r_{p}^{(2)}) \Big], \\ & \text{if } 0 = m_{1} \leq k_{1} \leq Z_{1}; r_{p}^{(2)} \leq m_{2} \leq k_{2} \leq Z_{2}; \quad (D-5) \\ \frac{1}{k_{1}-m_{1}} \cdot \frac{\beta}{\alpha+\beta} \Big[\sum_{p \in \mathcal{P}_{2} \cap \mathcal{P}_{1}^{c}} r_{p}^{(1)} \rho_{p}^{v} P(k_{1}-r_{p}^{(1)},m_{1};k_{2},m_{2}) + \sum_{p \in \mathcal{P}_{1} \cap \mathcal{P}_{2}} r_{p}^{(1)} \rho_{p}^{v} P(k_{1}-r_{p}^{(1)},m_{1};k_{2}-r_{p}^{(2)},m_{2}) \Big] \\ & \text{if } 0 = m_{1} < r_{p}^{(1)} \leq k_{1} \leq Z_{1}, 0 = m_{2} < r_{p}^{(2)} \leq k_{2} \leq Z_{2}; \\ 1; & \text{if } k_{1} = 0, m_{1} = 0, k_{2} = 0, m_{2} = 0 \\ 0; & \text{if } k_{1}, m_{1} \text{ are not positive integer multiples of } r_{p}^{(1)} \\ & \text{or } k_{2}, m_{2} \text{ are not positive integer multiples of } r_{p}^{(1)} \\ \end{array} \right\}$$

which provide the desirable recursions for evaluating $P(k_1, m_1, k_2, m_2)$ for all (k_1, m_1, k_2, m_2) after performing the normalization

$$P'(k_1, m_1, k_2, m_2) = P(k_1, m_1, k_2, m_2)/P(0, 0, 0, 0).$$

APPENDIX E

Proof of Proposition 6.1

(Pascal Approximation)

When $c = \infty$, we may rewrite equations (6.7) as

$$\hat{P}(k,m) = \begin{cases} \frac{\lambda_{1,k-1}}{k} \cdot \hat{P}(k-1,0), & m = 0; \ 1 \le k \le \infty \\ \frac{\lambda_{2,m-1}}{m} \cdot \hat{P}(k-1,m-1), & 1 \le m \le k; \ 1 \le k \le \infty \end{cases}$$

where

$$\sum_{k=0}^{\infty}\sum_{m=0}^{k}\hat{P}(k,m)=1.$$

To express $\hat{P}(k,m)$ as a function of $\hat{P}(0,0)$, we write

$$\hat{P}(k,m) = \begin{cases} \frac{\prod_{i=0}^{k-1} \lambda_{1,i}}{m!} \cdot \hat{P}(0,0), & m = 0; \ 1 \le k \le \infty \\ \frac{\prod_{i=0}^{m-1} \lambda_{2,i}}{m!} \frac{\prod_{i=0}^{k-m-1} \lambda_{1,i}}{(k-m)!} \cdot \hat{P}(0,0), & 1 \le m \le k-1, \ 2 \le k \le \infty \\ \frac{\prod_{i=0}^{m-1} \lambda_{2,i}}{m!} \cdot \hat{P}(0,0), & m = k, \ 1 \le k \le \infty \end{cases}$$

and equation

$$\sum_{k=0}^{\infty}\sum_{m=0}^{k}\hat{P}(k,m)=1$$

becomes

$$\hat{P}(0,0) + \sum_{k=1}^{\infty} \hat{P}(k,0) + \sum_{k=1}^{\infty} \hat{P}(k,k) + \sum_{k=2}^{\infty} \sum_{m=1}^{k-1} \hat{P}(k,m) = 1$$

which results in

$$\left[1+\sum_{k=1}^{\infty}\frac{\prod_{i=0}^{k-1}\lambda_{1,i}}{k!}+\sum_{k=1}^{\infty}\frac{\prod_{i=0}^{k-1}\lambda_{2,i}}{k!}+\sum_{k=2}^{\infty}\sum_{m=1}^{k-1}\frac{\prod_{i=0}^{m-1}\lambda_{2,i}}{m!}\cdot\frac{\prod_{i=0}^{k-m-1}\lambda_{1,i}}{(k-m)!}\right]\cdot\hat{P}(0,0)=1.$$

Let

$$a_j = \frac{\epsilon_j^2}{\sigma_j^2}$$

$$b_j = 1 - \frac{\epsilon_j}{\sigma_j^2}$$

for j = 1, 2. Then by definition,

$$\lambda_{j,i} = a_j + i \cdot b_j$$

for j = 1, 2. After certain manipulations, we have

$$1 + \sum_{k=1}^{\infty} \left(\frac{\prod_{i=0}^{k-1} \lambda_{1,i}}{k!} + \frac{\prod_{i=0}^{k-1} \lambda_{2,i}}{k!} \right) + \sum_{k=2}^{\infty} \sum_{m=1}^{k-1} \frac{\prod_{i=0}^{m-1} \lambda_{2,i}}{m!} \frac{\prod_{i=0}^{k-m-1} \lambda_{1,i}}{(k-m)!}$$
$$= \left(1 + \sum_{k=1}^{\infty} \frac{\prod_{i=0}^{k-1} \lambda_{1,i}}{k!} \right) \cdot \left(1 + \sum_{k=1}^{\infty} \frac{\prod_{i=0}^{k-1} \lambda_{2,i}}{k!} \right)$$
$$= (1 - b_1)^{-a_1/b_1} \cdot (1 - b_2)^{-a_2/b_2}$$

Then

$$\hat{P}(0,0) = (1-b_1)^{a_1/b_1} \cdot (1-b_2)^{a_2/b_2}$$

Consider the moment generating function

$$\begin{split} E[T^k S^m] &= \hat{P}(0,0) + \sum_{k=1}^{\infty} [\hat{P}(k,0)T^k + \hat{P}(k,k)T^k S^k] + \sum_{k=2}^{\infty} \sum_{m=1}^{k-1} \hat{P}(k,m)T^k S^m \\ &= \hat{P}(0,0) \cdot \left[1 + \sum_{k=1}^{\infty} \left(\frac{\prod_{i=0}^{k-1} (\lambda_{1,i} \cdot T)}{k!} + \frac{\prod_{i=0}^{k-1} (\lambda_{2,i} \cdot T \cdot S)}{k!} \right) \right. \\ &+ \sum_{k=2}^{\infty} \sum_{m=1}^{k-1} \frac{\prod_{i=0}^{m-1} (\lambda_{2,i} \cdot T \cdot S)}{m!} \cdot \frac{\prod_{i=0}^{k-m-1} (\lambda_{1,i} \cdot T)}{(k-m)!} \right]. \end{split}$$

Note that

$$\lambda_{1,i} \cdot T = a_1 \cdot T + ib_1 \cdot T$$

$$\lambda_{2,i} \cdot TS = a_2 \cdot TS + ib_2 \cdot T \cdot S$$

Therefore,

$$E[T^k S^m] = \left(\frac{1-b_1 T}{1-b_1}\right)^{-\frac{a_1}{b_1}} \left(\frac{1-b_2 T S}{1-b_2}\right)^{-\frac{a_2}{b_2}}.$$

Then the first and second moments of k and m are

-

$$E[k-m] = E[k] - E[m] = \frac{\partial E[T^k S^m]}{\partial T} \bigg|_{T=S=1} - \frac{\partial E[T^k S^m]}{\partial S} \bigg|_{T=S=1}$$
$$= \left(\frac{a_2}{1-b_2} + \frac{a_1}{1-b_1}\right) - \frac{a_2}{1-b_2}$$
$$= \frac{a_1}{1-b_1} = \epsilon_1.$$

$$E[m] = \frac{\partial E[T^k S^m]}{\partial S} \bigg|_{T=S=1}$$
$$= \frac{a_2}{1-b_2} = \epsilon_2.$$

$$var(m) = E[m^{2}] - E[m]^{2} = \frac{\partial^{2} E[T^{k} S^{m}]}{\partial S^{2}} \bigg|_{T=S=1} - E[m]^{2}$$
$$= \left(\frac{a_{2}^{2} + a_{2}b_{2}}{(1-b_{2})^{2}} + \frac{a_{2}}{(1-b_{2})}\right) - \left(\frac{a_{2}^{2}}{(1-b_{2})}\right)^{2}$$
$$= \frac{a_{2}}{(1-b_{2})^{2}} = \sigma_{2}^{2}.$$

.

$$\begin{aligned} var(k-m) &= \frac{\partial^2 E[T^k S^m]}{\partial T^2} \bigg|_{T=S=1} - 2\frac{\partial^2 E[T^k S^m]}{\partial T \partial S} \bigg|_{T=S=1} + var(m) + E[k] \cdot (2E[m] - E[k]) \\ &= \frac{a_1}{(1-b_1)^2} = \sigma_1^2. \\ cov(k-m,m) &= E[(k-m)m] - E[k-m] \cdot E[m] \\ &= \frac{\partial^2 E[T^k S^m]}{\partial T \partial S} \bigg|_{T=S=1} - E[k]E[m] - var(m) \\ &= 0. \end{aligned}$$

APPENDIX F

Proof of Proposition 6.2

(Pascal Approximation)

For the single-link, multi-rate, infinite capacity scheme, the steady-state probability of voice $P(\underline{N}^{v}, \underline{N}^{s})$ in Section 3 is

$$P(\underline{N^{v}},\underline{N^{s}}) = \frac{1}{G} \cdot \prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{s}!(n_{p}^{v} - n_{p}^{s})!} \frac{\alpha^{n_{p}^{s}}\beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v}}},$$

where

$$G = \sum_{\substack{0 \le n_p^{v} \le \infty \\ 0 \le n_p^{v} \le n_p^{v}, p \in \mathcal{P}}} \prod_{p \in \mathcal{P}} \frac{(\rho_p^{v})^{n_p^{v}}}{n_p^{s!}(n_p^{v} - n_p^{s})!} \frac{\alpha^{n_p^{s}} \beta^{n_p^{v} - n_p^{s}}}{(\alpha + \beta)^{n_p^{v}}}$$

and it is approximated by

$$G \approx \prod_{p \in \mathcal{P}} \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} \frac{(\rho_p^v)^{n_p^v}}{n_p^{s!}(n_p^v - n_p^s)!} \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s}}{(\alpha + \beta)^{n_p^v}}.$$

which means that the voice states of the different paths are assumed mutually independent in this single-link analysis. This is motivated by the same interlink blocking independence assumption discussed in Sections 4.1 and 5.1. Therefore, we may define

$$\tilde{P}(\underline{N}^{v},\underline{N}^{s}) = \prod_{p \in \mathcal{P}} p(n_{p}^{v},n_{p}^{s})$$

where

$$p(n_p^v, n_p^s) = \frac{(\rho_p^v)^{n_p^v} \frac{\alpha^{n_p^s} \beta^{n_p^s - n_p^s}}{(\alpha + \beta)n_p^v}}{n_p^s! (n_p^v - n_p^s)!} \cdot \frac{1}{G_p}$$

and

$$G_p = \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^v \le n_p^v}} \frac{(\rho_p^v)^{n_p^v} \frac{\alpha^{n_p^v \beta n_p^v - n_p^s}}{(\alpha + \beta) n_p^v}}{n_p^s! (n_p^v - n_p^s)!}$$

In the following we use $\tilde{P}(\underline{N}^{v}, \underline{N}^{s})$ instead of $P(\underline{N}^{v}, \underline{N}^{s})$ and $\tilde{P}(\underline{N}^{v})$ instead of $P(\underline{N}^{v})$ to distinguish the approximations from the exact values. Next we evaluate the first and second moments of $(\underline{N}^{v}, \underline{N}^{s})$ as follows

$$E[n_{v}] = \sum_{\substack{0 \le n_{p}^{v} \le \infty \\ 0 \le n_{p}^{v} \le n_{p}^{v}, p \in \mathcal{P}}} \sum_{p \in \mathcal{P}} r_{p} n_{p}^{v} \cdot \tilde{P}(\underline{N}^{v}, \underline{N}^{s}) = \sum_{\substack{0 \le n_{p}^{v} \le \infty \\ p \in \mathcal{P}}} \sum_{p \in \mathcal{P}} r_{p} n_{p}^{v} \cdot \tilde{P}(\underline{N}^{v})$$

where

$$\tilde{P}(\underline{N^{v}}) = \left[\prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{v!}}\right] \middle/ \left[\sum_{\substack{0 \leq n_{p}^{v} \leq \infty \\ p \in \mathcal{P}}} \prod_{p \in \mathcal{P}} \frac{(\rho_{p}^{v})^{n_{p}^{v}}}{n_{p}^{v!}}\right] = \prod_{p \in \mathcal{P}} p(n_{p}^{v})$$

and

$$p(n_p^v) = \left[\frac{(\rho_p^v)n_p^v}{n_p^{v!}}\right] \middle/ \left[\sum_{\substack{0 \le n_p^v \le \infty}} \frac{(\rho_p^v)n_p^v}{n_p^{v!}}\right]$$

 $\tilde{P}(\underline{N}^{v})$ is the approximate steady-state probability of state \underline{N}^{v} , where the state vector \underline{N}^{v} represents the number of channels occupied by voice taking path p, for $p \in \mathcal{P}$. Combining these results yields

$$E[n_v] = \sum_{\substack{0 \le n_v^v \le \infty \\ p \in \mathcal{P}}} \sum_{p \in \mathcal{P}} r_p n_p^v \prod_{p \in \mathcal{P}} p(n_p^v) = \sum_{p \in \mathcal{P}} r_p \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^v \le \infty}} n_p^v p(n_p^v) = \sum_{p \in \mathcal{P}} r_p \rho_p^v$$

and similarly

$$E[n_s] = \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v, p \in \mathcal{P}}} \sum_{p \in \mathcal{P}} r_p n_p^s \prod_{p \in \mathcal{P}} p(n_p^v, n_p^s) = \sum_{p \in \mathcal{P}} r_p \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} n_p^s p(n_p^v, n_p^s).$$

Note that using the definition of $p(n_p^v, n_p^s)$ above we can express $n_p^s \cdot p(n_p^v, n_p^s)$ as

$$n_{p}^{s} \cdot p(n_{p}^{v}, n_{p}^{s}) = \frac{(\rho_{p}^{v})^{n_{p}^{v}-1} \frac{\alpha^{n_{p}^{s}-1} \beta^{n_{p}^{v}-n_{p}^{s}}}{(\alpha+\beta)^{n_{p}^{v}-1}}}{(n_{p}^{s}-1)!(n_{p}^{v}-n_{p}^{s})!} \cdot \frac{\alpha \rho_{p}^{v}}{\alpha+\beta} \cdot \frac{1}{G_{p}}$$

and thus through summation obtain

$$\sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} n_p^s p(n_p^v, n_p^s) = \sum_{\substack{1 \le n_p^v \le \infty \\ 1 \le n_p^s \le n_p^v}} n_p^s p(n_p^v, n_p^s) = \frac{\alpha}{\alpha + \beta} \rho_p^v.$$

Therefore,

$$E[n_s] = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p \rho_p^v.$$

Since n_v depends on \underline{N}^v but not on \underline{N}^o , we can use directly the result of [6] for voice-only traffic (without silent periods):

$$var(n_v) = \sum_{p \in \mathcal{P}} r_p(\rho_p^v)^2.$$

Moreover,

$$\begin{split} E[n_s^2] &= \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v, p \in \mathcal{P}}} \left(\sum_{p \in \mathcal{P}} r_p n_p^s \right)^2 \tilde{P}(\underline{N}^v, \underline{N}^s) \\ &= \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v, p \in \mathcal{P}}} \left(\sum_{p \in \mathcal{P}} r_p^2 n_p^{s2} \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_p n_p^s r_q n_q^s \right) \tilde{P}(\underline{N}^v, \underline{N}^s) \\ &= \sum_{p \in \mathcal{P}} r_p \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} n_p^{s2} p(n_p^v, n_p^s) + \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_p r_q \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} n_p^{s2} p(n_q^v, n_p^s) + \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_p r_q \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v \otimes n_p^s \le n_p^v}} n_p^v n_p^s \cdot p(n_q^v, n_q^s). \end{split}$$

•

.

_ _ _ _ _ _ _ _ _

$$\begin{split} \sum_{\substack{a \leq n_{p}^{v} \leq \infty \\ 0 \leq n_{p}^{v} \leq n_{p}^{v}}} n_{p}^{s^{2}} p(n_{p}^{v}, n_{p}^{s}) \\ &= \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} (n_{p}^{s} - 1 + 1) \frac{\rho_{p}^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v} - 1}} \cdot \frac{\alpha \rho_{p}^{v}}{\alpha + \beta} \cdot \frac{1}{G_{p}} \\ &= \frac{\alpha}{\alpha + \beta} \rho_{p}^{v} \left[\sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} (n_{p}^{s} - 1) \frac{\rho_{p}^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v} - 1}} + \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(\alpha + \beta)^{n_{p}^{v} - 1}} + \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(n_{p}^{s} - 1)! (n_{p}^{v} - n_{p}^{s})!} + \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(n_{p}^{s} - 1)! (n_{p}^{v} - n_{p}^{s})!} + \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{s} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(n_{p}^{s} - 1)! (n_{p}^{v} - n_{p}^{s})!} + \sum_{\substack{1 \leq n_{p}^{v} \leq \infty \\ 1 \leq n_{p}^{v} \leq n_{p}^{v}}} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{v} - 1} \beta^{n_{p}^{v} - n_{p}^{s}}}{(n_{p}^{s} - 2)! (n_{p}^{v} - n_{p}^{s})!} \cdot \frac{\alpha}{\alpha + \beta} \rho_{p}^{v} + G_{p} \right] \cdot \frac{1}{G_{p}} \\ = \frac{\alpha}{\alpha + \beta} \rho_{p}^{v} \cdot \left(1 + \frac{\alpha}{\alpha + \beta} \rho_{p}^{v}\right) \end{split}$$

and

$$\sum_{\substack{0 \le n_y^v \le \infty \\ 0 \le n_q^v \le n_q^v = \alpha_p^v \le \alpha_p^v}} \sum_{\substack{n_p^s n_q^s \cdot p(n_p^v, n_p^s) \cdot p(n_q^v, n_q^s) = \sum_{\substack{0 \le n_y^v \le \infty \\ 0 \le n_q^v \le n_y^v = \alpha_p^v = \alpha_p^v}} n_p^s \cdot p(n_p^v, n_p^s) \sum_{\substack{0 \le n_y^v \le \infty \\ 0 \le n_q^v \le n_y^v = \alpha_p^v = \alpha_q^v}} n_q^s \cdot p(n_q^v, n_q^s)$$

$$= \frac{\alpha}{\alpha + \beta} \rho_p^v \cdot \frac{\alpha}{\alpha + \beta} \rho_q^v$$

.

we obtain

$$\begin{split} E[n_s^2] &= \sum_{p \in \mathcal{P}} r_p \left(\frac{\alpha}{\alpha + \beta} \rho_p^v + \left(\frac{\alpha}{\alpha + \beta} \rho_p^v \right)^2 \right) + \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_p r_q \rho_p^v \rho_q^v \left(\frac{\alpha}{\alpha + \beta} \right)^2 \\ &= \sum_{p \in \mathcal{P}} r_p \frac{\alpha}{\alpha + \beta} \rho_p^v + \left(\sum_{p \in \mathcal{P}} r_p \frac{\alpha}{\alpha + \beta} \rho_p^v \right)^2. \end{split}$$

From the above expressions for $E[n_s]$ and $E[n_s^2]$ we can easily derive that

$$var(n_s) = E[n_s^2] - (E[n_s])^2 = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v.$$

Similarly, we can write

$$\begin{split} E[(n_v - n_s)^2] &= \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} \left(\sum_{p \in \mathcal{P}} r_p (n_p^v - n_p^s) \right)^2 \tilde{P}(\underline{N}^v, \underline{N}^s) \\ &= \sum_{p \in \mathcal{P}} r_p^2 \sum_{\substack{0 \le n_q^v \le \infty \\ 0 \le n_q^v \le n_q^v}} (n_p^v - n_p^s)^2 p(n_p^v, n_p^s) \\ &+ \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_p r_q \cdot \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} (n_p^v - n_p^s) p(n_p^v, n_p^s) \sum_{\substack{0 \le n_q^v \le \infty \\ 0 \le n_q^v \le n_q^v}} (n_p^v - n_q^s) p(n_p^v, n_p^s) \sum_{\substack{0 \le n_q^v \le \infty \\ 0 \le n_q^v \le n_q^v}} (n_p^v - n_p^s) p(n_p^v, n_p^s) \sum_{\substack{0 \le n_q^v \le \infty \\ 0 \le n_q^v \le n_q^v}} (n_p^v - n_q^s) p(n_p^v, n_q^s) \\ \end{split}$$

and since

$$\begin{split} \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^v \le \infty \\ 0 \le n_p^v \le m_p^v}} (n_p^v - n_p^s)^2 (n_p^v, n_p^s) &= \sum_{\substack{1 \le n_p^v \le \infty \\ 0 \le n_p^v \le m_p^v - 1}} (n_p^v - n_p^s - 1 + 1) \frac{(\rho_p^v - 1) \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s - 1}}{(\alpha + \beta)^{n_p^v - 1}}}{(n_p^s)! (n_p^v - n_p^s - 1)!} \frac{\beta}{\alpha + \beta} \rho_p^v \cdot \frac{1}{G_p} \\ &= \left[\sum_{\substack{2 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v - 2}} \frac{\rho_p^{n_p^v - 2} \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s - 2}}{(\alpha + \beta)^{n_p^v - 2}}}{(n_p^s)! (n_p^v - n_p^s - 1)!} \frac{\beta}{\alpha + \beta} \rho_p^v + G_p \right] \cdot \frac{\beta}{\alpha + \beta} \rho_p^v \cdot \frac{1}{G_p} \\ &= \frac{\beta}{\alpha + \beta} \rho_p^v \cdot (1 + \frac{\beta}{\alpha + \beta} \rho_p^v) \end{split}$$

we derive that

$$E[(n_v - n_s)^2] = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v + \left(\frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p \rho_p^v\right)^2$$

and conclude that

$$var(n_v - n_s) = E[(n_v - n_s)^2] - (E[n_v - n_s])^2 = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v.$$

112

Finally, we write

$$\begin{split} E[n_{v} \cdot n_{s}] &= \sum_{\substack{0 \le n_{y}^{v} \le \infty \\ 0 \le n_{p}^{s} \le n_{p}^{v}}} \left(\sum_{p \in \mathcal{P}} r_{p} n_{p}^{v} \right) \left(\sum_{p \in \mathcal{P}} r_{p} n_{p}^{s} \right) \tilde{P}(\underline{N}^{v}, \underline{N}^{s}) \\ &= \sum_{p \in \mathcal{P}} r_{p}^{2} \sum_{\substack{0 \le n_{y}^{v} \le \infty \\ 0 \le n_{p}^{v} \le n_{p}^{v}}} n_{p}^{v} n_{p}^{s} p(n_{p}^{v}, n_{p}^{s}) \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_{p} r_{q} \left(\sum_{\substack{0 \le n_{y}^{v} \le \infty \\ 0 \le n_{p}^{v} \le n_{p}^{v}}} n_{p}^{v} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{y}^{v} \le \infty \\ 0 \le n_{p}^{v} \le n_{p}^{v}}} n_{p}^{s} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{y}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{p}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le \infty \\ 0 \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le n_{q}^{v} \le n_{q}^{v} \le n_{q}^{v}}} n_{q}^{v} p(n_{q}^{v}, n_{q}^{s}) \right) \left(\sum_{\substack{0 \le n_{q}^{v} \le n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} p(n_{q}^{v} \in n_{q}^{v} \le n_{q}^{v} \ge n_{q}^{v} p(n_{q}^{v} \in n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} \ge n_{q}^{v} p(n_{q}^{v} \in n_{q}^{v} \ge n_{q}^{v} \ge$$

and since we have that

$$\begin{split} \sum_{\substack{0 \le n_p^v \le \infty \\ 0 \le n_p^v \le \infty \\ 0 \le n_p^s \le n_p^v}} n_p^v n_p^s p(n_p^v, n_p^s) &= \frac{\alpha}{\alpha + \beta} \rho_p^v \cdot \left[\sum_{\substack{0 \le n_p^v \le \infty \\ 1 \le n_p^s \le n_p^v}} (n_p^v + 1) \frac{\rho_p^{n_p^v} \frac{\alpha^{n_p^v} \beta^{n_p^v} - n_p^s}{(\alpha + \beta)^{n_p^v}}}{[n_p^s]! (n_p^v - n_p^s)!} \right] \cdot \frac{1}{G_p} \\ &= \frac{\alpha}{\alpha + \beta} \rho_p^v \left[\sum_{\substack{0 \le n_p^v \le \infty \\ n_p^s \le \infty}} n_p^v p(n_p^v) + 1 \right] \\ &= \frac{\alpha}{\alpha + \beta} \rho_p^v (1 + \rho_p^v) \end{split}$$

we derive that

$$E[n_{v} \cdot n_{s}] = \sum_{p \in \mathcal{P}} r_{p}^{2} \frac{\alpha}{\alpha + \beta} (\rho_{p}^{v} + \rho_{p}^{v2}) \sum_{p \in \mathcal{P}} \sum_{\substack{q \in \mathcal{P} \\ q \neq p}} r_{p} r_{q} \frac{\alpha}{\alpha + \beta} \rho_{p}^{v} \rho_{q}^{v}$$
$$= \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_{p}^{2} \rho_{p}^{v} + \frac{\alpha}{\alpha + \beta} \left(\sum_{p \in \mathcal{P}} r_{p} \rho_{p}^{v} \right)^{2}$$

and conclude that

$$cov[n_v, n_s] = E[n_v \cdot n_s] - E[n_v] \cdot E[n_s] = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}} r_p^2 \rho_p^v.$$

Consequently, since

$$cov(n_v - n_s, n_s) = E[(n_v - n_s)n_s] - E[n_v - n_s] \cdot E[n_s] = cov(n_v, n_s) - var(n_s)$$

and as we established above $cov(n_v, n_s) = var(n_s)$, we also obtain that

$$cov(n_v-n_s,n_s)=0.$$

APPENDIX G

The Knapsack and Pascal Approximations to the Probabilities of Voice Blocking and Data Queueing Are Identical for Networks with Single-Rate Traffic

In this appendix we prove the validity of the claim made in the title. We show that the knapsack and Pascal methods provide identical approximations to the main performance measures of this report: the probability of blocking voice (Section 4) and the probability of queueing data (Sections 5 and 6). The same is true for the average queueing delay of data (Section 7), as well as for the probability distribution of the data queueing delay, but we omit the proof since it resembles the one for the probability of data queueing.

The single-rate case is characterized by

$$r_p = 1 \text{ for all } p \in \mathcal{P}$$

where 1 is the required data rate (bandwidth) of voice; the required bandwidth for data is then 1/r (where $r \ge 1$).

G.1 Probability of Voice Blocking for Single-Rate Traffic

Knapsack Approximation

In the single-rate case, the recursion of equations (4.6a)-(4.6b), which provides the weights necessary for the knapsack approximation to the voice blocking probability of (4.5), reduces to

$$w(n) = \frac{1}{n} \left(\sum_{p \in \mathcal{P}_l} \rho_p^v \right) w(n-1) \text{ for } n = 1, 2, \ldots, c_l.$$

Pascal Approximation

In the single-rate case, the recursion of equations (4.11b)-(4.12b), which provides the weights necessary for the Pascal approximation to the voice blocking probability of (4.13b), reduces to

$$q'(n) = \frac{\lambda_{n-1}}{n}q'(n-1)$$
 for $n = 1, 2, ..., c_n$

114

where

$$\lambda_{n-1} = \frac{\epsilon^2}{\sigma^2} + (n-1)\left(1 - \frac{\epsilon}{\sigma^2}\right)$$

with

$$\epsilon = E\{n_v\} = \sum_{p \in \mathcal{P}_l} r_p \rho_p^v = \sum_{p \in \mathcal{P}_l} \rho_p^v$$

and

$$\sigma^2 = var\{n_v\} = \sum_{p \in \mathcal{P}_l} r_p^2 \rho_p^v = \sum_{p \in \mathcal{P}_l} \rho_p^v.$$

Since $\epsilon = \sigma^2$, λ_{n-1} simplifies to

$$\lambda_{n-1} = \epsilon = \sum_{p \in \mathcal{P}_l} \rho_p^v$$

resulting in

$$q'(n) = \frac{1}{n} \left(\sum_{p \in \mathcal{P}_l} \rho_p^v \right) q'(n-1).$$

Since as we know

$$w(0)=q'(0)=1$$

comparison of the above simplified recursions for the knapsack and Pascal approximations shows that they are identical, that is

$$w(n) = q'(n)$$
 for $n = 0, 1, ..., c_l$.

Therefore, since the form of the approximations in (4.5) and (4.13b) is the same except for the w(n) and q'(n), the two approximations coincide in the single-rate case.

G.2 Probability of Data Queueing for Single-Rate Traffic

Knapsack Approximation

In the single-rate case, the recursion providing the quantities P'(k,m) of (5.10a) involved in the knapsack approximation to the probability of data queueing given by (5.9) reduces to

$$P'(k,m) = \begin{cases} \frac{1}{m} \left(\left[\alpha/(\alpha+\beta) \right] \sum_{p \in \mathcal{P}_l} \rho_p^v \right) P'(k-1,m-1) & 1 \le m \le k \le c_l \\ \frac{1}{k} \left(\left[\beta/(\alpha+\beta) \right] \sum_{p \in \mathcal{P}_l} \rho_p^v \right) P'(k-1,0) & 0 = m < k \le c_l. \end{cases}$$

Pascal Approximation

Similarly, the recursion providing the corresponding quantities P'(k,m) of (6.7b) involved in the Pascal approximation to the probability of data queueing given by (6.21) reduces in the single-rate case to

$$\hat{P}'(k,m) = \begin{cases} (\lambda_{2,m-1}/m)\hat{P}'(k-1,m-1) & 1 \le m \le k \le c_l \\ (\lambda_{1,k-1}/k)\hat{P}'(k-1,0) & m = 0 \end{cases}$$

where

$$\lambda_{2,m-1} = \frac{\epsilon_2^2}{\sigma_2^2} + (m-1)\left(1 - \frac{\epsilon_2}{\sigma_2^2}\right)$$

with

$$\epsilon_2 = E\{m\} = E\{n_s\} = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} r_p \rho_p^v = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} \rho_p^v$$

and

$$\sigma_2^2 = var\{m\} = var\{n_p\} = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_i} r_p^2 \rho_p^v = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_i} \rho_p^v.$$

Since $\epsilon_2 = \sigma_2^2$, $\lambda_{2,m-1}$ simplifies to

$$\lambda_{2,m-1} = \epsilon_2 = \frac{\alpha}{\alpha + \beta} \sum_{p \in \mathcal{P}_i} \rho_p^v.$$

Similarly,

$$\lambda_{1,k-1} = \frac{\epsilon_1^2}{\sigma_1^2} + (k-1)\left(1 - \frac{\epsilon_1}{\sigma_1^2}\right)$$

and since

$$\epsilon_1 = E\{k-m\} = E\{n_v\} = \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_i} r_p \rho_p^v = \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_i} \rho_p^v$$

and

$$\sigma_1^2 = var\{k - m\} = var\{n_v\} = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} r_p^2 \rho_p^v = \frac{\beta}{\alpha + \beta} \sum_{p \in \mathcal{P}_l} \rho_p^v$$

imply that $\epsilon_1 = \sigma_1^2$, $\lambda_{1,k-1}$ simplifies to

$$\lambda_{1,k-1} = \epsilon_1 = \frac{\beta}{\alpha+\beta} \sum_{p \in \mathcal{P}_l} \rho_p^v.$$

Using the above simplified expressions for $\lambda_{1,k-1}$ and $\lambda_{2,m-1}$ in the recursion for $\hat{P}'(k,m)$ results in an expression whose functional form is identical to that obtained for P'(k,m) in the knapsack recursion earlier in this subsection after the simplifications. Since we also have that

$$P'(0,0) = \tilde{P}'(0,0) = 1$$

the knapsack and Pascal recursions are completely identical and thus

$$P'(k,m) = \tilde{P}'(k,m)$$
 for all $0 \le m \le k \le c_l$.

Moreover, since the expressions providing the approximations to the probability of data queueing in (5.9) and (6.21) have the same functional form for both the knapsack and Pascal recursions, we deduce that the approximations in question are identical for the single-rate case.

APPENDIX H

Proof of Proposition 10.1

The starting point is the definition

$$W^{d} = \sum_{l \in \mathcal{L}} \gamma_{l}^{d} \rho_{l}^{d} \cdot \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega} P(\underline{N}^{v}, \underline{N}^{s}) \cdot \sum_{n_{l}^{d} = 0}^{c_{l}^{\prime} - 1} P(n_{l}^{d} | c_{l}^{\prime}) \cdot I(c_{l}^{\prime} > 0)$$
$$= \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega} P(\underline{N}^{v}, \underline{N}^{s}) \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c})$$

from which we obtain

$$\frac{\partial W^d}{\partial \rho_p^v} = \sum_{(\underline{N}^v, \underline{N}^s) \in \Omega} \frac{\partial P(\underline{N}^v, \underline{N}^s)}{\partial \rho_p^v} \cdot w^d(\underline{\gamma}^d, \underline{\rho}^d, \underline{c}).$$

Define

$$f_p = \frac{(\rho_p^v)^{n_p^v}}{n_p^s!(n_p^v - n_p^s)!} \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s}}{(\alpha + \beta)^{n_p^v}}$$

and $G(\underline{c})$ as in (3.3); then from (3.2) we have

$$\frac{\partial P(\underline{N}^{v},\underline{N}^{s})}{\partial \rho_{p}^{v}} = \frac{\partial}{\partial \rho_{p}^{v}} \frac{1}{G(\underline{c})} \cdot \prod_{p \in \mathcal{P}} f_{p} + \frac{1}{G(\underline{c})} \cdot \prod_{q \in \mathcal{P}q \neq p} f_{q} \cdot \frac{\partial f_{p}}{\partial \rho_{p}^{v}}.$$

Moreover,

$$\begin{aligned} \frac{\partial f_p}{\partial \rho_p^v} &= \frac{(\rho_p^v)^{n_p^v - 1} \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s}}{(\alpha + \beta)^{n_p^v}}}{n_p^{s!} (n_p^v - n_p^s)!} \cdot (n_p^s + n_p^v - n_p^s) \\ &= \frac{\alpha}{\alpha + \beta} \cdot \frac{(\rho_p^v)^{n_p^v - 1} \frac{\alpha^{n_p^s - 1} \beta^{n_p^v - n_p^s}}{(\alpha + \beta)^{n_p^v - 1}}}{(n_p^s - 1)! (n_p^v - n_p^s)!} \cdot I(n_p^s > 0) \\ &+ \frac{\beta}{\alpha + \beta} \cdot \frac{\rho_p^{v n_p^v - 1} \frac{\alpha^{n_p^s} \beta^{n_p^v - n_p^s - 1}}{(\alpha + \beta)^{n_p^s - 1}}}{n_p^{s!} (n_p^v - n_p^s - 1)!} \cdot I(n_p^v - n_p^s > 0) \end{aligned}$$

and

$$\frac{\partial}{\partial \rho_p^v} \frac{1}{G(\underline{c})} = -[G(\underline{c})]^{-2} \sum_{(\underline{N}^v, \underline{N}^e) \in \Omega(\underline{c})} \prod_{q \in \mathcal{P}_q \neq p} f_q \cdot \frac{\partial f_p}{\partial \rho_p^v}.$$

Therefore

$$\begin{split} \frac{\partial W^{d}}{\partial \rho_{p}^{v}} &= -[G(\underline{c})]^{-2} \sum_{(\underline{N}^{v}, \underline{N}^{*}) \in \Omega(\underline{c})} \prod_{p \in \mathcal{P}} f_{p} \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \\ & \left\{ \frac{\alpha}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{*}) \in \Omega(\underline{c})} \left[\prod_{q \in \mathcal{P}_{q \neq p}} f_{q} \frac{(\rho_{p}^{v})^{n_{p}^{v} - 1} \frac{\alpha^{n_{p}^{*} - 1} \beta^{n_{p}^{*} - n_{p}^{*}}{(\alpha + \beta)^{n_{p}^{*} - 1}}}{(\alpha + \beta)^{n_{p}^{*} - 1}} I(n_{p}^{*} > 0) \right] \\ & + \frac{\beta}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{*}) \in \Omega(\underline{c})} \left[\prod_{q \in \mathcal{P}_{q \neq p}} f_{q} \frac{(\rho_{p}^{v})^{n_{p}^{*} - 1} \frac{\alpha^{n_{p}^{*} - n_{p}^{*} - 1}}{(\alpha + \beta)^{n_{p}^{*} - 1}}}{n_{p}^{*} (n_{p}^{v} - n_{p}^{*} - 1)!} \cdot I(n_{p}^{v} - n_{p}^{*} > 0) \right] \right\} \\ & + [G(\underline{c})]^{-1} \left\{ \frac{\alpha}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{*}) \in \Omega(\underline{c})} \left[\prod_{q \in \mathcal{P}_{q \neq p}} f_{q} \frac{(\rho_{p}^{v})^{n_{p}^{*} - 1} \frac{\alpha^{n_{p}^{*} - 1} \beta^{n_{p}^{*} - n_{p}^{*} - 1}}{(\alpha + \beta)^{n_{p}^{*} - 1}} I(n_{p}^{*} > 0) w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \right] \\ & + \frac{\beta}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{*}) \in \Omega(\underline{c})} \left[\prod_{q \in \mathcal{P}_{q \neq p}} f_{q} \frac{(\rho_{p}^{v})^{n_{p}^{*} - 1} \frac{\alpha^{n_{p}^{*} - 1} \beta^{n_{p}^{*} - n_{p}^{*} - 1}}{(\alpha + \beta)^{n_{p}^{*} - 1}} I(n_{p}^{*} > 0) w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \right] \right\} \end{aligned}$$

The above expression can be put in the simpler form

$$\frac{\partial W^{d}}{\partial \rho_{p}^{v}} = -[G(\underline{c})]^{-2} \sum_{(\underline{N}^{v}, \underline{N}^{e}) \in \Omega(\underline{c})} \left[\frac{\alpha}{\alpha + \beta} \cdot (A) + \frac{\beta}{\alpha + \beta} (B) \right] \cdot \prod_{p \in \mathcal{P}} f_{p} \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \\ + G(\underline{c})^{-1} \left[\frac{\alpha}{\alpha + \beta} \cdot (C) + \frac{\beta}{\alpha + \beta} (D) \right]$$

•

where the terms (A), (B), (C), (D) are appropriately defined.

The term (A) can be obtained as

$$(A) = \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p \underline{c}_{p}})} \prod_{p \in \mathcal{P}} f_{p} = G(\underline{c} - r_{p \underline{c}_{p}}).$$

119

In order to simplify (B), let us consider $\Omega(\underline{c}) \bigcap \{n_p^v > n_p^s\}$, which can be rewritten as

$$\begin{split} \Big\{ (\underline{N}^{v}, \underline{N}^{s}) | 0 &\leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v} \leq c_{l}, l \in \mathcal{L}; 0 \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{s} \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v}, l \notin p, \\ 0 &\leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{s} < \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v}, l \in p, l \in \mathcal{L} \Big\} = \\ &= \Big\{ (\underline{N}^{v}, \underline{N}^{s}) | 0 \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v} \leq c_{l}, 0 \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{s} \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{v}, l \notin p; \\ 0 &\leq \sum_{q \in \mathcal{P}_{l} q \neq p} r_{q} n_{q}^{v} + r_{p} (n_{p}^{v} - 1) \leq c_{l} - r_{p}, 0 \leq \sum_{q \in \mathcal{P}_{l}} r_{q} n_{q}^{s} \leq \sum_{q \in \mathcal{P}_{l} q \neq p} r_{q} n_{q}^{v} + r_{p} (n_{p}^{v} - 1), l \in p \Big\}. \end{split}$$

Therefore $\Omega(\underline{c}) \cap \{n_p^v > n_p^s\}$ for vector $(\underline{N}^v, \underline{N}^s)$ is identical to $\Omega(\underline{c} - r_p \underline{e}_p)$ for vector $(\underline{N}_p^{v-}, \underline{N}^s)$, where $\underline{N}_p^{v-} = \underline{N}^v - \underline{e}_p$, and (B) can be represented as

$$(B) = G(\underline{c} - r_p \underline{e}_p).$$

From a similar argument as for obtaining (B) and from the fact that

$$\begin{aligned} c_l' &= c_l - \sum_{q \in \mathcal{P}_l} r_q n_q^v + \sum_{q \in \mathcal{P}_l} r_q n_q^s \\ &= (c_l - r_p) - \left[\sum_{q \in \mathcal{P}_l q \neq p} r_q n_q^v + r_p (n_p^v - 1) \right] + \sum_{q \in \mathcal{P}_l} r_q n_q^s, \quad l \in p, \end{aligned}$$

we can rewrite (D) as

$$(D) = G(\underline{c} - r_{\underline{p}}\underline{e}_{\underline{p}}) \cdot W^{d}(\underline{c} - r_{\underline{p}}\underline{e}_{\underline{p}}).$$

Also following similar arguments as in simplifying (A), we can obtain (C) as

$$(C) = \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p\underline{e}_{p}})} \prod_{p \in \mathcal{P}} f_{p} w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}).$$

Therefore

•

.

.

٠

$$\begin{split} \frac{\partial W^{d}}{\partial \rho_{p}^{v}} &= -[G(\underline{c})]^{-2} \cdot G(\underline{c} - r_{p} \underline{c}_{p}) \cdot \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c})} \prod_{p \in \mathcal{P}} f_{p} \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \\ &+ \frac{1}{G(\underline{c})} \left[\frac{\alpha}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p} \underline{c}_{p})} \prod_{p \in \mathcal{P}} f_{p} w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) + \frac{\beta}{\alpha + \beta} G(\underline{c} - r_{p} \underline{c}_{p}) W^{d}(\underline{c} - r_{p} \underline{c}_{p}) \right] \\ &= \frac{G(\underline{c} - r_{p} \underline{c}_{p})}{G(\underline{c})} \left\{ \frac{\beta}{\alpha + \beta} W^{d}(\underline{c} - r_{p} \underline{c}_{p}) - W^{d}(\underline{c}) \\ &+ \frac{\alpha}{\alpha + \beta} \cdot \frac{1}{G(\underline{c} - r_{p} \underline{c}_{p})} \cdot \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p} \underline{c}_{p})} \prod_{p \in \mathcal{P}} f_{p} \cdot w^{d}(\underline{\gamma}^{d}, \underline{\rho}^{d}, \underline{c}) \right\} \\ &= (1 - B_{p}) \left\{ \frac{\beta}{\alpha + \beta} W^{d}(\underline{c} - r_{p} \underline{c}_{p}) - W^{d}(\underline{c}) \\ &+ \frac{\alpha}{\alpha + \beta} \sum_{(\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p} \underline{c}_{p})} P_{(\underline{c} - r_{p} \underline{c}_{p})} (\underline{N}^{v}, \underline{N}^{s}) \cdot \sum_{l \in \mathcal{L}} \gamma_{l}^{l} \rho_{l}^{l} \overline{P}(\rho_{l}^{l}, c_{l}^{\prime}) \cdot I(c_{l}^{\prime} > 0) \right\} \end{split}$$

where

$$c'_{l} = c_{l} - \sum_{p \in \mathcal{P}_{l}} r_{p} n_{p}^{v} + \sum_{p \in \mathcal{P}_{l}} r_{p} n_{p}^{s}, \quad (\underline{N}^{v}, \underline{N}^{s}) \in \Omega(\underline{c} - r_{p} \underline{e}_{p}).$$



Figure 1. Two-Dimensional Birth-Death Process Used for the Pascal Approximation



Figure 2a. The ten-node multihop radio network of [4]







Figure 3a. The multi-rate star network of [3]

Voice Circuit	Links Used	Bandwidth Required
1	1,2	1
2	1,3	1
3	1,4	1
4	2,3	1
5	2,4	1
6	3,4	1
7	1,2	5
8	1,3	5
9	1,4	5
10	2,3	5
11	2,4	5
12	3,4	5

Figure 3b. The twelve voice circuits of the network of Figure 3a and their bandwidth requirements





Table 1. Probability of Voice Blocking for Different Traffic Loads, Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$, and no Admission Control

Offered		Ē		
Voice Load	Exact	Monte Carlo	Knapsack	Error
ρ °	Value	Summation	Approximation	%
0.1	0.000000	(0.000000, 0.000000)	0.000000	-
0.5	0.000160	(0.000096, 0.000224)	0. 000174	8.75
1.0	0.008654	(0.008355, 0.009262)	0. 009802	13.26
1.5	0.047401	(0.044829, 0.061187)	0.053400	12.66
2.0	0.112365	(0.107544, 0.121025)	0.122152	8.71
2.5	0.183607	(0.179594, 0.185808)	0.194000	5.66
3.5	0.307931	(0.302761, 0.316174)	0.317236	3.02
4.5	0.401453	(0.385203, 0.417037)	0.411132	2.41
5.5	0.472130	(0.434905, 0.514719)	0.483226	2.35
6.5	0.527213	(0.505836, 0.540619)	0.539849	2.39
7.5	0.571417	(0.546001, 0.603585)	0.585356	2.44
8.5	0.607742	(0.534647, 0.646869)	0.622658	2.45
10.0	0.651634	(0.616248, 0.705557)	0.667461	2.43
15.0	0.745317	(0.676021, 0.778351)	0.761318	2.15

Table 2. Probability of Voice Blocking at Each Path for Different Traffic Loads, Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

•

Voice	Offered		B _p		
Path	Voice Load	Exact	Monte Carlo	Knapsack	Error
p	ρ_p^v	Value	Summation	Approximation	%
1	1.0	0.014202	(0.011101, 0.018019)	0.016567	16.65
2	1.0	0.000730	(0.000000, 0.001098)	0.000837	14.66
3	1.0	0.007371	(0.005950, 0.011421)	0.007925	7.52
4	1.0	0.007371	(0.003915, 0.008254)	0.007925	7.52
5	1.0	0.013596	(0.010745, 0.017536)	0.015753	15.86
1	5.5	0.701485	(0.688131, 0.717825)	0.703886	0.34
2	5.5	0.209734	(0.174270, 0.210641)	0.247334	17.93
3	5.5	0.409218	(0.379417, 0.428384)	0.413480	1.04
4	5.5	0.409218	(0.384999, 0.433277)	0.413135	0.96
5	5.5	0.630995	(0.617493, 0.652789)	0.638292	1.16
1	10	0.885902	(0.858356, 0.887236)	0.869108	-1.89
2	10	0.421298	(0.361902, 0.432984)	0.475328	12.82
3	10	0.571870	(0.502697, 0.570099)	0.593330	3.75
4	10	0.571870	(0.542950, 0.607079)	0.592867	3.67
5	10	0.807232	(0.777385, 0.817837)	0.806672	0.07

		$\beta/(\alpha+\beta)=0.4$		$\beta/(\alpha+\beta)=0.8$			$\beta/(\alpha+\beta)=1.0$			
Voice	Data		Q		Q			$ar{Q}$		
Load	Load	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error
$\bar{\rho}^{v}$	$ar{oldsymbol{ ho}}^{c}$	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%
0.1	0.5	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
0.1	2.5	0.561	0.563	0.36	0.729	0.730	0.14	C.827	0.827	0.00
0.1	4.0	6.319	6.321	0.03	7.313	7.315	0.03	7.852	7.852	0.00
1.0	0.5	0.016	0.023	43.75	0.719	0.743	3.34	2.202	2.250	2.18
1.0	2.5	3.658	3.710	1.42	14.022	14.077	0.39	22.102	22.102	0.00
1.0	4.0	18.994	19.096	0.53	41.084	41.051	-0.08	52.352	52.223	-0.25
2.0	0.5	0.169	0.184	8.88	5.934	5.875	-0.99	17.536	17.368	-0.25
2.0	2.5	9.704	9.579	-1.29	41.319	40.400	-2.24	59.504	58.066	-2.42
2.0	4.0	34.945	33.929	-2.91	73.337	72.121	-1.66	84.803	83.551	-1.48
3.5	0.5	0.430	0.412	-4.18	13.750	13.344	-2.95	39.469	38.276	-3.02
3.5	2.5	15.318	14.991	-2.13	62.819	61.211	-2.56	82.570	80.611	-2.37
3.5	4.0	46.638	45.838	-1.72	87.403	72.121	-1.46	95.741	94.775	-1.01
5.5	0.5	0.625	0.597	-4.48	19.891	19.298	-2.98	55.611	54.038	-2.38
5.5	2.5	18.868	18.491	-1.99	74.061	72.677	-1.87	91.884	90.556	-1.15
5.5	4.0	53.027	52.286	-1.40	94.322	93.527	-0.84	98.617	98.172	-0.45
10.	0.5	0.865	0.804	-7.05	26.723	25.830	-3.34	70.170	70.262	-2.64
10.	2.5	22.286	21.816	-2.11	83.032	81.898	-1.36	97.437	96.739	-0.72
10.	4.0	58.569	57.815	-1.28	97.608	97.203	-0.41	99.761	99.624	-0.14

Table 4a. 100x Probability of Queueing Data, for Voice Loads $\bar{\rho}^v = \rho_p^v = 0.1$, for Different Data Loads and Voice Activity Factors, and Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$

		$\beta/(\alpha+\beta)=0.4$			$\beta/(\alpha+\beta)=0.8$			$\beta/(\alpha+\beta)=1.0$		
Data	Data		Q_l		Qı			Q_l		
Link	Load	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error
1	ρ_l^d	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%
1	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
2	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
3	0.5	0.00	0.00		0.00	0.00	-	0.00	0.00	-
4	0.5	0.00	0.00	_	0.00	0.00	-	0.00	0.00	-
5	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
6	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
7	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
8	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
9	0.5	0.00	0.00	-	0.00	0.00	-	0.00	0.00	-
1	2.5	0.514	0.515	0.19	0.616	0.617	0.16	0.673	0.673	0.00
2	2.5	0.514	0.515	0.19	0.616	0.617	0.16	0.673	0.673	0.00
3	2.5	0.605	0.606	1.56	0.822	0.822	0.00	0.946	0.946	0.00
4	2.5	0.562	0.564	0.35	0.734	0.7 34	0.00	0.833	0.834	0.12
5	2.5	0.562	0.564	1.31	0.734	0.734	0.00	0.833	0.834	0.12
6	2.5	0.980	0.980	0.00	0.834	0.8 3 4	0.00	0.980	0.980	0.00
7	2.5	0.562	0.564	1.31	0.734	0.734	0.00	0.833	0.8 3 4	0.12
8	2.5	0.562	0.564	1.31	0.734	0.734	0.00	0.833	0.834	0.12
9	2.5	0.562	0.564	1.31	0.734	0.734	0.00	0.833	0.834	0.12
1	4.0	6.013	6.015	0.03	6.649	6.650	0.02	6.986	6.986	0.00
2	4.0	6.013	6.015	0.03	6.649	6.650	0.02	6.986	6.986	0.00
3	4.0	6.601	6.603	0.03	7.882	7.883	0.02	8.566	8.566	0.00
4	4.0	6.324	6.326	0.03	7.334	7.336	0.03	7.885	7.885	0.00
5	4.0	6.324	6.326	0.03	7.334	7.336	0.03	7.885	7.885	0.00
6	4.0	6.624	6.625	0.015	7.971	7.972	0.02	8.705	8.706	0.00
7	4.0	6.324	6.324	0.00	7.334	7.336	0.03	7.885	7.885	0.00
8	4.0	6.324	6.326	0.03	7.334	7.336	0.03	7.885	7.885	0.00
9	4.0	6.324	6.326	0.03	7.334	7.336	0.03	7.885	7.885	0.00

Table 4b. 100x Probability of Queueing Data, for Voice Loads $\bar{\rho}^{\nu} = \rho_{p}^{\nu} = 1.0$, for Different Data Loads and Voice Activity Factors, and Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$

		β/	$\beta/(\alpha+\beta)=0.4$		$\beta/(\alpha+\beta)=0.8$			$\beta/(\alpha+\beta)=1.0$		
Data	Data		Q_l		Qi			Q_l		
Link	Load	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error
1	ρ_l^d	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%
1	0.5	0.002	0.006	200	0.130	0.143	10.00	0.410	0.430	4.88
2	0.5	0.002	0.006	200	0.130	0.143	10.00	0.410	0.430	4.88
3	0.5	0.019	0.033	73.68	0.889	0.921	3.60	2.719	2.780	2.24
4	0.5	0.018	0.028	55.55	0.795	0.824	3.65	2.446	2.511	2.66
5	0.5	0.018	0.028	55.55	0.795	0.824	3.65	2.446	2.511	2.66
6	0.5	0.031	0.018	41.94	1.348	1.395	2.13	4.052	4.052	0.00
7	0.5	0.018	0.028	55.55	0.824	1.359	64.93	2.446	2.551	4.29
8	0.5	0.018	0.028	55.55	0.795	0.824	3.65	2.446	2.551	4.29
9	0.5	0.018	0.028	55.55	0.795	0.795	0.00	2.446	2.551	4.29
1	2.5	2.073	2.115	2.03	6.659	6.73 0	1.07	10.486	10.581	0.91
2	2.5	2.073	2.115	2.03	6.659	6.730	1.07	10.486	10.581	0.91
3	2.5	4.479	4.549	1.56	17.166	17.217	0.29	26.923	26.862	-0.22
4	2.5	3.821	3.821	0.00	15.031	15.080	0.33	26.886	23.831	-11.36
5	2.5	3.821	3.871	1.31	15.031	15.080	0.33	23.886	23.831	-0.23
6	2.5	5.188	5.260	1.39	20.560	20.641	0.39	31.856	31.859	0.00
7	2.5	3.821	3.871	1.31	15.031	15.080	0.33	23.886	23.831	-11.36
8	2.5	3.821	3.871	1.31	15.031	15.081	0.33	23.886	23.831	-11. 3 6
9	2.5	3.821	3.871	1.31	15.031	15.080	0.33	23.886	23.831	-11. 3 6
1	4.0	13.490	13.606	0.86	26.953	26.988	0.13	34.960	34.904	-0.16
2	4.0	1 3.49 0	13.606	0.86	26.953	26.953	0.00	34.960	34.907	-0.15
3	4.0	22.250	22.353	0.46	48.131	48.13 1	0.00	60.682	60.524	-0.32
4	4.0	19.525	19.619	0.48	43.041	42.917	-0.29	55.032	54.856	-0.32
5	4.0	19.525	19.619	0.48	43.041	42.917	-0.29	55.032	54.856	-0.32
6	4.0	24.089	24.207	0.49	52.516	52.556	0.07	65.402	65.393	-0.01
7	4.0	19.525	19.619	0.48	43.041	42.917	-0.29	55.032	54.856	-0.32
8	4.0	19.525	19.619	0.48	43.041	42.917	-0.29	55.032	54.856	-0.32
9	4.0	19.525	19.619	0.48	43.041	42.917	-0.29	55.032	54.856	-0.32

Table 4c. 100x Probability of Queueing Data, for Voice Loads $\tilde{\rho}^v = \rho_p^v = 2.0$, for Different Data Loads and Voice Activity Factors, and Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$

		$\beta/(\alpha+\beta)=0.4$		$\beta/(\alpha+\beta)=0.8$			$\beta/(\alpha+\beta)=1.0$			
Data	Data		Q_l		Qı			Qi		
Link	Load	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error
1	ρ_l^d	Value	Approx.	%	Value	Approx.	%	Value	Approx.	7
1	0.5	0.193	0.095	-50.78	1.688	1.727	2.31	5.085	5.253	3.30
2	0.5	0.191	0.095	-50.28	1.688	1.727	2.31	5.085	5.253	3.30
3	0.5	0.437	0.354	-18.99	7.510	7.392	-2.41	21.926	21.562	-1.66
4	0.5	0.360	0.279	-22.50	6.418	6.333	-1.32	19.231	18.958	-1.42
5	0.5	0.360	0.279	-22.50	6.418	6.333	-1.32	19.231	18.958	-1.42
6	0.5	0.560	0.495	-11.61	10.430	10.368	-0.59	29.570	29.461	-0.37
7	0.5	0.360	0.279	-22.50	6.418	6.333	-1.32	19.231	18.958	-1.42
8	0.5	0.360	0.279	-22.50	6.418	6.331	-1.35	19.231	18.952	-1.45
9	0.5	0.360	0.279	-22.50	6.418	6.333	-1.32	19.231	18.956	-1.43
1	2.5	6.950	6.836	-1.64	22.198	21.868	-1.49	34.654	33.990	-1.92
2	2.5	6.950	6.836	-1.64	22.198	21.868	-1.49	34.654	33.990	-1.92
3	2.5	15.750	14.820	-5.84	48.730	48.559	-2.35	69.777	68.087	-2.92
4	2.5	12.493	12.140	-2.82	44.225	42.972	-2.83	63.989	62.043	-3.04
5	2.5	12.493	12.140	-2.82	44.225	42.972	-2.83	63.989	62.043	-3.04
6	2.5	17. 3 81	17.331	-0.29	56.619	56.462	-0.28	76.507	76.330	-0.23
7	2.5	12.493	12.140	-2.83	44.225	42.970	-2.92	63.989	62.041	-3.04
8	2.5	12.493	12.135	-2.87	44.225	42.965	-2.85	63.989	62.034	-3.04
9	2.5	12.493	12.140	-2.82	44.225	42.970	-2.92	63.989	62.041	-3.04
1	4.0	28.411	27.943	-1.65	53.795	52.836	-1.78	67.209	65.870	-1.98
2	4.0	28.411	27.943	-1.65	53.795	52.836	-1.78	67.209	65.879	-1.98
3	4.0	47.734	46.904	-1.74	82.588	81.412	-1.42	92.433	91.445	-1.07
4	4.0	41.161	40.244	-2.23	76.716	75.168	-2.02	88.363	86.855	-1.71
5	4.0	41.161	40.244	-2.23	76.716	75.168	-2.02	88.363	86.855	-1.71
6	4.0	51.141	51.031	-0.22	86.280	86.172	-0.13	94.567	94.487	-0.08
7	4.0	41.161	40.243	-2.23	76.716	75.166	-2.02	88.363	86.854	-1.71
8	4.0	41.161	40.234	-2.25	76.716	75.162	-2.02	88. 363	86.850	-1.71
9	4.0	41.161	40.243	-2.23	76.716	75.166	-2.02	88.363	86.854	-1.71

Table 5. Probability of Voice Blocking for Different Traffic Loads, Node Transceiver Vector $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4)$, Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$, and No Admission Control

Offered		Ē	
Voice Load	Exact	Knapsack	Error
<i>ρ</i> ۳	Value	Approximation	%
0.1	0.000292	0.000327	11.98
0.5	0.050059	0.058689	17.24
1.0	0.195044	0.215835	10.66
2.0	0.417121	0.442862	6.17
3.0	0.541520	0.569988	5.62
4.0	0.619669	0.649299	4.78
5.0	0.673842	0.703402	4.39
6.0	0.713922	0.742704	4.03
7.0	0.744934	0.772556	3.71
8.0	0.769724	0.796051	3.42
10.0	0.807006	0.830684	2.93
15.0	0.862375	0.880563	2.11

Table 6. 100x Probability of Queueing Data for Various Voice and Data Loads, Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$ and Node Transceiver Vector $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4, 4)$

Offered	Offered		$ar{Q}$	
Voice Load	Data Load	Exact	Knapsack	Percent
$\bar{ ho}^{v}$	<i>₽</i> ₫	Value	Approximation	Error (%)
0.1	0.2	0.033	0.034	3.03
0.1	1.0	2.860	2.860	0.00
0.1	2.0	20.796	19.687	-5.33
1.0	0.2	1.636	1.585	-3.12
1.0	1.0	14.590	14.202	-2.66
1.0	2.0	45.702	44.874	-1.81
5.0	0.2	4.762	4.505	-5.40
5.0	1.0	27.737	26.754	-3.54
5.0	2.0	64.540	63.299	-1.92
10.0	0.2	5.773	5.411	-6.27
10.0	1.0	31.143	29.915	-3.94
10.0	2.0	68.419	67.173	-1.82

Table 7a. 100x Probability of Queueing Data at Each Link for Voice Loads $\bar{\rho}^{\nu} = \rho_{p}^{\nu} = 0.1$, Different Data Loads, $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4, 4)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered		Qı	
Link	Data Load	Exact	Knapsack	Error
1	ρ_l^d	Value	Approximation	%
1	0.2	0.020	0.021	5.00
2	0.2	0.020	0.021	5.00
3	0.2	0.042	0.043	2.38
4	0.2	0.033	0.034	3.03
5	0.2	0.033	0.034	3.03
6	0.2	0.046	0.047	2.17
7	0.2	0.033	0.034	3.03
8	0.2	0.033	0.034	3.03
9	0.2	0.033	0.034	3.03
1	1.0	2.527	2.528	0.04
2	1.0	2.527	2.528	0.04
3	1.0	3.154	3.155	0.03
4	1.0	2.867	2.867	0.00
5	1.0	2.867	2.867	0.00
6	1.0	3.194	3.194	0.00
7	1.0	2.867	2.867	0.00
8	1.0	2.867	2.867	0.00
9	1.0	2.867	2.867	0.00
1	2.0	18.538	18.540	0.01
2	2.0	18.538	18.540	0.01
3	2.0	20.747	20.749	0.01
4	2.0	19.703	19.705	0.01
5	2.0	19.703	19.705	0.01
6	2.0	20.825	20.827	0.05
7	2.0	19.703	19.705	0.01
8	2.0	19.703	19.705	0.01
9	2.0	19.703	19.705	0.01

Table 7b. 100x Probability of Queueing Data at Each Link for Voice Loads $\bar{\rho}^{v} = \rho_{p}^{v} = 1.0$, Different Data Loads, $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4, 4)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered		Qı	
Link	Data Load	Exact	Knapsack	Error
1	ρđ	Value	Approximation	%
1	0.2	0.820	0.796	-2.93
2	0.2	0.820	0.796	-2.93
3	0.2	2.219	2.130	-4.01
4	0.2	1.645	1.589	-3.40
5	0.2	1.645	1.589	-3.40
6	0.2	2.639	2.599	-1.52
7	0.2	1.645	1.589	-3.40
8	0.2	1.645	1.589	-3.40
9	0.2	1.645	1.589	-3.40
1	1.0	9.928	9.676	-2.54
2	1.0	9.928	9.676	-2.54
3	1.0	18.381	17.832	-2.99
4	1.0	14.670	14.226	-3.02
5	1.0	14.670	14.226	-3.02
6	1.0	19.721	19.524	-0.99
7	1.0	14.670	14.226	-3.02
8	1.0	14.670	14.203	-3.18
9	1.0	14.670	14.226	-3.02
1	2.0	37.127	36.479	-1.75
2	2.0	37.127	36.479	-1.75
3	2.0	53.274	52.282	-1.86
4	2.0	45.880	44.920	-2.09
5	2.0	45.880	44.920	-2.09
6	2.0	54.387	54.070	-0.58
7	2.0	45.880	44.920	-2.09
8	2.0	45.880	44.879	-2.18
9	2.0	45.880	44.920	-2.09

Data	Offered		Q_l	
Link	Data Load	Exact	Knapsack	Error
1	ρ_l^d	Value	Approximation	%
1	0.2	3.560	3.202	-10.06
2	0.2	3.560	3.202	-10.06
3	0.2	7.074	6.263	-11.46
4	0.2	4.285	4.215	-1.63
5	0.2	4.285	4.216	-1.61
6	0.2	7.235	6.909	-4.50
7	0.2	4.285	4.213	-1.68
8	0.2	4.285	4.113	-4.01
9	0.2	4.285	4.213	-1.68
1	1.0	23.141	21.667	-6.37
2	1.0	23.141	21.667	-6.37
3	1.0	37.104	34.331	-7.47
4	1.0	25.917	25.615	-1.16
5	1.0	25.917	25.617	-1.16
6	1.0	36.666	35.430	-3.37
7	1.0	25.917	25.608	-1.19
8	1.0	25.917	25.238	-2.62
9	1.0	25.917	25.609	-1.19
1	2.0	58.607	56.549	-3.51
2	2.0	58.607	56.549	-3.51
3	2.0	76.910	73.960	-3.84
4	2.0	62.263	61.840	-0.68
5	2.0	62.263	61.840	-0.68
6	2.0	75.419	73.927	-1.98
7	2.0	62.263	61.832	-0.69
8	2.0	62.263	61.360	-1.45
9	2.0	62.263	61.832	-0.69

Table 7d. 100x Probability of Queueing Data at Each Link for Voice Loads $\bar{\rho}^{v} = \rho_{p}^{v} = 10.0$, Different Data Loads, $\underline{T} = (4, 4, 4, 4, 4, 4, 4, 4, 4, 4)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered	Qi				
Link	Data Load	Exact Knapsack		Error		
1	ρđ	Value	Approximation	%		
1	0.2	4.684	4.193	-10.48		
2	0.2	4.684	4.192	-10.50		
3	0.2	8.786	7.567	-13.87		
4	0.2	5.023	4.958	-1.29		
5	0.2	5.023	4.958	-1.29		
6	0.2	8.694	8.110	-6.72		
7	0.2	5.022	4.956	-1.31		
8	0.2	5.022	4.812	-4.18		
9	0.2	5.022	4.956	-1.31		
1	1.0	27.294	25.533	-6.45		
2	1.0	27.294	25.529	-6.47		
3	1.0	42.279	38.508	-9.39		
4	1.0	28.449	28.223	-0.79		
5	1.0	28.449	28.225	-0.79		
6	1.0	41.173	39.049	-5.16		
7	1.0	28.449	28.218	-0.81		
8	1.0	28.449	27.731	-2.52		
9	1.0	28.449	28.219	-3.40		
1	2.0	63.910	61.736	-3.40		
2	2.0	63 .910	61.732	-3.41		
3	2.0	81.697	78.150	-4.34		
4	2.0	65.276	65.007	-0.41		
5	2.0	65.276	65.009	-0.41		
6	2.0	79.873	77.490	-2.98		
7	2.0	65.276	65.002	-0.42		
8	2.0	65.276	65.430	0.24		
9	2.0	65.276	65.002	-0.42		

Table 8. Probability of Voice Blocking for Different Traffic LoadsNode Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$

Voice Transceiver Vector $\underline{T}^* = (5, 7, 7, 7, 4, 7, 4, 7, 7, 7)$ Data Link Capacity Vector $\underline{c}^d = (1, 1, 1, 1, 1, 1, 1, 1, 1)$ Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Voice	Offered	<i>B</i> ,					
Path	Voice Load	Exact	Monte Carlo	Knapsack	Error		
р	ρ_p^v	Value	Summation	Approximation	%		
1	2.5	0.634	(0.673, 0.686)	0.705	11.19		
2	2.5	0.144	(0.133, 0.150)	0.147	2.08		
3	2.5	0.429	(0.416, 0.433)	0.416	-3.03		
4	2.5	0.429	(0.416, 0.433)	0.416	-3.03		
5	2.5	0.641	(0.627, 0.642)	0.658	2.65		
Average	2.5	0.465	(0.453, 0.69)	0.468	0.65		
Offered	Offered	Ŵ					
------------	-----------------------	-------	----------------	-------	--		
Voice Load	Data Load	Exact	Exact Knapsack				
<i>ρ</i> ۳	<i>₽</i> ^d	Value	Approximation	%			
2.5	0.7	0.021	0.023	9.52			
2.5	0.9	0.062	0.071	14.51			
2.5	0.999	5.050	5.806	14.97			
10.	0.7	0.034	0.036	5.88			
10.	0.9	0.108	0.111	2.78			
10.	0.999	9.324	9.599	2.95			

Table 10a. Average Data Delay at Each Link for Voice Loads $\bar{\rho}^{v} = \rho_{p}^{v} = 2.5$, Different Data Loads, $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, $\underline{T}^{v} = (5, 7, 7, 7, 4, 7, 4, 7, 7, 7)$, $\underline{c}^{d} = (1, 1, 1, 1, 1, 1, 1, 1)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered	<u> </u>	Wi			
Link	Data Load	Exact	Exact Knapsack			
1	ρ_l^d	Value	Approximation	%		
1	0.7	0.004	0.006	50.00		
2	0.7	0.004	0.006	50.00		
3	0.7	0.026	0.028	7.69		
4	0.7	0.023	0.026	13.04		
5	0.7	0.023	0.026	13.04		
6	0.7	0.039	0.039	0.00		
7	0.7	0.023	0.026	13.04		
8	0.7	0.023	0.026	13.04		
9	0.7	0.023	0.026	13.04		
1	0.9	0.011	0.017	54.55		
2	0.9	0.011	0.017	54.55		
3	0.9	0.077	0.086	11.69		
4	0.9	0.068	0.079	16.18		
5	0.9	0.068	0.079	16.18		
6	0.9	0.119	0.120	0.84		
7	0.9	0.068	0.079	16.18		
8	0.9	0.068	0.079	16.18		
9	0.9	0.068	0.079	16.18		
1	0.999	0.743	1.149	54.64		
2	0.999	0.743	1.149	54.64		
3	0.999	6.210	7.097	14.28		
4	0.999	5.516	6.512	18.06		
5	0.999	5.516	6.512	18.06		
6	0.999	10.175	10.297	1.20		
7	0.999	5.516	6.512	18.06		
8	0.999	5.516	6.511	18.04		
9	0.999	5.516	6.512	18.04		

Table 10b. Average Data Delay at Each Link for Voice Loads $\bar{\rho}^{v} = \rho_{p}^{v} = 10.0$, Different Data Loads, $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, $\underline{T}^{v} = (5, 7, 7, 7, 4, 7, 4, 7, 7, 7)$, $\underline{c}^{d} = (1, 1, 1, 1, 1, 1, 1, 1)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered	Wi			
Link	Data Load	Exact	Knapsack	Error	
1	ρď	Value	Approximation	%	
1	0.7	0.013	0.013	0.00	
2	0.7	0.013	0.013	0.00	
3	0.7	0.046	0.046	0.00	
4	0.7	0.035	0.037	5.71	
5	0.7	0.035	0.037	5.71	
6	0.7	0.063	0.063	0.00	
7	0.7	0.035	0.037	5.71	
8	0.7	0.035	0.037	5.71	
9	0.7	0.035	0.037	5.71	
1	0.9	0.039	0.039	0.00	
2	0.9	0.039	0.039	0.00	
3	0.9	0.143	0.142	-0.70	
4	0.9	0.110	0.116	5.45	
5	0.9	0.110	0.116	5.45	
6	0.9	0.202	0.200	-0.99	
7	0.9	0.110	0.116	5.45	
8	0.9	0.110	0.115	4.45	
9	0.9	0.110	0.116	5.45	
1	0.999	3.032	3.029	-0.09	
2	0.999	3.032	3.029	-0.09	
3	0.999	12.390	12.350	-0.32	
4	0.999	9.458	10.008	5.82	
5	0.999	9.458	10.009	5.83	
6	0.999	18.173	18.007	-0.91	
7	0.999	9.458	10.006	5.79	
8	0.999	9.458	9.949	5.19	
9	0.999	9.458	10.006	5.79	



100*Probability of Voice Blocking (<u>a</u>)



Table 11. Percentage of Voice Calls Blocked for Different Capacity Allocations in the Multi-Rate Network of [6]

Capacity Allocation 1 $\underline{c} = (90, 100, 110, 120)$

Capacity Allocation 2 c = (18, 20, 22, 24)

Capacity Allocation 3 $\underline{c} = (9, 10, 11, 12)$

Capacity Allocation 4 $\underline{c} = (5, 5, 6, 6)$

Vector of Voice Path Rates $\underline{r} = (1, 1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5, 5)$

Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

	Offered	 B			
Capacity	Voice Load	Monte Carlo	Knapsack	Pascal	
Allocation	٣	Summation	Approximation	Approximation	
1	8.0	(0.052, 0.058)	0.061	0.072	
1	10.0	(0.669, 0.732)	0.741	0.756	
1	15.0	(9.947, 10.275)	10.359	10.487	
2	0.3	(0.042, 0.042)	0.042	0.109	
2	0.7	(0.707, 0.785)	0.779	0.921	
2	1.7	(9.466, 9.686)	10.137	9.758	
3	0.004	(0.049, 0.070)	0.061	0.038	
3	0.04	(0.630, 0.696)	0.681	0.437	
3	0.4	(9.884, 10.096)	10.613	8.857	
4	0.0002	(0.046, 0.046)	0.048	0.045	
4	0.003	(0.629, 0.679)	0.714	0.677	
4	0.05	(9.626, 9.864)	10.584	9.999	

 $\beta/(\alpha+\beta)=0.4$ B_p Voice Offered Path Voice Load Monte Carlo Pascal Knapsack Summation Approximation ρ_p^v Approximation р (0.008, 0.030)1 8.0 0.028 0.034 2 8.0 (0.002, 0.034)0.025 0.030 3 (0.002, 0.034)8.0 0.025 0.029 (0.001, 0.001)4 8.0 0.003 0.005 5 8.0 (0.001, 0.001)0.003 0.004 6 8.0 (0.000, 0.004)0.000 0.001 7 (0.194, 0.207)0.247 1.6 0.213 8 1.6 (0.144, 0.217)0.191 0.218 9 (0.157, 0.199)1.6 0.189 0.214 10 0.038 1.6 (0.000, 0.048)0.028 11 1.6 (0.000, 0.052)0.025 0.034 12 1.6 (0.000, 0.022)0.003 0.005 0.36 1 10.0 (0.34, 0.35)0.36 2 10.0 (0.29, 0.30)0.30 0.31 3 0.30 10.0 (0.28, 0.29)0.29 0.08 4 10.0 (0.07, 0.07)0.07 5 10.0 (0.06, 0.06)0.06 0.07 6 10.0 0.01 0.01 (0.01, 0.01)7 2.35 2.38 2.0 (2.29, 2.30)8 2.0 (1.96, 1.98)1.99 2.01 1.92 1.93 9 2.0 (1.90, 1.92)0.53 0.57 10 2.0 (0.49, 0.50)11 2.0 (0.43, 0.44)0.46 0.49 12 0.09 0.11 1.6 (0.08, 0.08)

Table 12a. Percentage of Voice Calls Blocked at Each Path for Average Voice Loads $\bar{\rho}^v = 8.0$, 10.0, and 15.0 Capacity Allocation 1, $\underline{c} =$ $(90, 100, 110, 120), \underline{r} = (1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Voice Offered Β, Path Voice Load Monte Carlo Knapsack Pascal Approximation Summation Approximation ρ_p^v p 1 (5.70, 5.70)5.70 15.0 5.70 2 15.0 (4.60, 4.60) 4.60 4.60 3 15.0 (4.10, 4.20) 4.20 4.20 4 15.0 (2.40, 2.40)2.50 2.50 5 15.0 (1.90, 1.90)2.00 2.00 6 (0.80, 0.80) 0.90 0.90 15.0 7 3.0 (28.00, 28.10)28.60 28.50 8 3.0 (23.30, 23.40) 23.80 23.70 9 (21.40, 21.40)21.60 21.60 3.0 10 (13.20, 13.20) 13.70 3.0 13.80 (10.90, 11.00)11 11.30 3.0 11.40 12 3.0 (4.80, 4.80)5.40 5.40

Table 12a (cont'd)

Table 12b. Percentage of Voice Calls Blocked at Each Path for Average Voice Loads $\bar{\rho}^v = 0.3, 0.7, \text{ and } 1.7, \text{ Capacity Allocation } 2, \underline{c} = (18, 20, 22, 24),$ $\underline{r} = (1, 1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5)$ and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Voice	Offered	B,			
Path	Voice Load	Monte Carlo	Knapsack	Pascal	
р	ρ_p^v	Summation	Approximation	Approximation	
1	0.3	(0.003, 0.003)	0.006	0.025	
2	0.3	(0.003, 0.003)	0.005	0.021	
3	0.3	(0.003, 0.003)	0.004	0.019	
4	0.3	(0.000, 0.000)	0.002	0.011	
5	0.3	(0.000, 0.000)	0.002	0.009	
6	0.3	(0.000, 0.000)	0.001	0.005	
7	0.06	(0.134, 0.176)	0.148	0.339	
8	0.06	(0.090, 0.130)	0.107	0.281	
9	0.06	(0.078, 0.134)	0.102	0.256	
10	0.06	(0.034, 0.086)	0.059	0.149	
11	0.06	(0.056, 0.056)	0.054	0.123	
12	0.06	(0.000, 0.023)	0.012	0.056	
1	0.7	(0.166, 0.227)	0.202	0.243	
2	0.7	(0.148, 0.209)	0.184	0.202	
3	0.7	(0.143, 0.186)	0.167	0.182	
4	0.7	(0.033, 0.085)	0.069	0.119	
5	0.7	(0.030, 0.058)	0.052	0.099	
6	0.7	(0.000, 0.059)	0.034	0.058	
7	0.14	(2.214, 2.402)	2.410	2.708	
8	0.14	(1.791, 1.966)	1.946	2.258	
9	0.14	(1.616, 1.773)	1.737	2.033	
10	0.14	(1.022, 1.149)	1.147	1.352	
11	0.14	(0.836, 0.958)	0.937	1.125	
12	0.14	(0.375, 0.464)	0.465	0.668	

Voice Offered Β, Path Voice Load Monte Carlo Pascal Knapsack Summation Approximation Approximation ρ_p^v p 1 (3.135, 3.368) 3.422 3.284 1.7 2 (2.574, 2.778)1.7 2.836 2.780 3 1.7 (2.255, 2.459)2.526 2.480 4 1.7 (1.826, 2.002)2.012 1.935 1.7 (1.499, 1.665) 1.700 1.633 5 1.7 (0.916, 1.046)1.120 6 1.103 7 (24.198, 24.763) 0.34 25.706 24.626 8 (21.002, 21.547) 0.34 22.265 21.311 0.34 (19.007, 19.601) 9 20.065 19.215 0.34 (14.912, 15.391)10 16.246 15.670 (12.699, 13.148)13.875 11 0.34 13.423 12 0.34 (8.795, 9.179)9.885 9.616

Table 12b (cont'd)

Table 13. 100x Probability of Queueing Data for Different Capacity Allocations in the Multi-Rate Network of [6] Capacity Allocation 1 $\underline{c} = (90, 100, 110, 120)$ Capacity Allocation 2 $\underline{c} = (18, 20, 22, 24)$ Capacity Allocation 3 $\underline{c} = (9, 10, 11, 12)$ Capacity Allocation 4 $\underline{c} = (5, 5, 6, 6)$ Vector of Voice Path Rates $\underline{r} = (1, 1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5)$ Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

	Offered	Offered	Q			
Capacity	Voice Load	Data Load	Monte Carlo	Knapsack	Pascal	
Allocation	ρ٣	$\bar{\rho}^{d}$	Summation	Approximation	Approximation	
1	8.0	39.0	(0.072, 0.078)	0.072	0.088	
1	8.0	49.0	(0.933, 0.956)	0.945	0.988	
1	8.0	62.0	(9.747, 9.824)	9.776	9.719	
1	10.0	35.0	(0.092, 0.099)	0.094	0.107	
1	10.0	43.0	(0.958, 0.982)	0.776	0.807	
1	10.0	57.0	(9.643, 9.725)	9.679	9.611	
1	15.0	30.0	(0.079, 0.088)	0.083	0.081	
1	15.0	38.0	(0.931, 0.966)	0.948	0.931	
1	15.0	50.0	(9.947, 10.275)	10.106	10.080	
2	0.3	6.0	(0.069, 0.074)	0.071	0.123	
2	0.3	9.0	(0.800, 0.817)	0.806	0.806	
2	0.3	14.0	(13.693, 13.763)	13.724	13.584	
2	0.7	4.0	(0.076, 1.083)	0.081	0.136	
2	0.7	7.5	(0.985, 1.012)	1.006	1.062	
2	0.7	12.0	(10.302, 10.387)	10.353	9.818	
2	1.7	2.0	(0.089, 0.098)	0.091	0.102	
2	1.7	5.0	(1.084, 1.116)	1.079	1.068	
2	1.7	10.0	(12.222, 12.335)	12.144	11.693	

Table 13 (cont'd)

	Offered	Offered	 Q			
Capacity	Voice Load	Data Load	Monte Carlo	Knapsack	Pascal	
Allocation	۶	$\bar{\rho}^{d}$	Summation	Approximation	Approximation	
3	0.004	2.0	(0.080, 0.084)	0.013	0.016	
3	0.004	4.0	(1.676, 1.703)	0.901	0.888	
3	0.004	6.5	(15.230, 15.290)	12.866	12.864	
3	0.04	1.8	(0.091, 0.095)	0.051	0.073	
3	0.04	3.6	(1.326, 1.353)	0.822	0.737	
3	0.04	6.0	(11.691, 11.759)	9.621	9.5 3 6	
3	0.4	0.3	(0.059, 0.067)	0.070	0.119	
3	0.4	2.0	(1.071, 1.099)	1.053	1.007	
3	0.4	5.0	(11.887, 11.996)	10.929	10.409	
4	0.0002	0.8	(0.083, 0.086)	0.084	0.082	
4	0.0002	1.4	(0.889, 0.893)	0.891	0.889	
4	0.0002	2.8	(12.345, 12.348)	12.347	12. 34 8	
4	0.003	0.7	(0.104, 0.116)	0.106	0.068	
4	0.003	1.4	(0.964, 0.978)	0.966	0.940	
4	0.003	2.8	(12.460, 12.473)	12.463	12.475	
4	0.05	0.01	(0.495, 0.525)	0.539	0.121	
4	0.05	1.0	(1.248, 1.298)	1.343	0.795	
4	0.05	2.6	(11.255, 11.304)	11.348	11.471	

Table 14. 100x Probability of Queueing Data at Each Link for Different Voice and Data Loads, Capacity Allocation 2, $\underline{c} = (18, 20, 22, 24), \underline{r} = (1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5),$ and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Data	Offered	Offered		Qı	
Link	Voice Load	Data Load	Monte Carlo	Knapsack	Pascal
1	$\bar{\rho}^{v}$	ρ_1^d	Summation	Approximation	Approximation
1	0.3	6.0	(0.208, 0.226)	0.214	0.306
2	0.3	6.0	(0.049, 0.056)	0.054	0.119
3	0.3	6.0	(0.012, 0.017)	0.014	0.047
4	0.3	6.0	(0.003, 0.004)	0.003	0.019
1	0.3	9.0	(2.197, 2.247)	2.214	2.223
2	0.3	9.0	(0.719, 0.749)	0.736	0.805
3	0.3	9.0	(0.208, 0.221)	0.215	0.298
4	0.3	9.0	(0.058, 0.065)	0.060	0.114
1	0.3	14.0	(30.647, 30.793)	30.682	31.033
2	0.3	14.0	(15.219, 15.350)	15.284	14.515
3	0.3	14.0	(6.345, 6.419)	6.403	6.261
4	0.3	14.0	(2.503, 2.549)	2.527	2.528
1	0.7	4.0	(0.206, 0.226)	0.218	0.321
2	0.7	4.0	(0.068, 0.082)	0.075	0.138
3	0.7	4.0	(0.018, 0.024)	0.023	0.060
4	0.7	4.0	(0.004, 0.006)	0.007	0.026
1	0.7	7.5	(2.540, 2.616)	2.583	2.513
2	0.7	7.5	(0.957, 1.001)	0.981	1.075
3	0.7	7.5	(0.317, 0.340)	0.340	0.461
4	0.7	7.5	(0.106, 0.119)	0.119	0.199
1	0.7	12.0	(23.175, 23.380)	23.283	21.744
2	0.7	12.0	(10.997, 11.132)	11.039	10.540
3	0.7	12.0	(4.894, 4.985)	4.976	4.840
4	0.7	12.0	(2.069, 2.126)	2.113	2.147

Table 14 (cont'd)

Data	Offered	Offered		Q1			
Link	Voice Load	Data Load	Monte Carlo	Knapsack	Pascal		
1	ρَ۳	ρ_l^d	Summation	Approximation	Approximation		
1	1.7	2.0	(0.212, 0.238)	0.218	0.229		
2	1.7	2.0	(0.090, 0.107)	0.098	0.107		
3	1.7	2.0	(0.031, 0.040)	0.036	0.049		
4	1.7	2.0	(0.011, 0.018)	0.014	0.023		
1	1.7	5.0	(2.541, 2.632)	2.521	2.351		
2	1.7	5.0	(1.088, 0.107)	1.104	1.1 3 0		
3	1.7	5.0	(0.481, 0.520)	0.488	0.537		
4	1.7	5.0	(0.184, 0.207)	0.202	0.254		
1	1.7	10.0	(24.714, 24.963)	24.514	23.500		
2	1.7	10.0	(13.433, 13.621)	13.383	12.965		
3	1.7	10.0	(7.186, 7.331)	7.169	6.830		
4	1.7	10.0	(3.443, 3.539)	3.508	3.479		

•

Offered		Ē			
Voice Load	Exact	Monte Carlo	Knapsack	Percent	Thresholds
Ē	Value	Summation	Approximation	Error (%)	
2.5	0.185545	(0.179594, 0.185808)	0.201855	8.79	(6,6,6,6,6)
3.5	0.312365	(0.302761, 0.316174)	0.331030	5.98	(6,6,6,6,6)
4.5	0.408949	(0.385203, 0.417037)	0.429108	4.93	(6,6,6,6,6)
5.5	0.482665	(0.434905, 0.514719)	0.503491	4.31	(6,6,6,6,6)
6.5	0.540419	(0.505836, 0.540619)	0.561224	3.85	(6,6,6,6,6)
7.5	0.586800	(0.546001, 0.603585)	0.607042	3.45	(6,6,6,6,6)
8.5	0.624825	(0.534647, 0.646869)	0.644257	3.11	(6,6,6,6,6)
10.	0.670493	(0.616248, 0.705557)	0.688468	2.68	(6,6,6,6,6)
15.	0.766014	(0.676021, 0.778351)	0.779502	1.76	(6,6,6,6,6)

(i) Equal Thresholds = 6 on Individual Voice Path Traffic

(ii) Optimal Individual Thresholds ≤ 6

Offered		Ē			
Voice Load	Exact	Monte Carlo	Knapsack	Percent	Thresholds
$ar{ ho}^{m{v}}$	Value	Summation	Approximation	Error (%)	
2.5	0.185512	(0.183525, 0.186201)	0.202398	9.10	(5,6,6,6,6)
3.5	0.310001	(0.308286, 0.311606)	0.317236	2.33	(2,6,6,6,5)
4.5	0.400543	(0.398409, 0.402028)	0.431677	7.77	(2,6,6,6,3)
5.5	0.469876	(0.468602, 0.471287)	0.509798	8.49	(1,6,6,6,2)
6.5	0.524440	(0.523098, 0.526017)	0.562409	7.24	(1, 6, 6, 6, 2)
7.5	0.569865	(0.568632, 0.570945)	0.613359	7.63	(1, 6, 6, 6, 1)
8.5	0.607519	(0.606093, 0.608414)	0.647777	6.62	(1, 6, 6, 6, 1)
10.	0.654345	(0.653336, 0.655637)	0.689589	5. 39	(1, 6, 6, 6, 1)
15.	0.755467	(0.754448, 0.756588)	0.778174	3.00	(1,6,6,6,1)

Table 15 (cont'd)

Offered	- <u> </u>	$ar{B}$			
Voice Load	Exact	Monte Carlo	Knapsack	Percent	Thresholds
ρ°	Value	Summation	Approximation	Error (%)	
2.5	0.185511	(0.183525, 0.186201)	0.220061	18.62	(5,6,6,6,6,8,8,7,8,8)
3.5	0.309905	(0.308566, 0.312519)	0.355329	14.66	(3,6,6,6,5,8,8,5,8,8)
4.5	0.399708	(0.399676, 0.403976)	0.451758	13.02	(2,6,6,6,4,8,8,4,8,8)
5.5	0.468216	(0.469929, 0.474508)	0.524826	12.09	(2,6,6,6,3,8,8,3,8,8)
6.5	0.522607	(0.525087, 0.529143)	0.584340	11.81	(2,6,6,6,2,8,8,2,8,8)
7.5	0.567119	(0.571885, 0.576496)	0.625008	10.21	(2,6,6,6,2,8,8,2,8,8)
8.5	0.605317	(0.610714, 0.615877)	0.658781	8.83	(2,6,6,6,2,8,8,2,8,8)
10.	0.652700	(0.658680, 0.664670)	0.699545	7.18	(2,6,6,6,2,8,8,2,8,8)
15.	0.754687	(0.757665, 0.766253)	0.785185	4.04	(2,6,6,6,2,8,8,2,8,8)

(iii) Optimal Full Admission Controls for $x_p \leq 6$

	Thresholds = (6, 6, 6, 6, 6)		Threshold	s=(6,6,6,6,5)	Thresholds	s=(6,6,6,6,4)	Thresholds	=(6,6,6,6,3)
Voice		$\overline{B_p}$		B _p		B _p	B _p	
Path	Exact	Knaps.	Exact	Knaps.	Exact	Exact Knaps.		Knaps.
р	Value	Approx.	Value	Approx.	Value	Approx.	Value	Approx.
1	0.289575	0.316342	0.288406	0.313684	0.282039	0.306347	0.263071	0.289559
2	0.047056	0.064568	0.047098	0.064794	0.047404	0.065425	0.048638	0.066899
3	0.160361	0.168923	0.159399	0.167262	0.154592	0.162693	0.141113	0.152319
4	0.160361	0.168906	0.159399	0.167265	0.154592	0.162744	0.141113	0.152452
5	0.270387	0.290538	0.274139	0.300755	0.295341	0.329219	0.363408	0.395894
Mean	0.185548	0.201855	0.185688	0.202752	0.186794	0.205286	0.191469	0.211425

	Thresholds	s=(6,6,6,6,2)	Threshold	s=(6,6,6,6,1)	Thresholds	s=(5,6,6,6,6)	Thresholds = (5, 6, 6, 6, 5)	
Voice		B_p		B _p		B _p	B _p	
Path	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.
p	Value	Арргох.	Value	Approx.	Value	Approx.	Value	Approx.
1	0.228059	0.258735	0.182889	0.214820	0.291666	0.324804	0.290500	0.322249
2	0.051652	0.069716	0.056587	0.074001	0.046530	0.063860	0.046571	0.064073
3	0.117384	0.133544	0.088578	0.107302	0.159831	0.167579	0.158866	0.165881
4	0.117384	0.133729	0.088578	0.107406	0.159831	0.167581	0.158866	0.165905
5	0.505113	0.525033	0.722943	0.729526	0.269719	0.288164	0.273474	0.298491
Mean	0.203918	0.224151	0.227915	0.246611	0.185515	0.202398	0.185655	0.203320

	Threshold	s = (5, 6, 6, 6, 4)	Thresholds = (5, 6, 6, 6, 3)		Thresholds	s=(5,6,6,6,2)	Thresholds = (5, 6, 6, 6, 1)	
Voice		B _p		B_p		B _p	<i>B</i> _p	
Path	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.
р	Value	Approx.	Value	Арргох.	Value	Approx.	Value	Approx.
1	0.284161	0.315225	0.265329	0.299243	0.230826	0.270156	0.187319	0.229298
2	0.046869	0.064666	0.048063	0.066049	0.050938	0.068647	0.055425	0.072548
3	0.154044	0.161223	0.140500	0.150680	0.116583	0.131662	0.087227	0.105221
4	0.154044	0.161299	0.140506	0.140841	0.116583	0.131875	0.087227	0.103339
5	0.294691	0.327185	0.362802	0.394364	0.304595	0.523975	0.722627	0.729078
Mean	0.186762	0.205920	0.191440	0.210235	0.163905	0.225263	0.227965	0.247897

Table 17. Comparison of Exact Value and Knapsack Approximation of the Probability of Voice Blocking for Different Full Admission Controls, Voice Loads $\bar{\rho}^v = \rho_p^v = 2.5$, $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

	Th=(6,6,6	5,6,6,8,8,8,8,8)	Th=(6,6,6	5,6,6,8,8,7,8,8)	Th=(6,6,6	5,6,6,8,8,6,8,8)	Th = (6, 6, 6, 6, 6, 8, 8, 5, 8, 8)		
Voice		B _p		B _p		B _p	B _p		
Path	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	
р	Value	Approx.	Value	Approx.	Value	Approx.	Value	Approx.	
1	0.289575	0.331588	0.290114	0.341043	0.299332	0.358541	0.338108	0.394865	
2	0.047056	0.063302	0.046979	0.062597	0.045832	0.061112	0.041908	0.058403	
3	0.160361	0.192089	0.159809	0.188046	0.151728	0.179596	0.124889	0.164308	
4	0.160361	0.192232	0.159808	0.188228	0.151728	0.179853	0.124889	0.164312	
5	0.270387	0.309768	0.271007	0.319049	0.281622	0.338953	0.325117	0.377960	
Mean	0.185548	0.217796	0.185543	0.219793	0.186048	0.223611	0.190982	0.231969	

	Th=(6,6,6,6,5,8,8,8,8,8)		Th=(6,6,6,6,5,8,8,7,8,8)		Th=(6,6,6	,6,5,8,8,6,8,8)	Th=(6,6,6,6,5,8,8,5,8,8)		
Voice		B _p	B _p			B _p	Bp		
Path	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	Exact	Knaps.	
р	Value	Approx.	Value	Approx.	Value	Approx.	Value	Approx.	
1	0.288406	0.329649	0.288937	0.338184	0.298430	0.356612	0.338108	0.393149	
2	0.047098	0.063465	0.047019	0.062760	0.045855	0.061269	0.041908	0.058581	
3	0.159399	0.190586	0.158921	0.186694	0.151316	0.178507	0.124889	0.163610	
4	0.159399	0.190735	0.158921	0.186875	0.151316	0.178752	0.124889	0.163605	
5	0.274139	0.318288	0.274612	0.327005	0.283788	0.345839	0.325117	0.383145	
Mean	0.185688	0.218545	0.185682	0.220304	0.186141	0.224196	0.190982	0.232418	

	Th=(6,6,6,6,4,8,8,8,8,8) Th=(6,6,6,6,4,			6,6,4,8,8,7,8,8)	Th=(6,6,6	5,6,4,8,8,6,8,8)	Th=(6,6,6	6,6,4,8,8,5,8,8)
Voice	B _p		B _p			B _p	B _p	
Path	Exact	Knaps.	Exact	Knaps.	Exact	Exact Knaps.		Knaps.
р	Value	Approx.	Value	Approx.	Value	Approx.	Value	Approx.
1	0.282039	0.324108	0.282489	0.332499	0.291800	0.350841	0.333782	0.387673
2	0.047404	0.063930	0.047328	0.063232	0.046101	0.061737	0.041997	0.058941
3	0.154592	0.186303	0.154254	0.182783	0.148010	0.175258	0.123861	0.161392
4	0.154592	0.186474	0.154254	0.182972	0.148010	0.175487	0.123861	0.161365
5	0.295341	0.342958	0.295602	0.350382	0.301656	0.366705	0.333778	0.399807
Mean	0.186794	0.220755	0.186785	0.222374	0.191456	0.226006	0.187115	0.233836

Table 18. 100x Probability of Queueing Data for Different Voice Loads and Thresholds, Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$, Data Loads $\bar{\rho}^d = \rho_l^d = 2.5$, and Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Offered			$ar{Q}$	
Voice Load	Path	Exact	Knapsack	Percent
<i>ρ</i> •	Thresholds	Value	Approximation	Error (%)
2.5	(5,6,6,6,6)	12.073	11.611	-3.83
3.5	(5,6,6,6,5)	14.054	13.358	-4.95
4.5	(2,6,6,6,3)	16.012	14.794	-7.61
5.5	(1,6,6,6,2)	15.284	13.834	-9.49
6.5	(1,6,6,6,2)	16.778	15.104	-9.98
7.5	(1,6,6,6,1)	16.839	14.511	-13.82
8.5	(1,6,6,6,1)	17.897	15.357	-14.19
10.	(1,6,6,6,1)	19.110	16. 34 5	-14.47
15.	(1,6,6,6,1)	21.397	18.474	-13.66

	Voic	Voice Load $\bar{\rho}^v = 2.5$			e Load $\bar{\rho}^r =$	3.5	Voice Load $\bar{\rho}^{\nu} = 4.5$			
	Thres	holds=(5,6,6	6,6,6)	Thresholds = (5, 6, 6, 6, 5)			Thresholds = (2, 6, 6, 6, 3)			
Data	Qı			Qı				Q_l		
Link	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error	
1	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%	
1	6.864	6.633	-3.37	7.814	7.443	-4.75	10.232	9.325	-8.86	
2	6.864	6.633	-3.37	7.814	7.444	-4.74	10.232	9.325	-8.86	
3	15.103	14.573	-3.51	17.818	16.854	-5.41	20.562	18.905	-8.06	
4	12.490	11.998	-3.94	14.491	13.727	-5.27	16.014	14.811	-7.51	
5	12.490	11.998	-3.94	14.491	13.726	-5.28	16.014	14.810	-7.52	
6	17.376	17.123	-1.46	20.586	19.864	-3.51	23.009	21.579	-6.21	
7	12.490	11.998	-3.94	14.491	13.730	-5.25	16.015	14.816	-7.49	
8	12.490	11.994	-3.97	14.491	13.707	-5.41	16.015	14.760	-7.84	
9	12 490	11.998	-3.94	14.491	13.730	-5.25	16.015	14.816	-7.49	

	Voic	e Load $\bar{\rho}^v =$	5.5	Voic	e Load $\bar{\rho}^{v} =$	6.5	Voice Load $\bar{\rho}^{*} = 7.5$		
	Three	bolds = (1, 6, 6)	6,6,2)	Thres	bolds = (1, 6,	6,6,2)	Thresholds = (1, 6, 6, 6, 1)		
Data		Q_l			Q_l		Q_l		
Link	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error
1	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%
1	9.280	9.127	-1.65	10.390	10.217	-1.66	11.626	11.214	-3.54
2	9.280	9.127	-1.65	10.390	10.217	-1.66	11.626	11.214	-3.54
3	19.603	17.888	-8.75	21.527	19.503	-9.40	21.915	19.077	-12.95
4	15.302	13.642	-10.85	16.753	14.843	-11.40	16.329	13.783	-15.59
5	15.302	13.642	-10.85	16.753	14.843	-11.40	16.329	13.783	-15.59
6	22.885	20.317	-11.22	24.930	21.992	-11.78	24.733	20.581	-16.79
7	15. 3 01	13.646	-10.82	16.753	14.847	-11.38	16.330	13.786	-15.58
8	15.301	13.472	-11.95	16.753	14.623	-12.71	16.330	13.371	-18.12
9	15. 3 01	13.646	-10.82	16.753	14.847	-11.38	16.330	13.786	-15.58

[^{**}	Voic	the Load $\bar{\rho}^{\nu} =$: 8.5	Voice	e Load $\bar{\rho}^v =$	10.0	Voic	e Load $\bar{\rho}^{\nu} =$	15.0	
	Three	bolds = (1, 6,	6,6,1)	Thresholds = (1, 6, 6, 6, 1)			Thresholds = (1, 6, 6, 6, 1)			
Data	Q_{l}			Qı				Qi		
Link	Exact	Knapsack	Error	Exact	Knapsack	Error	Exact	Knapsack	Error	
1	Value	Approx.	%	Value	Approx.	%	Value	Approx.	%	
1	12.389	11.962	-3.45	13.268	12.869	-3.00	14.932	14.793	-0.93	
2	12.389	11.962	-3.45	13.268	12.870	-3.00	14.932	14.793	-0.93	
3	23.239	20.114	-13.45	24.747	21.356	-13.70	27.569	23.974	-13.04	
4	17.373	14.570	-16.13	18.572	15.520	-16.43	20.842	17.534	-15.87	
5	17.373	14.570	-16.13	18.572	15.520	-16.43	20.842	17.535	-15.87	
6	26.189	21.654	-17.32	27.844	22.936	-17.63	30.931	25.648	-17.08	
7	17.373	14.577	-16.09	18.572	15.527	-16.39	20.842	17.551	-15.79	
8	17.373	14.101	-18.83	18.572	14.981	-19.33	20.842	16.888	-18.97	
9	17.373	14.577	-16.09	18.572	15.528	-16.39	20.842	17.552	-15.78	

Table 19 (cont'd)

	No C	Control	Kna	psack-Based	Thresholds		Optimal Thresholds		
Voice		₿		Ē			<u></u> <u> </u> <i>B</i>		
Load	Exact	Knapsack	Knapsack	Path	Exact	Gain	Exact	Path	Gain
ρ̈́υ	Value	Approx	Approx	Thresh	Value	%	Value	Thresh	%
3.5	0.307931	0.317236	0.318865	(3,8,8,8,8)	0.305456	0.80	0.304054	(2,8,8,8,6)	0.83
4.5	0.401453	0.411132	0.407689	(1,8,8,8,8)	0.386530	3.72	0.385776	(1,8,8,8,3)	3.91
5.5	0.472130	0.483226	0.477817	(1,8,8,8,8)	0.453197	4.01	0.448496	(1,8,8,8,2)	5.01
6.5	0.527213	0.539849	0.534412	(1,8,8,8,4)	0.507246	3.79	0.498001	(1,8,8,8,1)	5.53
7.5	0.571417	0.585356	0.580005	(1,8,8,8,3)	0.550631	3.64	0.541157	(1,8,8,8,1)	5.29
8.5	0.607742	0.622658	0.617512	(1,8,8,8,2)	0.584775	3.78	0.578903	(1,8,8,8,1)	4.74
10.	0.651634	0.667461	0.662817	(1,8,8,8,2)	0.632611	2.92	0.626443	(1,8,8,8,1)	3.86
15.	0.745317	0.761318	0.758245	(1,8,8,8,1)	0.730823	1.94	0.730823	(1,8,8,8,1)	1.94

Table 21. Revenue Sensitivities for the Multi-Rate Network of [6] Link Capacity Vector $\underline{c} = (90, 100, 110, 120)$ Vector of Voice Loads $\underline{\rho}^v = (10, 10, 10, 10, 10, 10, 2, 2, 2, 2, 2, 2)$ Vector of Data Loads $\underline{\rho}^d = (50, 50, 50, 50)$ Vector of Voice Rates $\underline{r} = (1, 1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5, 5, 5)$ Vector of Voice Revenue Rates $\underline{\gamma}^v = (1.0, 1.2, 1.4, 1.6, 1.8, 2.0, 3.0, 3.6, 4.2, 4.8, 5.4, 6.0)$ Vector of Data Revenue Rates $\underline{\gamma}^d = (1.0, 1.2, 1.4, 1.6)$ Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

Table 21a. Voice Revenue Sensitivity With Respect to Voice Loads

Voice	Offered		$\partial W^v / \partial \rho_p^v$	
Path	Voice Load	Monte Carlo	Knapsack	Pascal
р	ρ_p^v	Summation	Approximation	Approximation
1	10.0	(0.88, 0.99)	0.90	0.90
2	10.0	(1.13, 1.24)	1.12	1.12
3	10.0	(1.27, 1.39)	1.33	1.33
4	10.0	(1.50, 1.61)	1.57	1.57
5	10.0	(1.61, 1.81)	1.77	1.77
6	10.0	(1.97, 2.09)	1.99	1.99
7	2.0	(2.19, 2.43)	2.35	2.37
8	2.0	(3.00, 3.24)	3.07	3.08
9	2.0	(3.54, 3.79)	3.69	3.70
10	2.0	(4.48, 4.73)	4.58	4.57
11	2.0	(5.19, 5.24)	5.21	5.21
12	2.0	(5.75, 6.02)	5.95	5.94

Voice	Offered	$\partial W^d / \partial \rho_p^v$		
Path	Voice Load	Monte Carlo	Knapsack	Pascal
p	ρ_p^v	Summation	Approximation	Approximation
1	10.0	(-0.413765, -0.391275)	-0.395828	-0.385843
2	10.0	(-0.334608, -0.313041)	-0.321129	-0.314085
3	10.0	(-0.321502, -0.300196)	-0.307425	-0.299321
4	10.0	(-0.117725, -0.101117)	-0.106280	-0.106447
5	10.0	(-0.104447, -0.088175)	-0.092525	-0.091626
6	10.0	(-0.024743, -0.009671)	-0.017448	-0.019493
7	2.0	(-2.552438, -2.500914)	-2.501584	-2.444003
8	2.0	(-2.038939, -1.992859)	-1.997447	-1.960893
9	2.0	(-1.937221, -1.892167)	-1.892145	-1.852493
10	2.0	(-0.787522, -0.758234)	-0.767471	-0.759875
11	2.0	(-0.682234, -0.654666)	-0.659719	-0.648903
12	2.0	(-0.144140, -0.126442)	-0.138872	-0.149429

Table 21b. Data Revenue Sensitivity With Respect to Voice Loads

.

>

Table 21c. Data Revenue Sensitivity With Respect to Data Loads

Data	Offered	$\partial W^d / \partial \rho_l^d$		
Link	Data Load	Monte Carlo	Knapsack	Pascal
1	ρ_l^d	Summation	Approximation	Approximation
1	50.0	(0.000519, 0.012731)	0.130102	0.148864
2	50.0	(0.927529, 0.934309)	0.940640	0.940621
3	50.0	(1.356462, 1.359009)	1.356561	1.352288
4	50.0	(1.595059, 1.595804)	1.595164	1.593700

Table 22. Revenue Sensitivities for the Single-Rate Radio Network of [11] Node Transceiver Vector $\underline{T} = (8, 8, 8, 8, 8, 8, 8, 8, 8, 8, 8)$ Vector of Voice Revenue Rates $\underline{\gamma}^v = (1.0, 1.2, 1.4, 1.6, 1.8)$ Vector of Data Revenue Rates $\underline{\gamma}^d = (1.0, 1.2, 1.4, 1.6, 1.8, 2.0, 2.2, 2.4, 2.6)$ Voice Activity Factor $\beta/(\alpha + \beta) = 0.4$

	Voice	Offered	$\partial W^v / \partial \rho_p^v$			
Path		Voice Load	Exact	Knapsack	Percent	
	р	ρ_p^v	Value	Approximation	Error (%)	
	1	5.5	-0.276503	-0.221452	-19.91	
	2	5.5	0.628294	0.503593	-19.85	
	3	5.5	0.412164	0.373352	-9.42	
	4	5.5	0.455148	0.415531	-8.70	
	5	5.5	0.103284	0.114084	10.46	

Table 22a. Voice Revenue Sensitivity With Respect To Voice Loads

Table 22b. Data Revenue Sensitivity With Respect To Voice Loads

Voice	Offered	$\partial W^d / \partial \rho_p^v$		
Path	Voice Load	Exact	Knapsack	Percent
p	ρ_p^v	Value	Approximation	Error (%)
1	5.5	-0.313773	-0.435772	38.88
2	5.5	-0.301060	-0.376670	25.11
3	5.5	-0.096402	-0.277122	187.46
4	5.5	-0.327656	-0.427354	30.43
5	5.5	-0.262378	-0.382214	45.67

Voice	Offered	$\partial W^d / \partial ho_p^v$		
Path	Voice Load	Exact	Knapsack	Percent
p	ρ_p^v	Value	Approximation	Error (%)
1	5.5	-0.068434	-0.089658	31.01
2	5.5	-0.056307	-0.067183	19.31
3	5.5	-0.029062	-0.059503	104.76
4	5.5	-0.073209	-0.088104	20.35
5	5.5	-0.061949	-0.081633	31.77

Table 22c. Data Revenue Sensitivity With Respect To Data LoadsVector of Voice Loads $\rho^{v} = (5.5, 5.5, 5.5, 5.5, 5.5)$

.

?

Data	Offered	$\partial W^d / \partial \rho_l^d$		
Link	Data Load	Exact	Knapsack	Percent
1	ρ_l^d	Value	Approximation	Error (%)
1	4.0	-0.204096	-0.189882	-6.96
2	4.0	-0.244915	-0.227802	-6.99
3	4.0	-0.499528	-0.496486	-0.61
4	4.0	-0.440876	-0.431286	-2.17
5	4.0	-0.495986	-0.485410	-2.13
6	4.0	-0.672845	-0.671192	-0.25
7	4.0	-0.606184	-0.593026	-2.17
8	4.0	-0.661292	-0.645434	-2.40
9	4.0	-0.716399	-0.718822	0.34
1	2.0	0.732983	0.742983	1.36
2	2.0	0.879579	0.891355	1. 34
3	2.0	0.755747	0.774882	2.53
4	2.0	1.044098	1.052112	0.77
5	2.0	1.174610	1.183667	0.77
6	2.0	1.048758	1.052783	0.38
7	2.0	1.435631	1.446897	0.78
8	2.0	1.566143	1.580366	0.91
9	2.0	1.696655	2.079021	22.54

	Radio I	Network	Multi-Rate Network	
	$\underline{T} = (8, 8, 8, 8)$,8,8,8,8,8,8)	$\underline{c} = (90, 100, 110, 120)$	
Approximation	$\underline{r} = (1, 1, 1, 1, 1)$		$\underline{r} = (1, 1, 1, 1, 1, 1, 5, 5, 5, 5, 5, 5)$	
Method	Probability of	Probability of	Probability of	Probability of
	Voice Blocking	Data Queueing	Voice Blocking	Data Queueing
Exact	20 sec	20 sec	prohibitive	prohibitive
Monte Carlo	1 min	1 min	25 min	25 min
Knapsack	2 sec	2 sec	2.5 sec	2.5 sec
Pascal	2 sec	2 sec	2.5 sec	2.5 sec

 Table 23. Computational Effort Required for the Different Approximations and

 Network Models

.

1

.